Section 3.10. Nomination. Nominations for the election of Directors may be made only (A) by the Board of Directors or (B) by any shareholder of record entitled to vote in the election of Directors generally at the record date of the meeting and also on the date of the meeting at which Directors are to be elected. However, any shareholder entitled to vote in the election of Directors generally may nominate one or more persons for election as Directors at a meeting only if written notice of such shareholder's intention to make such nomination or nominations has been delivered personally to, or been mailed to and received by the Corporation at, the principal executive offices of the Corporation, addressed to the attention of the President, (a) with respect to an election to be held at an annual meeting that is called for a date that is within 30 days before or after the anniversary date of the immediately preceding annual meeting of shareholders, not less than 90 days nor more than 120 days prior to such anniversary date, and (b) with respect either to an election to be held at an annual meeting that is called for a date that is not within 30 days before or after the anniversary date of the immediately preceding annual meeting, or to a special meeting of shareholders called for the purpose of electing Directors, not later than the close of business on the tenth day following the day on which notice of the date of the meeting was mailed or public disclosure of the date of the meeting was made, whichever occurs first. Each such notice shall set forth: (i) the name and address of the shareholder intending to make the nomination and of the person or persons to be nominated; (ii) a representation that the shareholder is a holder of record of shares of the Corporation entitled to vote at such meeting and intends to appear in person or by proxy at the meeting to nominate the person or persons specified in the notice; (iii) a description of all arrangements or understandings between the shareholder and each nominee and any other person or persons (naming such person or persons) pursuant to which the nomination or nominations are to be made by the shareholder; (iv) such other information regarding each nominee proposed by such shareholder as would have been required to be included in a proxy statement filed pursuant to the proxy rules of the Securities and Exchange Commission had the nominee been nominated by the Board of Directors; and (v) the written consent of each nominee to serve as a Director of the Corporation if so elected. The presiding officer of the meeting may, in his or her sole discretion, declare invalid or refuse to acknowledge any nomination not made in compliance with the foregoing procedure.

ARTICLE 4 OFFICERS

Section 4.01. *Election and Office.* The Corporation shall have a Chairman of the Board, a Chief Executive Officer, a President, a Secretary and a Treasurer who, subject to Article SIXTH of the Articles of Incorporation, shall be elected by the Board of Directors. Subject to Article SIXTH of the Articles of Incorporation, the Board of Directors may create the positions of, define the powers and duties of and elect as additional officers one or more Vice Chairmen of the Board, one or more Vice Presidents, and one or more other officers or assistant officers. Any number of offices may be held by the same person. The Chairman of the Board and any Vice Chairman of the Board must be a Director of the Corporation. The initial officers of the Corporation (other than the Chairman of the Board) shall be selected by the Chief Executive Officer in consultation with the Chairman of the Board.

Section 4.02. Term. Each officer of the Corporation shall hold office until his successor is selected and qualified or until his earlier death, resignation or removal. Subject to Article SIXTH of the Articles of Incorporation, any officer may be removed by a vote of a majority of the Directors then in office. The terms of the Chairman of the Board and the Chief Executive Officer are fixed pursuant to Article SIXTH of the Articles of Incorporation.

Section 4.03. Powers and Duties of the Chairman of the Board. The Chairman of the Board shall have such powers and shall perform such duties as are provided in Article SIXTH of the Articles of Incorporation.

Section 4.04. Powers and Duties of the Chief Executive Officer. The Chief Executive Officer shall have such powers and shall perform such duties as are provided in Article SIXTH of the Articles of Incorporation.

Section 4.05 *Powers and Duties of the President.* The President shall have such powers and shall perform such duties as may, subject to Article SIXTH of the Articles of Incorporation, from time to time be assigned to the President by the Board of Directors.

Section 4.06. *Powers and Duties of the Secretary.* Unless otherwise determined by the Board of Directors, the Secretary shall be responsible for the keeping of the minutes of all meetings of the shareholders, the Board of Directors, and all committees of the Board, in books provided for that purpose, and for the giving and serving of all notices for the Corporation. The Secretary shall perform all other duties ordinarily incident to the office of Secretary and shall have such other powers and perform such other duties as may be assigned to the Secretary by the Board of Directors. The minute books of the Corporation may be held by a person other than the Secretary.

Section 4.07. Powers and Duties of the Treasurer. Unless otherwise determined by the Board of Directors, the Treasurer shall have charge of all the funds and securities of the Corporation. When necessary or proper, unless otherwise determined by the Board of Directors, the Treasurer shall endorse for collection on behalf of the Corporation checks, notes and other obligations, and shall deposit the same to the credit of the Corporation to such banks or depositories as the Board of Directors may designate and may sign all receipts and vouchers for payments made to the Corporation. The Treasurer shall be responsible for the regular entry in books of the Corporation to be kept for such purpose of a full and accurate account of all funds and securities received and paid by the Treasurer on account of the Corporation. Whenever required by the Board of Directors, the Treasurer shall render a statement of the financial condition of the Corporation. The Treasurer shall have such other powers and shall perform the duties as may be assigned to such officer from time to time by the Board of Directors. The Treasurer shall give such bond, if any, for the faithful performance of the duties of such office as shall be required by the Board of Directors.

Section 4.08. Powers and Duties of the Vice Chairmen, Vice Presidents and Assistant Officers. Unless otherwise determined by the Board of Directors and subject to Article SIXTH of the Articles of Incorporation, each Vice Chairman, Executive Vice President, Senior Vice President, Vice President and each assistant officer shall have the powers and perform the duties of his or her respective superior officer, except to the extent such powers and duties are limited by such superior officer or by the Board of Directors. Executive Vice Presidents, Senior Vice Presidents, Vice Presidents and assistant officers shall have such rank as may be designated by the Board of Directors, with Executive Vice Presidents serving as superior officers to Vice Presidents. Executive Vice Presidents, Senior Vice Presidents and Vice Presidents and Senior Vice Presidents serving as superior officers to Vice Presidents, in which event such Executive Vice Presidents, Senior Vice Presidents or Vice Presidents as having responsibility for a specific area of the Corporation's affairs, in which event such Executive Vice Presidents, Senior Vice Presidents or Vice Presidents shall be superior to the other Executive Vice Presidents, Senior Vice Presidents, Isenior Vice Presidents or Vice Presidents shall be the superior officer of the Executive Vice Presidents, vice Presidents, in relation to matters within his or her area. The President shall be the superior officer of the Executive Vice Presidents, Vice Presidents, Vice Presidents and all other officer positions created by the Board of Directors unless the Board of Directors provides otherwise. The Treasurer and Secretary shall be the superior officers of the Assistant Treasurers and Assistant Secretaries, respectively.

Section 4.09. Vacancies. Subject to Article SIXTH of the Articles of Incorporation, the Board of Directors shall have the power to fill any vacancies in any office occurring for any reason.

Section 4.10. Delegation of Office. Subject to Article SIXTH of the Articles of Incorporation, the Board of Directors may delegate the powers or duties of any officer of the Corporation to any other person from time to time.

ARTICLE 5 CAPITAL STOCK

Section 5.01. Share Certificates.

(a) Execution. Except as otherwise provided in Section 5.05, the shares of the Corporation shall be represented by certificates. Unless otherwise provided by the Board of Directors, every share certificate shall

be signed by two officers and sealed with the corporate seal, which may be a facsimile, engraved or printed, but where such certificate is signed by a transfer agent or a registrar, the signature of any corporate officer upon such certificate may be a facsimile, engraved or printed. In case any officer who has signed, or whose facsimile signature has been placed upon, any share certificate shall have ceased to be such officer because of death, resignation or otherwise, before the certificate is issued, it may be issued with the same effect as if the officer had not ceased to be such at the date of its issue. The provisions of this Section shall be subject to any inconsistent or contrary agreement at the time between the Corporation and any transfer agent or registrar.

(b) Designations, Voting Rights, Preferences, Limitations and Special Rights. To the extent the Corporation is authorized to issue shares of more than one class or series, every certificate shall set forth upon the face or back of the certificate (or shall state on the face or back of the certificate that the Corporation will furnish to any shareholder upon request and without charge) a full or summary statement of the designations, voting rights, preferences, limitations and special rights of the shares of each class or series authorized to be issued so far as they have been fixed and determined and the authority of the Board of Directors to fix and determine the designations, voting rights, preferences, limitations and special rights of the classes and series of shares of the Corporation.

(c) Fractional Shares. Except as otherwise determined by the Board of Directors, shares or certificates therefor may be issued as fractional shares for shares held by any dividend reinvestment plan or employee benefit plan created or approved by the Corporation's Board of Directors, but not by any other person.

Section 5.02. Transfer of Shares. Transfer of shares shall be made on the books of the Corporation only upon surrender of the share certificate, duly endorsed or with duly executed stock powers attached and otherwise in proper form for transfer, which certificate shall be canceled at the time of the transfer

Section 5.03. Determination of Shareholders of Record.

(a) Fixing Record Date. The Board of Directors of the Corporation may fix a time prior to the date of any meeting of shareholders as a record date for the determination of the shareholders entitled to notice of, or to vote at, the meeting, which time, except in the case of an adjourned meeting, shall be not more than 90 days prior to the date of the meeting of shareholders. Only shareholders of record on the date fixed shall be so entitled notwithstanding any transfer of shares on the books of the Corporation after any record date fixed as provided in this subsection. The Board of Directors may similarly fix a record date for the determination of shareholders of record for any other purpose. When a determination of shareholders of record has been made as provided in this Section 5.03 for purposes of a meeting, the determination shall apply to any adjournment thereof unless the Board of Directors fixes a new record date for the adjourned meeting.

(b) Determination when No Record Date Fixed. If a record date is not fixed:

(i) The record date for determining shareholders entitled to notice of or to vote at a meeting of shareholders shall be at the close of business on the day next preceding the day on which notice is given or, if notice is waived, at the close of business on the day immediately preceding the day on which the meeting is held.

(ii) The record date for determining shareholders for any other purpose shall be at the close of business on the day on which the Board of Directors adopts the resolution relating thereto.

(c) Certification by Nominee. The Board of Directors may adopt a procedure whereby a shareholder of the Corporation may certify in writing to the Corporation that all or a portion of the shares registered in the name of the shareholder are held for the account of a specified person or persons. The resolution of the Board of Directors may set forth:

- (i) the classification of shareholder who may certify;
- (ii) the purpose or purposes for which the certification may be made;
- (iii) the form of certification and information to be contained therein;

(iv) if the certification is with respect to a record date, the time after the record date within which the certification must be received by the Corporation; and

(v) such other provisions with respect to the procedure as are deemed necessary or desirable.

- - - - - - -

Upon receipt by the Corporation of a certification complying with the procedure, the persons specified in the certification shall be deemed, for the purposes set forth in the certification, to be the holders of record of the number of shares specified in place of the shareholder making the certification.

Section 5.04. Lost Share Certificates. Unless waived in whole or in part by the Board of Directors or any of the Chairman, any Vice Chairman, the President, any Senior Vice President, Secretary or Treasurer, unless the Board of Directors prohibits such waiver by such officer, any person requesting the issuance of a new certificate in lieu of an alleged lost, destroyed, mislaid or wrongfully taken certificate shall (a) give to the Corporation his or her bond of indemnity with an acceptable surety, and (b) satisfy such other requirements as may be imposed by the Corporation. Thereupon, a new share certificate shall be issued to the registered owner or his or her assigns in lieu of the alleged lost, destroyed, mislaid or wrongfully taken certificate; provided that the request therefor and issuance thereof have been made before the Corporation has notice that such shares have been acquired by a bona fide purchaser.

Section 5.05. Uncertificated Shares. Notwithstanding anything herein to the contrary, any or all classes and series of shares, or any part thereof, may be represented by uncertificated shares to the extent determined by the Board of Directors, except that shares represented by a certificate that is issued and outstanding shall continue to be represented thereby until the certificate is surrendered to the Corporation. Within a reasonable time after the issuance or transfer of uncertificated shares, the Corporation shall send to the registered owner thereof a written notice containing the information required to be set forth or stated on certificates. The rights and obligations of the holders of shares represented by certificates and the rights and obligations of the holders of shares represented by certificates and the rights and obligations of the holders of shares of the same class and series shall be identical. Notwithstanding anything herein to the contrary, the provisions of Section 5.02 shall be inapplicable to uncertificated shares and in lieu thereof the Board of Directors shall adopt alternative procedures for registration of transfers.

ARTICLE 6 NOTICES; COMPUTING TIME PERIODS

Section 6.01. *Contents of Notice.* Whenever any notice of a meeting of the Board of Directors or of shareholders is required to be given pursuant to these By-Laws or the Articles of Incorporation of the Corporation, as the same may be amended from time to time, or otherwise, the notice shall specify the geographic location, if any, date and time of the meeting; in the case of a special meeting of shareholders or where otherwise required by law or the By-Laws, the general nature of the business to be transacted at such meeting; and any other information required by law.

Section 6.02. Method of Notice. Any notice required to be given to any person under the provisions of the Articles of Incorporation or these By-Laws shall be given to the person either personally or by sending a copy thereof (i) by first class or express mail, postage prepaid, or courier service, charges prepaid, to such person's postal address appearing on the books of the Corporation, or, in the case of a Director, supplied by such Director to the Corporation for the purpose of notice or (ii) by facsimile transmission, e-mail or other electronic communication to such person's facsimile number or address for e-mail or other electronic communication supplied by such person to the Corporation for purposes of notice. Notice delivered pursuant to clause (i) of the preceding sentence shall be deemed to have been given to the person entitled thereto when deposited in the United States mail or with a courier service for delivery to that person, and notice pursuant to clause (ii) of the preceding sentence shall be deemed to have been given to the person estillated by the Board of Directors, notices of meetings may be given by, or at the direction of, the Secretary.

Section 6.03. Computing Time Periods.

(a) Days to be Counted. In computing the number of days for purposes of these By-Laws, all days shall be counted, including Saturdays, Sundays and any Holiday; *provided, however*, that if the final day of any time period falls on a Saturday, Sunday or Holiday, then the final day shall be deemed to be the next day which is not a Saturday, Sunday or Holiday. In computing the number of days for the purpose of giving notice of any meeting, the date upon which the notice is given shall be counted but the day set for the meeting shall not be counted.

(b) One Day Notice. In any case where only one day's notice is being given, notice must be given at least 24 hours in advance of the date and time specified for the meeting in question by delivery in person or by telephone, telex, telecopier or similar means of communication.

Section 6.04. Waiver of Notice. Whenever any notice is required to be given under the provisions of the Pennsylvania BCL or other applicable law or the Articles of Incorporation or these By-Laws, a waiver thereof in writing, signed by the person or persons entitled to the notice, whether before or after the time stated therein, shall be deemed equivalent to the giving of the notice. Except as otherwise required by law or the next sentence, neither the business to be transacted at, nor the purpose of, a meeting need be specified in the waiver of notice of the meeting. In the case of a special meeting of shareholders, the waiver of notice shall specify the general nature of the business to be transacted. Attendance of a person at any meeting shall constitute a waiver of notice of the meeting except where a person attends a meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business because the meeting was not lawfully called or convened.

Section 6.05. Modification of Proposal Contained in Notice. Whenever the language of a proposed resolution is included in a written notice of a meeting required to be given under the provisions of the Pennsylvania BCL or the Articles of Incorporation or these By-Laws, the meeting considering the resolution may without further notice adopt it with such clarifying or other amendments as do not enlarge its original purpose.

Section 6.06. Bulk Mail. Notice of any regular or special meeting of the shareholders, or any other notice required by the Pennsylvania BCL or by the Articles of Incorporation or these By-Laws to be given to all shareholders or to all holders of a class or a series of shares, may be given by any class of post-paid mail if the notice is deposited in the United States mail at least 20 days prior to the day named for the meeting or any corporate or shareholder action specified in the notice.

Section 6.07. Shareholders Without Forwarding Addresses. Notice or other communications need not be sent to any shareholder with whom the Corporation has been unable to communicate for more than 24 consecutive months because communications to the shareholder have been returned unclaimed or the shareholder has otherwise failed to provide the Corporation with a current address. Whenever the shareholder provides the Corporation with a current address, the corporation shall commence sending notices and other communications to the shareholder in the same manner as to other shareholders.

ARTICLE 7

LIMITATION OF DIRECTORS' LIABILITY AND INDEMNIFICATION OF DIRECTORS, OFFICERS AND OTHER PERSONS

Section 7.01. Limitation of Directors' Liability. No Director of the Corporation shall be personally liable for monetary damages as such for any action taken or any failure to take any action unless: (a) the Director has breached or failed to perform the duties of his or her office under Subchapter B of Chapter 17 of the Pennsylvania BCL (relating to standard of care and justifiable reliance), and (b) the breach or failure to perform constitutes self-dealing, wilful misconduct or recklessness; *provided, however*, that the provisions of this Section shall not apply to the responsibility or liability of a Director pursuant to any criminal statute, or to the liability of a Director for the payment of taxes pursuant to local, state or federal law.

Section 7.02. Indemnification and Insurance.

(a) Indemnification of Directors and Officers.

(i) Each Indemnitee (as defined below) shall be indemnified and held harmless by the Corporation for all actions taken by him or her and for all failures to take action (regardless of the date of any such action or failure to take action) to the fullest extent permitted by Pennsylvania law against all expense, liability and loss (including without limitation attorneys fees, judgments, fines, taxes, penalties, and amounts paid or to be paid in settlement) reasonably incurred or suffered by the Indemnitee in connection with any Proceeding (as defined below). No indemnification pursuant to this Section shall be made, however, in any case where the act or failure to act giving rise to the claim for indemnification is determined by a court to have constituted wilful misconduct or recklessness.

(ii) The right to indemnification provided in this Section shall include the right to have the expenses incurred by the Indemnitee in defending any Proceeding paid by the Corporation in advance of the final disposition of the Proceeding to the fullest extent permitted by Pennsylvania law; *provided* that, if Pennsylvania law continues so to require, the payment of such expenses incurred by the Indemnitee in advance of the final disposition of a Proceeding shall be made only upon delivery to the Corporation of an undertaking, by or on behalf of the Indemnitee, to repay all amounts so advanced without interest if it shall ultimately be determined that the Indemnitee is not entitled to be indemnified under this Section or otherwise.

(iii) To the extent that an Indemnitee has been successful on the merits or otherwise in defense of any Proceeding or in defense of any claim, issue or matter therein, the Corporation shall indemnify such person against expenses (including attorneys' fees) actually and reasonably incurred by such person in connection therewith.

(iv) Indemnification pursuant to this Section shall continue as to an Indemnitee who has ceased to be a Director or officer and shall inure to the benefit of his or her heirs, executors and administrators.

(v) For purposes of this Article, (A) "Indemnitee" shall mean each Director and each officer of the Corporation who was or is a party to, or is threatened to be made a party to, or is otherwise involved in, any Proceeding, by reason of the fact that he or she is or was a Director or officer of the Corporation or is or was serving in any capacity at the request or for the benefit of the Corporation as a Director, officer, employee, agent, partner, or fiduciary of, or in any other capacity for, another corporation or any partnership, joint venture, trust, employee benefit plan, or other enterprise; and (B) "Proceeding" shall mean any threatened, pending or completed action, suit or proceeding (including without limitation an action, suit or proceeding by or in the right of the Corporation), whether civil, criminal, administrative or investigative.

(b) Indemnification of Employees and Other Persons. The Corporation may, by action of its Board of Directors and to the extent provided in such action, indemnify employees and other persons, and provide for advancement of expenses to such persons in the manner set forth in (a)(ii), above, as though they were Indemnitees, except that, if Pennsylvania law continues to so require, to the extent that an employee or agent of the Corporation has been successful on the merits or otherwise in defense of any Proceeding or in defense of any claim, issue or matter therein, the Corporation shall indemnify such person against expenses (including attorneys' fees) actually and reasonably incurred by such person in connection therewith. Directors and officers of entities that have merged into, or have been consolidated with, or have been liquidated into, the Corporation shall not be Indemnitees with respect to Proceedings involving any action or failure to act of such Director or officer prior to the date of such merger, consolidation or liquidation, but such persons may be indemnified by the Board of Directors pursuant to the first sentence of this Section 7.02(b).

(c) Non-Exclusivity of Rights. The rights to indemnification and to the advancement of expenses provided in or pursuant to this Article shall not be exclusive of any other rights that any person may have or hereafter acquire under any statute, provision of the Articles of Incorporation or By-Laws, agreement, vote of shareholders or Directors, or otherwise.

(d) Insurance. The Corporation may purchase and maintain insurance, at its expense, for the benefit of any person on behalf of whom insurance is permitted to be purchased by Pennsylvania law against any expense, liability or loss, whether or not the Corporation would have the power to indemnify such person under Pennsylvania or other law. The Corporation may also purchase and maintain insurance to insure its indemnification obligations whether arising hereunder or otherwise.

(e) Fund For Payment of Expenses. The Corporation may create a fund of any nature, which may, but need not be, under the control of a trustee, or otherwise may secure in any manner its indemnification obligations, whether arising hereunder, under the Articles of Incorporation, by agreement, vote of shareholders or Directors, or otherwise.

Section 7.03. Amendment. The provisions of this Article 7 relating to the limitation of Directors' and officers' liability, to indemnification and to the advancement of expenses shall constitute a contract between the Corporation and each of its Directors and officers which may be modified as to any Director or officer only with that person's consent or as specifically provided in this Section. Notwithstanding any other provision of these By-Laws relating to their amendment generally, any repeal or amendment of this Article 7 which is adverse to any Director or officer shall apply to such Director or officer only on a prospective basis, and shall not reduce any limitation on the personal liability of a Director of the Corporation, or limit the rights of an Indemnifice to indemnification or to the advancement of expenses with respect to any action or failure to act occurring prior to the time of such repeal or amendment. Notwithstanding any other provision of these By-Laws, no repeal or amendment of these By-Laws shall affect any or all of this Article so as either to reduce the limitation of Directors' liability or limit indemnification or the advancement of expenses in any manner unless adopted by (a) the unanimous vote of the Directors of the Corporation then serving, or (b) the affirmative vote of shareholders entitled to cast at least eighty percent (80%) of the votes that all shareholders are entitled to cast in the election of Directors; *provided* that no such amendment shall have retroactive effect inconsistent with the preceding sentence.

Section 7.04. Changes in Pennsylvania Law. References in this Article to Pennsylvania law or to any provision thereof shall be to such law, as it existed on the date this Article was adopted or as such law thereafter may be changed; provided that (a) in the case of any change which expands the liability of Directors or limits the indemnification rights or the rights to advancement of expenses which the Corporation may provide, the rights to limited liability, to indemnification and to the advancement of expenses provided in this Article shall continue as theretofore to the extent permitted by law; and (b) if such change permits the Corporation without the requirement of any further action by shareholders or Directors to limit further the liability of Directors (or limit the liability of officers) or to provide broader indemnification rights to the advancement of expenses than the Corporation was permitted to provide prior to such change, then liability thereupon shall be so limited and the rights to indemnification and the advancement of expenses shall be so broadened to the extent permitted by law.

ARTICLE 8 FISCAL YEAR

Section 8.01. Determination of Fiscal Year. Determination of Fiscal Year. The Board of Directors shall have the power by resolution to fix the fiscal year of the Corporation. If the Board of Directors shall fail to do so, the Chief Executive Officer shall fix the fiscal year.

ARTICLE 9 ARTICLES OF INCORPORATION

Section 9.01. Inconsistent Provisions. In the event of any conflict between the provisions of these By-Laws and the provisions of the Articles of Incorporation, including, but not limited to, Article SIXTH of the Articles of Incorporation, the provisions of the Articles of Incorporation shall govern and control.



ARTICLE 10 AMENDMENTS

Section 10.01. Amendments. Except as otherwise provided in these By-Laws or in the Articles of Incorporation, including Article SIXTH, Article SEVENTH and Article TENTH of the Articles of Incorporation:

(a) Shareholders. The shareholders entitled to vote thereon shall have the power to alter, amend or repeal these By-Laws, by the vote of a majority of the votes cast at a duly organized meeting of shareholders by the holders of shares entitled to vote thereon, at any regular or special meeting, duly convened after notice to the shareholders of such purpose. In the case of a meeting of shareholders to amend or repeal these By-Laws, written notice shall be given to each shareholder that the purpose, or one of the purposes, of the meeting is to consider the adoption, amendment or repeal of the By-Laws.

(b) Board of Directors. The Board of Directors (but not a committee thereof) shall have the power to alter, amend and repeal these By-Laws, regardless of whether the shareholders have previously adopted the By-Law being amended or repealed, subject to the power of the shareholders to change such action; *provided, however*, that the Board of Directors shall not have the power to amend these By-Laws on any subject that is expressly committed to the shareholders by the express terms hereof, by the Pennsylvania BCL or otherwise.

ARTICLE 11

INTERPRETATION OF BY-LAWS; SEPARABILITY

Section 11.01. Interpretation. All words, terms and provisions of these By-Laws shall be interpreted and defined by and in accordance with the Pennsylvania BCL.

Section 11.02. Separability. The provisions of these By-Laws are independent of and separable from each other, and no provision shall be affected or rendered invalid or unenforceable by virtue of the fact that for any reason any other or others of them may be invalid or unenforceable in whole or in part.

ARTICLE 12 DETERMINATIONS BY THE BOARD

Section 12.01. Effect of Board Determinations. Any determination involving interpretation or application of these By-Laws made in good faith by the Board of Directors shall be final, binding and conclusive on all parties in interest.

<u>OuickLinks</u> — Click here to rapidly navigate through this document

Exhibit 10.4

COMCAST CORPORATION 2002 DEFERRED COMPENSATION PLAN

ARTICLE 1-COVERAGE OF PLAN

1.1. Background, Continuation and Freeze of Plan.

(a) Comcast Corporation, a Pennsylvania corporation, hereby amends and restates the Comcast Corporation 2002 Deferred Compensation Plan (the "*Plan*"), effective February 24, 2004. The Plan was initially adopted effective February 12, 1974 and was amended and restated effective August 15, 1996, June 21, 1999, December 19, 2000, October 26, 2001, April 29, 2002, July 9, 2002, November 18, 2002, March 3, 2003, December 1, 2003, January 30, 2004 and February 24, 2004.

(b) In order to preserve the favorable tax treatment available to deferrals that were made under the Plan before January 1, 2005 in light of the American Jobs Creation Act of 2004 and the regulations issued by the Department of the Treasury thereunder (the "AJCA"), no Compensation may be deferred under the Plan pursuant to an Initial Election after December 31, 2004, other than amounts that (i) were subject to an Initial Election before January 1, 2005, (ii) would, but for such Initial Election, have been paid in 2005 and (iii) are treated as earned and vested as of December 31, 2004 under *IRS Notice 2005–1*.

(c) Amounts earned and vested prior to January 1, 2005 are and will remain subject to the terms and conditions of the Plan

1.2. Plan Unfunded and Limited to Outside Directors and Select Group of Management or Highly Compensated Employees. The Plan is unfunded and is maintained primarily for the purpose of providing outside directors and a select group of management or highly compensated employees the opportunity to defer the receipt of compensation otherwise payable to such outside directors and eligible employees in accordance with the terms of the Plan.

ARTICLE 2—DEFINITIONS

2.1. "Account" means the bookkeeping accounts established pursuant to Section 5.1 and maintained by the Administrator in the names of the respective Participants, to which all amounts deferred and earnings allocated under the Plan shall be credited, and from which all amounts distributed pursuant to the Plan shall be debited.

- 2.2. "Active Participant" means:
 - (a) Each Participant who is in active service as an Outside Director; and
 - (b) Each Participant who is actively employed by a Participating Company as an Eligible Employee.
- 2.3. "Administrator" means the Committee.

2.4. "Affiliate" means, with respect to any Person, any other Person that, directly or indirectly, is in control of, is controlled by, or is under common control with, such Person. For purposes of this definition, the term "control," including its correlative terms "controlled by" and "under common control with," mean, with respect to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract or otherwise.

2.5. "Annual Rate of Pay" means, as of any date, an employee's annualized base pay rate. An employee's Annual Rate of Pay shall not include sales commissions or other similar payments or awards.

2.6. "Applicable Interest Rate" means:

(a) Except as otherwise provided in Sections 2.6(b) or (c), the Applicable Interest Rate means the interest rate that, when compounded daily pursuant to rules established by the Administrator from time to time, is mathematically equivalent to 12% per annum, compounded annually.

(b) Except to the extent otherwise required by Section 10.2, effective for the period beginning as soon as administratively practicable following a Participant's employment termination date to the date

the Participant's Account is distributed in full, the Administrator, in its sole discretion, may designate the term "Applicable Interest Rate" for such Participant's Account to mean the lesser of (i) the rate in effect under Section 2.6(a) or (ii) the Prime Rate plus one percent. Notwithstanding the foregoing, the Administrator may delegate its authority to determine the Applicable Interest Rate under this Section 2.6(b) to an officer of the Company or committee of two or more officers of the Company.

(c) Except to the extent otherwise required by Section 10.2, the Applicable Interest Rate for Severance Pay deferred pursuant to Article 3 shall be determined by the Administrator, in its sole discretion, provided that the Applicable Interest Rate shall not be less than the lower of the Prime Rate or LIBOR, nor more than the rate specified in Section 2.6(a). Notwithstanding the foregoing, the Administrator may delegate its authority to determine the Applicable Interest Rate under this Section 2.6(c) to an officer of the Company.

2.7. "Beneficiary" means such person or persons or legal entity or entities, including, but not limited to, an organization exempt from federal income tax under section 501(c)(3) of the Code, designated by a Participant or Beneficiary to receive benefits pursuant to the terms of the Plan after such Participant's or Beneficiary's death. If no Beneficiary is designated by the Participant or Beneficiary, or if no Beneficiary survives the Participant or Beneficiary (as the case may be), the Participant's Beneficiary shall be the Participant's Surviving Spouse if the Participant has a Surviving Spouse and otherwise the Participant's estate, and the Beneficiary of a Beneficiary shall be the Beneficiary's Surviving Spouse if the Beneficiary has a Surviving Spouse and otherwise the Beneficiary's estate.

2.8. "Board" means the Board of Directors of the Company.

2.9. "CCCHI" means Comcast Cable Communications Holdings, Inc., formerly known as AT&T Broadband Corp.

2.10. "Change of Control" means any transaction or series of transactions as a result of which any Person who was a Third Party immediately before such transaction or series of transactions owns then-outstanding securities of the Company such that such Person has the ability to direct the management of the Company, as determined by the Board in its discretion. The Board may also determine that a Change of Control shall occur upon the completion of one or more proposed transactions. The Board's determination shall be final and binding.

2.11. "CHC" means Comcast Holdings Corporation, formerly known as Comcast Corporation.

2.12. "Code" means the Internal Revenue Code of 1986, as amended.

2.13. "Committee" means the Compensation Committee of the Board of Directors of the Company.

2.14. "Company" means Concast Corporation, a Pennsylvania corporation, as successor to CHC, including any successor thereto by merger, consolidation, acquisition of all or substantially all the assets thereof, or otherwise.

2.15. "Company Stock" means:

(a) except as provided in Section 2.15(b), Comcast Corporation Class A Special Common Stock, par value, \$0.01, including a fractional share; and

(b) with respect to amounts credited to the Company Stock Fund pursuant to deferral elections by Outside Directors made pursuant to Section 3.1(a), Comcast Corporation Class A Common Stock, par value \$0.01, including a fractional share;

and such other securities issued by Comcast Corporation as may be subject to adjustment in the event that shares of either class of Company Stock are changed into, or exchanged for, a different number or kind of shares of stock or other securities of the Company, whether through merger, consolidation, reorganization, recapitalization, stock dividend, stock split-up or other substitution of securities of the Company. In such event, the Committee shall make appropriate equitable anti-dilution adjustments to the number and class of hypothetical shares of Company Stock credited to Participants' Accounts under the Company Stock Fund. Any reference to the term "Company Stock" in the Plan shall be a reference to the appropriate number and

class of shares of stock as adjusted pursuant to this Section 2.15. The Committee's adjustment shall be effective and binding for all purposes of the Plan.

2.16. "Company Stock Fund" means a hypothetical investment fund pursuant to which income, gains and losses are credited to a Participant's Account as if the Account, to the extent deemed invested in the Company Stock Fund, were invested in hypothetical shares of Company Stock, and all dividends and other distributions paid with respect to Company Stock were held uninvested in cash, and reinvested in additional hypothetical shares of Company Stock as of the next succeeding December 31 (to the extent the Account continues to be deemed invested in the Company Stock Fund through such December 31), based on the Fair Market Value of the Company Stock for such December 31.

2.17. "Compensation" means:

(a) In the case of an Outside Director, the total remuneration payable in cash or payable in Company Stock (as elected by the Outside Director pursuant to the Comcast Corporation 2003 Director Compensation Plan) for services as a member of the Board and as a member of any Committee of the Board; and

(b) In the case of an Eligible Employee, the total cash remuneration for services payable by a Participating Company, excluding sales commissions or other similar payments or awards.

2.18. "Death Tax Clearance Date" means the date upon which a Deceased Participant's or a deceased Beneficiary's Personal Representative certifies to the Administrator that (i) such Deceased Participant's or deceased Beneficiary's Death Taxes have been finally determined, (ii) all of such Deceased Participant's or deceased Beneficiary's Account have been paid in full and (iii) all potential liability for Death Taxes with respect to the Deceased Participant's or deceased Beneficiary's Account has been satisfied.

2.19. "Death Taxes" means any and all estate, inheritance, generation-skipping transfer, and other death taxes as well as any interest and penalties thereon imposed by any governmental entity (a "taxing authority") as a result of the death of the Participant or the Participant's Beneficiary.

2.20. "Deceased Participant" means a Participant whose employment, or, in the case of a Participant who was an Outside Director, a Participant whose service as an Outside Director, is terminated by death.

2.21. "Disabled Participant" means:

(a) A Participant whose employment or, in the case of a Participant who is an Outside Director, a Participant whose service as an Outside Director, is terminated by reason of disability;

(b) The duly-appointed legal guardian of an individual described in Section 2.21(a) acting on behalf of such individual.

2.22. "Eligible Employee" means:

(a) Each employee of a Participating Company who, as of December 31, 1989, was eligible to participate in the Prior Plan.

(b) Each employee of a Participating Company who was, at any time before January 1, 1995, eligible to participate in the Prior Plan and whose Annual Rate of Pay is \$90,000 or more as of both (i) the date on which an Initial Election is filed with the Administrator and (ii) the first day of each calendar year beginning after December 31, 1994.

(c) Each individual who was an employee of an entity that was a Participating Company in the Plan as of June 30, 2002 and who has an Annual Rate of Pay of \$125,000 as of each of (i) June 30, 2002; (ii) the date on which an Initial Election is filed with the Administrator and (iii) the first day of each calendar year beginning after December 31, 2002.

(d) Each employee of a Participating Company whose Annual Rate of Pay is \$200,000 or more as of both (i) the date on which an Initial Election is filed with the Administrator and (ii) the first day of the calendar year in which such Initial Election is filed.

(e) Each New Key Employee.

(f) Each employee of a Participating Company who (i) as of December 31, 2002, was an "*Eligible Employee*" within the meaning of Section 2.34 of the AT&T Broadband Deferred Compensation Plan (as amended and restated, effective November 18, 2002) with respect to whom an account was maintained, and (ii) for the period beginning on December 31, 2002 and extending through any date of determination, has been actively and continuously in service to the Company or an Affiliate.

(g) Each other employee of a Participating Company who is designated by the Committee, in its discretion, as an Eligible Employee.

2.23. "Fair Market Value"

(a) If shares of Company Stock are listed on a stock exchange, Fair Market Value shall be determined based on the last reported sale price of a share on the principal exchange on which shares are listed on the date of determination, or if such date is not a trading day, the next trading date.

(b) If shares of Company Stock are not so listed, but trades of shares are reported on the Nasdaq National Market, Fair Market Value shall be determined based on the last quoted sale price of a share on the Nasdaq National Market on the date of determination, or if such date is not a trading day, the next trading date.

(c) If shares of Company Stock are not so listed nor trades of shares so reported, Fair Market Value shall be determined by the Committee in good faith.

2.24. "Former Eligible Employee" means an employee of a Participating Company who, as of any relevant date, does not satisfy the requirements of an "Eligible Employee" but who previously met such requirements under the Plan or the Prior Plan.

2.25. "Grandfathered Participant" means an Inactive Participant who, on or before December 31, 1991, entered into a written agreement with the Company to terminate service to the Company or gives written notice of intention to terminate service to the Company, regardless of the actual date of termination of service.

2.26. "Hardship" means a Participant's severe financial hardship due to an unforeseeable emergency resulting from a sudden and unexpected illness or accident of the Participant, or, a sudden and unexpected illness or accident of a dependent (as defined by section 152(a) of the Code) of the Participant, or loss of the Participant's property due to casualty, or other similar and extraordinary unforeseeable circumstances arising as a result of events beyond the control of the Participant. A need to send the Participant's child to college or a desire to purchase a home is not an unforesceable emergency. No Hardship shall be deemed to exist to the extent that the financial hardship is or may be relieved (a) through reimbursement or compensation by insurance or otherwise, (b) by borrowing from commercial sources on reasonable commercial terms to the extent that this borrowing would not itself cause a severe financial hardship, (c) by cessation of deferrals under the Plan, or (d) by liquidation of the Participant's other assets (including assets of the Participant's spouse and minor children that are reasonably available to the Participant) to the extent that this liquidation would not itself cause severe financial hardship. For the purposes of the preceding sentence, the Participant's resources shall be deemed to include those assets of his spouse and minor children that are reasonably available to the Participant; however, property held for the Participant's child under an irrevocable trust or under a Uniform Gifts to Minors Act custodianship or Uniform Transfers to Minors Act custodianship shall not be treated as a resource of the Participant. The Board shall determine whether the circumstances of the Participant constitute an unforeseeable emergency and thus a Hardship within the meaning of this Section. Following a uniform procedure, the Board's determination shall consider any facts or conditions deemed necessary or advisable by the Board, and the Participant shall be required to submit any evidence of the Participant's circumstances that the Board requires. The determination as to whether the Participant's circumstances are a case of Hardship shall be based on the facts of each case; provided however, that all determinations as to Hardship shall be uniformly and consistently made according to the provisions of this Section for all Participants in similar circumstances.

2.27. "Inactive Participant" means each Participant (other than a Retired Participant, Deceased Participant or Disabled Participant) who is not in active service as an Outside Director and is not actively employed by a Participating Company.

2.28. "Income Fund" means a hypothetical investment fund pursuant to which income, gains and losses are credited to a Participant's Account as if the Account, to the extent deemed invested in the Income Fund, were credited with interest at the Applicable Interest Rate.

2.29. "Initial Election" means a written election on a form provided by the Administrator, filed with the Administrator in accordance with Article 3, pursuant to which an Outside Director or an Eligible Employee may:

(a) Elect to defer all or any portion of the Compensation payable for the performance of services as an Outside Director or as an Eligible Employee (including Severance Pay, to the extent permitted with respect to an Eligible Employee pursuant to Section 3.2) following the time that such election is filed; and

(b) Designate the time of payment of the amount of deferred Compensation to which the Initial Election relates.

2.30. "Insider" means an Eligible Employee or Outside Director who is subject to the short-swing profit recapture rules of section 16(b) of the Securities Exchange Act of 1934, as amended.

2.31. "LIBOR" means, for any calendar year, the interest rate that, when compounded daily pursuant to rules established by the Administrator from time to time, is mathematically equivalent to the annual London Inter Bank Offered Rate (compounded annually), as published in the Eastern Edition of *The Wall Street Journal*, on the last business day preceding the first day of such calendar year, and as adjusted as of the last business day preceding the first day of each calendar year beginning thereafter.

2.32. "New Key Employee" means each employee of a Participating Company:

(a) who becomes an employee of a Participating Company and has an Annual Rate of Pay of \$200,000 or more as of his employment commencement date, or

(b) who has an Annual Rate of Pay that is increased to \$200,000 or more and who, immediately preceding such increase, was not an Eligible Employee.

2.33. "Normal Retirement" means:

(a) For a Participant who is an employee of a Participating Company immediately preceding his termination of employment, a termination of employment that is treated by the Participating Company as a retirement under its employment policies and practices as in effect from time to time; and

- (b) For a Participant who is an Outside Director immediately preceding his termination of service, his normal retirement from the Board.
- 2.34. "Outside Director" means a member of the Board, who is not an employee of a Participating Company.

2.35. "Participant" means each individual who has made an Initial Election, or for whom an Account is established pursuant to Section 5.1, and who has an undistributed amount credited to an Account under the Plan, including an Active Participant, a Deceased Participant and an Inactive Participant.

2.36. "Participating Company" means:

- (a) The Company;
- (b) CHC;
- (c) Comcast Cable Communications, LLC, and its subsidiaries;
- (d) Comcast International Holdings, Inc.;
- (e) Comcast Online Communications, Inc.;

- (f) Comcast Business Communications, Inc.;
- (g) CCCHI and its subsidiaries;

(h) Comcast Shared Services Corporation ("CSSC"), to the extent individual employees of CSSC or groups of CSSC employees, categorized by their secondment, are designated as eligible to participate by the Committee or its delegate; and

(i) Any other entities that are subsidiaries of the Company as designated by the Committee in its sole discretion.

2.37. "Person" means an individual, a corporation, a partnership, an association, a trust or any other entity or organization.

2.38. "Plan" means the Comcast Corporation 2002 Deferred Compensation Plan, as set forth herein, and as amended from time to time.

2.39. "Prime Rate" means, for any calendar year, the interest rate that, when compounded daily pursuant to rules established by the Administrator from time to time, is mathematically equivalent to the prime rate of interest (compounded annually) as published in the Eastern Edition of *The Wall Street Journal* on the last business day preceding the first day of such calendar year, and as adjusted as of the last business day preceding the first day of each calendar year beginning thereafter.

2.40. "Prior Plan" means the Comcast Corporation 1996 Deferred Compensation Plan, as in effect immediately preceding the amendment, restatement and renaming of the Plan as the Comcast Corporation 2002 Deferred Compensation Plan.

2.41. "Retired Participant" means a Participant who has terminated service pursuant to a Normal Retirement.

2.42. "Severance Pay" means any amount that is payable in cash and is identified by a Participating Company as severance pay, or any amount which is payable on account of periods beginning after the last date on which an employee (or former employee) is required to report for work for a Participating Company.

2.43. "Subsequent Election" means a written election on a form provided by the Administrator, filed with the Administrator in accordance with Article 3, pursuant to which a Participant or Beneficiary may elect to defer (or, in limited cases, accelerate) the time of payment or to change the manner of payment of amounts previously deferred in accordance with the terms of a previously made Initial Election or Subsequent Election.

2.44. "Surviving Spouse" means the widow or widower, as the case may be, of a Deceased Participant or a Deceased Beneficiary (as applicable).

- 2.45. "Terminating Event" means either of the following events:
 - (a) the liquidation of the Company; or
 - (b) a Change of Control.

2.46. "Third Party" means any Person, together with such Person's Affiliates, provided that the term "Third Party" shall not include the Company or an Affiliate of the Company.

ARTICLE 3-INITIAL AND SUBSEQUENT ELECTIONS

3.1.

Elections.

(a) Initial Elections. Each Outside Director and Eligible Employee shall have the right to defer all or any portion of the Compensation (including bonuses, if any, and, in the case of Outside Directors, including any portion of an Outside Director's Compensation payable in the form of Company Stock) that he would otherwise be entitled to receive in a calendar year by filing an Initial Election at the time and in the manner described in this Article 3; provided that Severance Pay shall be included as "Compensation" for purposes of this Section 3.1 only to the extent permitted, and subject to such rules regarding the length of any initial deferral period and subsequent deferral period, if any, established by the Administrator in its sole discretion. The Compensation of such Outside Director or Eligible Employee for a calendar year shall be reduced in an amount equal to the portion of the Compensation deferred by such Outside Director or Eligible Employee for such calendar year pursuant to such Outside Director's or Eligible Employee's Initial Election. Such reduction shall be effected on a pro rata basis from each periodic installment payment of such Outside Director's or Eligible Employee's Compensation for the calendar year (in accordance with the general pay practices of the Participating Company), and credited, as a bookkeeping entry, to such Outside Director's or Eligible Employee's Account in accordance with Section 5.1. Amounts credited to the Accounts of Outside Directors in the form of Company Stock shall be credited to the Company Stock Fund and credited with income, gains and losses in accordance with Section 5.2(c).

(b) Subsequent Elections. Each Participant or Beneficiary shall have the right to elect to defer (or, in limited cases, accelerate) the time of payment or to change the manner of payment of amounts previously deferred in accordance with the terms of a previously made Initial Election pursuant to the terms of the Plan by filing a Subsequent Election at the time, to the extent, and in the manner described in this Article 3.

3.2. Filing of Initial Election: General. An Initial Election shall be made on the form provided by the Administrator for this purpose. Except as provided in Section 3.3, no such Initial Election shall be effective unless it is filed with the Administrator on or before December 31 of the calendar year preceding the calendar year to which the Initial Election applies; provided that an Initial Election with respect to Severance Pay shall not be effective unless it is filed within 30 days following the date of written notification to an Eligible Employee from the Administrator or its duly authorized delegate of such Eligible Employee's eligibility to defer Severance Pay.

3.3. Filing of Initial Election by New Key Employees and New Outside Directors.

(a) New Key Employees. Notwithstanding Section 3.1 and Section 3.2, a New Key Employee may elect to defer all or any portion of his Compensation that he would otherwise be entitled to receive in the calendar year in which the New Key Employee was employed, beginning with the payroll period next following the filing of an Initial Election with the Administrator and before the close of such calendar year by making and filing the Initial Election with the Administrator within 60 days of such New Key Employee's date of hire or within 60 days of the date such New Key Employee first becomes eligible to participate in the Plan. Any Initial Election by such New Key Employee for succeeding calendar years shall be made in accordance with Section 3.1 and Section 3.2.

(b) New Outside Directors. Notwithstanding Section 3.1 and Section 3.2, an Outside Director may elect to defer all or any portion of his Compensation that he would otherwise be entitled to receive in the calendar year in which an Outside Director's election as a member of the Board becomes effective (provided that such Outside Director is not a member of the Board immediately preceding such effective date), beginning with Compensation payable following the filing of an Initial Election with the Administrator and before the close of such calendar year by making and filing the Initial Election with the Administrator within 60 days of the effective date of such Outside Director's election. Any Initial Election by such Outside Director for succeeding calendar years shall be made in accordance with Section 3.1 and Section 3.2



3.4. Calendar Years to which Initial Election May Apply. A separate Initial Election may be made for each calendar year as to which an Outside Director or Eligible Employee desires to defer all or any portion of such Outside Director's or Eligible Employee's Compensation. The failure of an Outside Director or Eligible Employee to make an Initial Election for any calendar year shall not affect such Outside Director's or Eligible Employee's right to make an Initial Election for any other calendar year.

(a) Initial Election of Distribution Date. Each Outside Director or Eligible Employee shall, contemporaneously with an Initial Election, also elect the time of payment of the amount of the deferred Compensation to which such Initial Election relates; provided, however, that, subject to acceleration pursuant to Section 3.5(e) or (f), Section 3.7, Section 7.1, 7.2, or Article 8, no distribution may commence earlier than January 2nd of the second calendar year beginning after the date the Initial Election is filed with the Administrator, nor later than January 2nd of the eleventh calendar year beginning after the date the Initial Election is filed with the Administrator. Further, each Outside Director or Eligible Employee may select with each Initial Election the manner of distribution in accordance with Article 4.

3.5. Subsequent Elections.

(a) Active Participants. Each Active Participant, who has made an Initial Election, or who has made a Subsequent Election, may elect to change the manner of distribution or defer the time of payment of any part or all of such Participant's Account for a minimum of two and a maximum of ten additional years from the previously-elected payment date, by filing a Subsequent Election with the Administrator on or before the close of business on June 30 of the calendar year preceding the calendar year in which the lump-sum distribution or initial installment payment would otherwise be made. The number of Subsequent Elections under this Section 3.5(a) shall not be limited.

(b) Inactive Participants. The Committee may, in its sole and absolute discretion, permit an Inactive Participant to make a Subsequent Election to change the manner of distribution, or defer the time of payment of any part or all of such Inactive Participant's Account for a minimum of two years and a maximum of ten additional years from the previously-elected payment date, by filing a Subsequent Election with the Administrator on or before the close of business on June 30 of the calendar year preceding the calendar year in which the lump-sum distribution or initial installment payment would otherwise be made. The number of Subsequent Elections under this Section 3.5(b) shall be determined by the Committee in its sole and absolute discretion.

(c) Surviving Spouses.

.

(i) General Rule. A Surviving Spouse who is a Deceased Participant's Beneficiary may elect to change the manner of distribution, or defer the time of payment, of any part or all of such Deceased Participant's Account the payment of which would be made neither within six (6) months after, nor within the calendar year of, the date of such election. Such election shall be made by filing a Subsequent Election with the Administrator in which the Surviving Spouse shall specify the change in the manner of distribution or the change in the time of payment, which shall be no less than two nor more than ten years from the previously-elected payment date, or such Surviving Spouse may elect to defer payment until such Surviving Spouse's death. A Surviving Spouse may make a total of two (2) Subsequent Elections under this Section 3.5(c)(i), with respect to all or any part of the Deceased Participant's Account. Subsequent Elections pursuant to this Section 3.5(c)(i) may specify different changes with respect to different parts of the Deceased Participant's Account.

(ii) Exception. Notwithstanding the above Section 3.5(c)(i), a Subsequent Election may be made by a Surviving Spouse within sixty (60) days of the Deceased Participant's death; provided, however, such election may only be made with respect to amounts which would not be paid under the Deceased Participant's election as in effect on the date of the Deceased Participant's death until a date which is at least six (6) months from the Deceased Participant's date of death. Such election shall be made by filing a Subsequent Election with the Administrator in which the Surviving Spouse shall specify the change in the manner of distribution or the change in the time of payment, which

shall be no less than two (2) nor more than ten (10) years from the previously-elected payment date, or such Surviving Spouse may elect to defer payment until such Surviving Spouse's death. A Surviving Spouse may only make one (1) Subsequent Election under this Section 3.5(c)(ii) with respect to all or any part of the Deceased Participant's Account. Such Surviving Spouse may, however, make one additional Subsequent Election under Section 3.5(c)(i) in accordance with the terms of Section 3.5(c)(i). The one (1) Subsequent Election permitted under this Section 3.5(c)(ii) may specify different changes for different parts of the Deceased Participant's Account.

(d) Beneficiary of a Deceased Participant Other Than a Surviving Spouse.

(i) General Rule. A Beneficiary of a Deceased Participant (other than a Surviving Spouse) may elect to change the manner of distribution, or defer the time of payment, of any part or all of such Deceased Participant's Account the payment of which would be made neither within six (6) months after, nor within the calendar year of, the date of such election. Such election shall be made by filing a Subsequent Election with the Administrator in which the Beneficiary shall specify the change in the manner of distribution or the change in the time of payment, which shall be no less than two (2) nor more than ten (10) years from the previously-elected payment date. A Beneficiary may make one (1) Subsequent Election under this Section 3.5(d)(i), with respect to all or any part of the Deceased Participant's Account. Subsequent Elections pursuant to this Section 3.5(d)(i) may specify different changes for different parts of the Deceased Participant's Account.

(ii) Exception. Notwithstanding the above Section 3.5(d)(i), a Subsequent Election may be made by a Beneficiary within sixty (60) days of the Deceased Participant's death; provided, however, such election may only be made with respect to amounts which would not be paid under the Deceased Participant's election as in effect on the date of the Deceased Participant's death until a date which is at least six (6) months from the Deceased Participant's date of death. Such election shall be made by filing a Subsequent Election with the Administrator in which the Beneficiary shall specify the change in the manner of distribution or the change in the time of payment, which shall be no less than two (2) nor more than ten (10) years from the previously-elected payment date. A Beneficiary may make one (1) Subsequent Election under this Section 3.5(d)(ii) with respect to all or any part of the Deceased Participant's Account. Subsequent Elections pursuant to this Section 3.5(d)(ii) may specify different changes for different parts of the Deceased Participant's Account.

(e) Other Deferral and Acceleration by a Beneficiary. Any Beneficiary (other than a Surviving Spouse who has made a Subsequent Election under Section 3.5(c) or a Beneficiary who has made a Subsequent Election under Section 3.5(d)) may elect to change the manner of distribution from the manner of distribution in which payment of a Deceased Participant's Account would otherwise be made, and

(i) Defer the time of payment of any part or all of the Deceased Participant's Account or deceased Beneficiary's Account for one additional year from the date a payment would otherwise be made or begin (provided that if a Subsequent Election is made pursuant to this Section 3.5(e)(i), the Deceased Participant's Account or deceased Beneficiary's Account shall be in all events distributed in full on or before the fifth anniversary of the Deceased Participant's or a deceased Beneficiary's death); or

(ii) Accelerate the time of payment of a Deceased Participant's Account or deceased Beneficiary's Account from the date or dates that payment would otherwise be made or begin to the date that is the later of (A) six (6) months after the date of the Deceased Participant's or deceased Beneficiary's death and (B) January 2nd of the calendar year beginning after the Deceased Participant's or deceased Beneficiary's death, provided that if a Subsequent Election is made pursuant to this Section 3.5(e)(ii), the Deceased Participant's Account or deceased Beneficiary's Account shall be distributed in full on such accelerated payment date.

A Subsequent Election pursuant to this Section 3.5(e) must be filed with the Administrator within one hundred and twenty (120) days following the Deceased Participant's or deceased Beneficiary's death. One and only one Subsequent Election shall be permitted pursuant to this Section 3.5(e) with respect to a Deceased Participant's Account or deceased Beneficiary's Account, although if such Subsequent Election is filed pursuant to Section 3.5(e)(i), it may specify different changes for different parts of the Account.

(f) Disabled Participant. A Disabled Participant (who has not been permitted to make a Subsequent Election under Section 3.5(h)) may elect to change the form of distribution from the form of distribution that the payment of the Disabled Participant's Account would otherwise be made and may elect to accelerate the time of payment of the Disabled Participant's Account from the date payment would otherwise be made to January 2nd of the calendar year beginning after the Participant became disabled. A Subsequent Election pursuant to this Section 3.5(f) must be filed with the Administrator on or before the close of business on the later of (i) the June 30 following the date the Participant becomes a Disabled Participant on or before May 1 of a calendar year; (ii) the 60th day following the date the Participant becomes a Disabled Participant after May 1 and before November 2 of a calendar year or (iii) the December 31 following the date the Participant after November 1 of a calendar year.

(g) Retired Participant. A Retired Participant (who has not been permitted to make a Subsequent Election under Section 3.5(h)) may elect to change the form of distribution from the form of distribution that payment of the Retired Participant's Account would otherwise be made and may elect to defer the time of payment of the Retired Participant's Account for a minimum of two additional years from the date payment would otherwise be made (provided that if a Subsequent Election is made pursuant to this Section 3.5(g), the Retired Participant's Account shall be distributed in full on or before the fifth anniversary of the Retired Participant's Normal Retirement). A Subsequent Election pursuant to this Section 3.5(g) must be filed with the Administrator on or before the close of business on the later of (i) the June 30 following the Participant's Normal Retirement after May 1 and before November 2 of a calendar year or (ii) the December 31 following the Participant's Normal Retirement after November 1 of a calendar year.

(h) Retired Participants and Disabled Participants. The Committee may, in its sole and absolute discretion, permit a Retired Participant or a Disabled Participant to make a Subsequent Election to change the form of distribution that the payment of the Retired Participant's account would otherwise be made or to defer the time of payment of any part or all of such Retired or Disabled Participant's Account for a minimum of two years and a maximum of ten additional years from the previously-elected payment date, by filing a Subsequent Election with the Administrator on or before the close of business on June 30 of the calendar year preceding the calendar year in which the lump-sum distribution or initial installment payment would otherwise be made. The number of Subsequent Elections under this Section 3.5(h) shall be determined by the Committee in its sole and absolute discretion.

(i) Most Recently Filed Initial Election or Subsequent Election Controlling. Subject to acceleration pursuant to Section 3.5(e) or 3.5(f), Section 3.7 or Section 7.1, no distribution of the amounts deferred by a Participant for any calendar year shall be made before the payment date designated by the Participant or Beneficiary on the most recently filed Initial Election or Subsequent Election with respect to each deferred amount.

3.6. Distribution in Full Upon Terminating Event. The Company shall give Participants at least thirty (30) days notice (or, if not practicable, such shorter notice as may be reasonably practicable) prior to the anticipated date of the consummation of a Terminating Event. The Committee may, in its discretion, provide in such notice that notwithstanding any other provision of the Plan or the terms of any Initial Election or Subsequent Election, upon the consummation of a Terminating Event, the Account balance of each Participant shall be distributed in full and any outstanding Initial Elections or Subsequent Elections shall be revoked.

3.7. Withholding and Payment of Death Taxes.

(a) Notwithstanding any other provisions of this Plan to the contrary, including but not limited to the provisions of Article 3 and Article 7, or any Initial or Subsequent Election filed by a Deceased Participant or a Deceased Participant's Beneficiary (for purposes of this Section, the *"Decedent"*), the Administrator shall apply the terms of Section 3.7(b) to the Decedent's Account unless the Decedent affirmatively has elected, in writing, filed with the Administrator, to waive the application of Section 3.7(b).

(b) Unless the Decedent affirmatively has elected, pursuant to Section 3.7(a), that the terms of this Section 3.7(b) not apply:

(i) The Administrator shall prohibit the Decedent's Beneficiary from taking any action under any of the provisions of the Plan with regard to the Decedent's Account other than the Beneficiary's making of a Subsequent Election pursuant to Section 3.5;

(ii) The Administrator shall defer payment of the Decedent's Account until the later of the Death Tax Clearance Date and the payment date designated in the Decedent's Initial Election or Subsequent Election;

(iii) The Administrator shall withdraw from the Decedent's Account such amount or amounts as the Decedent's Personal Representative shall certify to the Administrator as being necessary to pay the Death Taxes apportioned against the Decedent's Account; the Administrator shall remit the amounts so withdrawn to the Personal Representative, who shall apply the same to the payment of the Decedent's Death Taxes, or the Administrator may pay such amounts directly to any taxing authority as payment on account of Decedent's Death Taxes, as the Administrator elects;

(iv) If the Administrator makes a withdrawal from the Decedent's Account to pay the Decedent's Death Taxes and such withdrawal causes the recognition of income to the Beneficiary, the Administrator shall pay to the Beneficiary from the Decedent's Account, within thirty (30) days of the Beneficiary's request, the amount necessary to enable the Beneficiary to pay the Beneficiary's income tax liability resulting from such recognition of income; additionally, the Administrator shall pay to the Beneficiary from the Decedent's Account, within thirty (30) days of the Beneficiary's request, such additional amounts as are required to enable the Beneficiary to pay the Beneficiary's income tax liability attributable to the Beneficiary's recognition of income resulting from a distribution from the Decedent's Account pursuant to this Section 3.7(b)(iv);

(v) Amounts withdrawn from the Decedent's Account by the Administrator pursuant to Sections 3.7(b)(ii) and 3.7(b)(iv) shall be withdrawn from the portions of Decedent's Account having the earliest distribution dates as specified in Decedent's Initial Election or Subsequent Election; and

(vi) Within a reasonable time after the later to occur of the Death Tax Clearance Date and the payment date designated in the Decedent's Initial Election or Subsequent Election, the Administrator shall pay the Decedent's Account to the Beneficiary.

ARTICLE 4—MANNER OF DISTRIBUTION

4.1. Manner of Distribution.

(a) Amounts credited to an Account shall be distributed, pursuant to an Initial Election or Subsequent Election in either (i) a lump sum payment or (ii) substantially equal annual installments over a five (5), ten (10) or fifteen (15) year period or (iii) substantially equal monthly installments over a period not exceeding fifteen (15) years. Installment distributions payable in the form of shares of Company Stock shall be rounded to the nearest whole share.

(b) Notwithstanding any Initial Election or Subsequent Election or any other provision of the Plan to the contrary:

(i) distributions pursuant to Initial Elections or Subsequent Elections shall be made in one lump sum payment unless the portion of a Participant's Account subject to distribution, as of both the date of the Initial Election or Subsequent Election and the benefit commencement date, has a value of more than \$10,000;

(ii) following a Participant's termination of employment for any reason, if the amount credited to the Participant's Account has a value of \$25,000 or less, the Administrator may, in its sole discretion, direct that such amount be distributed to the Participant (or Beneficiary, as applicable) in one lump sum payment; provided, however, that this Section 4.1(b)(ii) shall not apply to any amount credited to a Participant's Account until the expiration of the deferral period applicable under any Initial Election or Subsequent Election in effect as of April 29, 2002.

4.2. Determination of Account Balances for Purposes of Distribution. The amount of any distribution made pursuant to Section 4.1 shall be based on the balances in the Participant's Account on the date of distribution. For this purpose, the balance in a Participant's Account shall be calculated by crediting income, gains and losses under the Company Stock Fund and Income Fund, as applicable, through the date immediately preceding the date of distribution.

4.3. Plan-to-Plan Transfers. The Administrator may delegate its authority to arrange for plan-to-plan transfers as described in this Section 4.3 to an officer of the Company or committee of two or more officers of the Company.

(a) The Administrator may, with a Participant's consent, make such arrangements as it may deem appropriate to transfer the Company's obligation to pay benefits with respect to such Participant which have not become payable under this Plan, to another employer, whether through a deferred compensation plan, program or arrangement sponsored by such other employer or otherwise, or to another deferred compensation plan, program or arrangement sponsored by the Company or an Affiliate. Following the completion of such transfer, with respect to the benefit transferred, the Participant shall have no further right to payment under this Plan.

(b) Pursuant to Q-A 19(c) of *IRS Notice* 2005-1, to the extent provided by the Committee or its delegate, on or before December 31, 2005, a Participant may, with respect to all or any portion of his or her Account, make new payment elections as to the form and timing of payment of such amounts as may be permitted under the Comcast Corporation 2005 Deferred Compensation Plan, provided that following the completion of such new payment election, such amounts shall not be treated as grandfathered benefits under this Plan, but instead shall be treated as non-grandfathered benefits, subject to the rules of the Comcast Corporation 2005 Deferred Compensation Plan.

ARTICLE 5-BOOK ACCOUNTS

5.1. Deferred Compensation Account. A deferred Compensation Account shall be established for each Outside Director and Eligible Employee when such Outside Director or Eligible Employee becomes a Participant. Compensation deferred pursuant to the Plan shall be credited to the Account on the date such Compensation would otherwise have been payable to the Participant.

5.2. Crediting of Income, Gains and Losses on Accounts.

(a) In General. Except as otherwise provided in this Section 5.2, the Administrator shall credit income, gains and losses with respect to each Participant's Account as if it were invested in the Income Fund.

(b) Investment Fund Elections.

(i) Except for amounts credited to the Accounts of Participants who are Outside Directors who have elected to defer the receipt of Compensation payable in the form of Company Stock, all amounts

credited to Participants' Accounts on and after July 9, 2002 shall be credited with income, gains and losses as if it were invested in the Income Fund. Each Participant who, as of July 9, 2002, has all or any portion of his or her Account credited with income, gains and losses as if it were invested in the Company Stock Fund may direct, as of any business day, to have all or any portion of the amount credited to the Company Stock Fund deemed transferred to the Income Fund, in accordance with procedures established by the Administrator from time to time. No portion of the Participant's Account credited to the Income Fund may be deemed transferred to the Company Stock Fund.

(ii) With respect to amounts credited to Participants' Accounts through July 9, 2002, investment fund elections shall continue in effect until revoked or superseded. Except for amounts credited to the Accounts of Participants who are Outside Directors who have elected to defer the receipt of Compensation payable in the form of Company Stock, all amounts credited to Participants' Accounts on and after July 9, 2002 shall be deemed to be invested in the Income Fund. Except for amounts described in Section 5.2(c), notwithstanding any investment fund election to the contrary, as of the valuation date (as determined under Section 4.2) for the distribution of all or any portion of a Participant's Account that is subject to distribution in the form of installments described in Section 4.1(a) or (b), such Account, or portion thereof, shall be deemed invested in the Income Fund (and transferred from the Company Stock Fund to the Income Fund, to the extent necessary) until such Account, or portion thereof, is distributed in full.

(iii) Investment fund elections under this Section 5.2(b) shall be effective as soon as practicable following the Participant's election, pursuant to procedures established by the Administrator. An Active Participant may not make an investment fund election with respect to Compensation to be deferred for a calendar year.

(iv) Except for amounts described in Section 5.2(c), if a Participant ceases to continue in service as an Active Participant, then, notwithstanding any election to the contrary, such Participant's Account shall be deemed invested in the Income Fund, effective as of the first day of any calendar year beginning after such Participant ceases to continue in service as an Active Participant.

(c) Outside Director Stock Fund Credits. Amounts credited to the Accounts of Outside Directors in the form of Company Stock shall be credited with income, gains and losses as if they were invested in the Company Stock Fund. No portion of such Participant's Account attributable to amounts credited after December 31, 2002 to the Company Stock Fund may be deemed transferred to the Income Fund. Distributions of amounts credited to the Company Stock Fund with respect to Outside Directors' Accounts after December 31, 2002 shall be distributable in the form of Company Stock, rounded to the nearest whole share.

(d) Timing of Credits. Compensation deferred pursuant to the Plan shall be deemed invested in the Income Fund on the date such Compensation would otherwise have been payable to the Participant. Accumulated Account balances subject to an investment fund election under Section 5.2(b) shall be deemed invested in the applicable investment fund as of the effective date of such election. The value of amounts deemed invested in the Company Stock Fund shall be based on hypothetical purchases and sales of Company Stock at Fair Market Value as of the effective date of an investment election

5.3. Status of Deferred Amounts. Regardless of whether or not the Company is a Participant's employer, all Compensation deferred under this Plan shall continue for all purposes to be a part of the general funds of the Company.

5.4. Participants' Status as General Creditors. Regardless of whether or not the Company is a Participant's employer, an Account shall at all times represent a general obligation of the Company. The Participant shall be a general creditor of the Company with respect to this obligation, and shall not have a secured or preferred position with respect to the Participant's Accounts. Nothing contained herein shall be deemed to create an escrow, trust, custodial account or fiduciary relationship of any kind. Nothing contained herein shall be construed to eliminate any priority or preferred position of a Participant in a bankruptcy matter with respect to claims for wages.

ARTICLE 6-NO ALIENATION OF BENEFITS; PAYEE DESIGNATION

Except as otherwise required by applicable law, the right of any Participant or Beneficiary to any benefit or interest under any of the provisions of this Plan shall not be subject to encumbrance, attachment, execution, garnishment, assignment, pledge, alienation, sale, transfer, or anticipation, either by the voluntary or involuntary act of any Participant or any Participant's Beneficiary or by operation of law, nor shall such payment, right, or interest be subject to any other legal or equitable process. However, subject to the terms and conditions of the Plan, a Participant or Beneficiary may direct that any amount payable pursuant to an Initial Election or a Subsequent Election on any date designated for payment be paid to any person or persons or legal entity or entities, including, but not limited to, an organization exempt from federal income tax under section 501(c)(3) of the Code, instead of to the Participant or Beneficiary. Such a payee designation shall be provided to the Administrator by the Participant or Beneficiary in writing on a form provided by the Administrator, and shall not be effective unless it is provided inimediately preceding the time of payment. The Company's payment pursuant to such a payee designation shall relieve the Company and its Affiliates of all liability for such payment.

ARTICLE 7-DEATH OF PARTICIPANT

7.1. Death of Participant. A Deceased Participant's Account shall be distributed in accordance with the last Initial Election or Subsequent Election made by the Deceased Participant before the Deceased Participant's death, unless the Deceased Participant's Surviving Spouse or other Beneficiary timely elects to accelerate or defer the time or change the manner of payment pursuant to Section 3.5.

7.2. Designation of Beneficiaries. Each Participant and Beneficiary shall have the right to designate one or more Beneficiaries to receive distributions in the event of the Participant's or Beneficiary's death by filing with the Administrator a Beneficiary designation on the form provided by the Administrator for such purpose. The designation of a Beneficiary or Beneficiaries may be changed by a Participant or Beneficiary at any time prior to such Participant's or Beneficiary's death by the delivery to the Administrator of a new Beneficiary designation form.

ARTICLE 8—HARDSHIP DISTRIBUTIONS

Notwithstanding the terms of an Initial Election or Subsequent Election, if, at the Participant's request, the Board determines that the Participant has incurred a Hardship, the Board may, in its discretion, authorize the immediate distribution of all or any portion of the Participant's Account.

ARTICLE 9—INTERPRETATION

9.1. Authority of Committee. The Committee shall have full and exclusive authority to construe, interpret and administer this Plan and the Committee's construction and interpretation thereof shall be binding and conclusive on all persons for all purposes.

9.2. Claims Procedure. If an individual (hereinafter referred to as the "Applicant," which reference shall include the legal representative, if any, of the individual) does not receive timely payment of benefits to which the Applicant believes he is entitled under the Plan, the Applicant may make a claim for benefits in the manner hereinafter provided.

An Applicant may file a claim for benefits with the Administrator on a form supplied by the Administrator. If the Administrator wholly or partially denies a claim, the Administrator shall provide the Applicant with a written notice stating:

- (a) The specific reason or reasons for the denial;
- (b) Specific reference to pertinent Plan provisions on which the denial is based;

(c) A description of any additional material or information necessary for the Applicant to perfect the claim and an explanation of why such material or information is necessary; and

(d) Appropriate information as to the steps to be taken in order to submit a claim for review.

Written notice of a denial of a claim shall be provided within 90 days of the receipt of the claim, provided that if special circumstances require an extension of time for processing the claim, the Administrator may notify the Applicant in writing that an additional period of up to 90 days will be required to process the claim.

If the Applicant's claim is denied, the Applicant shall have 60 days from the date of receipt of written notice of the denial of the claim to request a review of the denial of the claim by the Administrator. Request for review of the denial of a claim must be submitted in writing. The Applicant shall have the right to review pertinent documents and submit issues and comments to the Administrator in writing. The Administrator shall provide a written decision within 60 days of its receipt of the Applicant's request for review, provided that if special circumstances require an extension of time for processing the review of the Applicant's request for review.

It is intended that the claims procedures of this Plan be administered in accordance with the claims procedure regulations of the Department of Labor set forth in 29 CFR § 2560.503-1.

Claims for benefits under the Plan must be filed with the Administrator at the following address:

Comcast Corporation 1500 Market Street Philadelphia, PA 19102 Attention: General Counsel

ARTICLE 10—AMENDMENT OR TERMINATION

10.1. Amendment or Termination. Except as otherwise provided by Section 10.2, the Company, by action of the Board or by action of the Committee, shall have the right at any time, or from time to time, to amend or modify this Plan. The Company, by action of the Board, shall have the right to terminate this Plan at any time.

10.2. Amendment of Rate of Credited Earnings. No amendment shall change the Applicable Interest Rate with respect to the portion of a Participant's Account that is attributable to an Initial Election or Subsequent Election made with respect to Compensation earned in a calendar year and filed with the Administrator before the date of adoption of such amendment by the Board. For purposes of this Section 10.2, a Subsequent Election to defer the payment of part or all of an Account for an additional period after a previously-elected payment date (as described in Section 3.5) shall be treated as a separate Subsequent Election with respect to such Account.

ARTICLE 11-WITHHOLDING OF TAXES

Whenever the Participating Company is required to credit deferred Compensation to the Account of a Participant, the Participating Company shall have the right to require the Participant to remit to the Participating Company an amount sufficient to satisfy any federal, state and local withholding tax requirements prior to the date on which the deferred Compensation shall be deemed credited to the Account of the Participant, or take any action whatever that it deems necessary to protect its interests with respect to tax liabilities. The Participating Company's obligation to credit deferred Compensation to an Account shall be conditioned on the Participant's compliance, to the Participating Company's satisfaction, with any withholding requirement. To the maximum extent possible, the Participating Company shall satisfy all applicable withholding tax requirements by withholding tax from other Compensation payable by the Participating Company to the Participating, or by the Participant's delivery of cash to the Participating Company in an amount equal to the applicable withholding tax.

ARTICLE 12—MISCELLANEOUS PROVISIONS

12.1. No Right to Continued Employment. Nothing contained herein shall be construed as conferring upon any Participant the right to remain in service as an Outside Director or in the employment of a Participating Company as an executive or in any other capacity.

12.2. Expenses of Plan. All expenses of the Plan shall be paid by the Participating Companies.

12.3. Gender and Number. Whenever any words are used herein in any specific gender, they shall be construed as though they were also used in any other applicable gender. The singular form, whenever used herein, shall mean or include the plural form, and vice versa, as the context may require.

12.4. Law Governing Construction. The construction and administration of the Plan and all questions pertaining thereto, shall be governed by the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), and other applicable federal law and, to the extent not governed by federal law, by the laws of the Commonwealth of Pennsylvania.

12.5. Headings Not a Part Hereof. Any headings preceding the text of the several Articles, Sections, subsections, or paragraphs hereof are inserted solely for convenience of reference and shall not constitute a part of the Plan, nor shall they affect its meaning, construction, or effect.

12.6. Severability of Provisions. If any provision of this Plan is determined to be void by any court of competent jurisdiction, the Plan shall continue to operate and, for the purposes of the jurisdiction of that court only, shall be deemed not to include the provision determined to be void.

ARTICLE 13—EFFECTIVE DATE

The effective date of this amendment and restatement of the Plan shall be February 16, 2005.

IN WITNESS WHEREOF, COMCAST CORPORATION has caused this Plan to be executed by its officers thereunto duly authorized, and its corporate seal to be affixed hereto, as of the 16th day of February, 2005.

COMCAST CORPORATION

BY:

ATTEST:

QuickLinks

COMCAST CORPORATION 2002 DEFERRED COMPENSATION PLAN ARTICLE 1—COVERAGE OF PLAN ARTICLE 2—DEFINITIONS ARTICLE 2—DEFINITIONS ARTICLE 4—MANNER OF DISTRIBUTION ARTICLE 4—MANNER OF DISTRIBUTION ARTICLE 5—BOOK ACCOUNTS ARTICLE 6—NO ALIENATION OF BENEFITS: PAYEE DESIGNATION ARTICLE 7—DEATH OF PARTICIPANT ARTICLE 8—HARDSHIP DISTRIBUTIONS ARTICLE 8—HARDSHIP DISTRIBUTIONS ARTICLE 9—INTERPRETATION ARTICLE 10—AMENDMENT OR TERMINATION ARTICLE 11—WITHHOLDING OF TAXES ARTICLE 12—MISCELLANEOUS PROVISIONS ARTICLE 13—EFFECTIVE DATE

Exhibit 10.5

COMCAST CORPORATION 2005 DEFERRED COMPENSATION PLAN

ARTICLE 1-BACKGROUND AND COVERAGE OF PLAN

1.1. Background and Adoption of Plan. In recognition of the services provided by certain key employees and in order to make additional retirement benefits and increased financial security available on a tax-favored basis to those individuals, the Board of Directors of Comcast Corporation, a Pennsylvania corporation (the "Board"), hereby amends and restates the Comcast Corporation 2005 Deferred Compensation Plan (the "Plan"), effective January 1, 2005 (the "Effective Date").

Prior to the Effective Date, the Comcast Corporation 2002 Deferred Compensation Plan (the "2002 Plan") was in effect. In order to preserve the favorable tax treatment available to deferrals under the 2002 Plan in light of the American Jobs Creation Act of 2004 and the regulations issued by the Department of the Treasury thereunder (the "AJCA"), the Board has prohibited future deferrals under the 2002 Plan of amounts earned and vested on and after the Effective Date. Amounts earned and vested prior to the Effective Date are and will remain subject to the terms of the 2002 Plan. Amounts earned and vested on and after the Effective Date will be available to be deferred pursuant to the Plan, subject to its terms and conditions.

1.2. Reservation of Right to Amend to Comply with AJCA. The Board reserves the right to amend the Plan, either retroactively or prospectively, in whatever respect is required to achieve and maintain compliance with the requirements of the AJCA.

1.3. Plan Unfunded and Limited to Outside Directors and Select Group of Management or Highly Compensated Employees. The Plan is unfunded and is maintained primarily for the purpose of providing outside directors and a select group of management or highly compensated employees the opportunity to defer the receipt of compensation otherwise payable to such outside directors and eligible employees in accordance with the terms of the Plan.

ARTICLE 2—DEFINITIONS

2.1. "Account" means the bookkeeping accounts established pursuant to Section 5.1 and maintained by the Administrator in the names of the respective Participants, to which all amounts deferred and earnings allocated under the Plan shall be credited, and from which all amounts distributed pursuant to the Plan shall be debited.

- 2.2. "Active Participant" means:
 - (a) Each Participant who is in active service as an Outside Director; and
 - (b) Each Participant who is actively employed by a Participating Company as an Eligible Employee.
- 2.3. "Administrator" means the Committee.

2.4. "Affiliate" means, with respect to any Person, any other Person that, directly or indirectly, is in control of, is controlled by, or is under common control with, such Person. For purposes of this definition, the term "control," including its correlative terms "controlled by" and "under common control with," mean, with respect to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract or otherwise.

2.5. "Annual Rate of Pay" means, as of any date, an employee's annualized base pay rate. An employee's Annual Rate of Pay shall not include sales commissions or other similar payments or awards.

2.6. "Applicable Interest Rate" means:

(a) Except as otherwise provided in Sections 2.7(b), the Applicable Interest Rate means the interest rate that, when compounded daily pursuant to rules established by the Administrator from time to time, is mathematically equivalent to 12% per annum, compounded annually.

(b) Except to the extent otherwise required by Section 10.2, effective for the period beginning as soon as administratively practicable following a Participant's employment termination date to the date the Participant's Account is distributed in full, the Administrator, in its sole discretion, may designate the term "Applicable Interest Rate" for such Participant's Account to mean the lesser of (i) the rate in effect under Section 2.6(a) or (ii) the Prime Rate plus one percent. Notwithstanding the foregoing, the Administrator may delegate its authority to determine the Applicable Interest Rate under this Section 2.6(b) to an officer of the Company or committee of two or more officers of the Company.

2.7. "Beneficiary" means such person or persons or legal entity or entities, including, but not limited to, an organization exempt from federal income tax under section 501(c)(3) of the Code, designated by a Participant or Beneficiary to receive benefits pursuant to the terms of the Plan after such Participant's or Beneficiary's death. If no Beneficiary is designated by the Participant or Beneficiary, or if no Beneficiary survives the Participant or Beneficiary (as the case may be), the Participant's Beneficiary shall be the Participant's Surviving Spouse if the Participant has a Surviving Spouse and otherwise the Beneficiary's estate.

2.8. "Board" means the Board of Directors of the Company.

- 2.9. "Change of Control" means any transaction or series of transactions that constitutes:
 - (a) a change in the ownership of the Company, within the meaning of Q&A 12 of IRS Notice 2005-1;
 - (b) a change in effective control of the Company, within the meaning of Q&A 13 of IRS Notice 2005-1; or
 - (c) a change in the ownership of a substantial portion of the assets of the Company, within the meaning of Q&A 14 of IRS Notice 2005-1.

2.10. "Code" means the Internal Revenue Code of 1986, as amended.

2.11. "Committee" means the Compensation Committee of the Board of Directors of the Company.

2.12. "Company" means Comcast Corporation, a Pennsylvania corporation, including any successor thereto by merger, consolidation, acquisition of all or substantially all the assets thereof, or otherwise.

2.13. "Company Stock" means with respect to amounts credited to the Company Stock Fund pursuant to deferral elections by Outside Directors made pursuant to Section 3.1(a), Comcast Corporation Class A Common Stock, par value \$0.01, including a fractional share, and such other securities issued by Comcast Corporation as may be subject to adjustment in the event that shares of either class of Company Stock are changed into, or exchanged for, a different number or kind of shares of stock or other securities of the Company, whether through merger, consolidation, reorganization, recapitalization, stock dividend, stock split-up or other substitution of securities of the Company. In such event, the Committee shall make appropriate equitable anti-dilution adjustments to the number and class of hypothetical shares of Company Stock credited to Participants' Accounts under the Company Stock Fund. Any reference to the term "Company Stock" in the Plan shall be a reference to the appropriate number and class of stock as adjusted pursuant to this Section 2.13. The Committee's adjustment shall be effective and binding for all purposes of the Plan.

2.14. "Company Stock Fund" means a hypothetical investment fund pursuant to which income, gains and losses are credited to a Participant's Account as if the Account, to the extent deemed invested in the Company Stock Fund, were invested in hypothetical shares of Company Stock, and all dividends and other distributions paid with respect to Company Stock were held uninvested in cash, and reinvested in additional hypothetical shares of Company Stock as of the next succeeding December 31, based on the Fair Market Value of the Company Stock for such December 31.

2.15. "Compensation" means:

(a) In the case of an Outside Director, the total remuneration payable in cash or payable in Company Stock (as elected by the Outside Director pursuant to the Comcast Corporation 2003 Director

Compensation Plan) for services as a member of the Board and as a member of any Committee of the Board; and

(b) In the case of an Eligible Employee, the total cash remuneration for services payable by a Participating Company, excluding (i) Severance Pay and (ii) sales commissions or other similar payments or awards.

2.16. "Death Tax Clearance Date" means the date upon which a Deceased Participant's or a deceased Beneficiary's Personal Representative certifies to the Administrator that (i) such Deceased Participant's or deceased Beneficiary's Death Taxes have been finally determined, (ii) all of such Deceased Participant's or deceased Beneficiary's Account have been paid in full and (iii) all potential liability for Death Taxes with respect to the Deceased Participant's or deceased Beneficiary's Account has been satisfied.

2.17. "Death Taxes" means any and all estate, inheritance, generation-skipping transfer, and other death taxes as well as any interest and penalties thereon imposed by any governmental entity (a "taxing authority") as a result of the death of the Participant or the Participant's Beneficiary.

2.18. "Deceased Participant" means a Participant whose employment, or, in the case of a Participant who was an Outside Director, a Participant whose service as an Outside Director, is terminated by death.

2.19. "Disability" means:

(a) an individual's inability to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment which can be expected to result in death or can be expected to last for a continuous period of not less than 12 months; or

(b) circumstances under which, by reason of any medically determinable physical or mental impairment which can be expected to result in death or can be expected to last for a continuous period of not less than 12 months, an individual is receiving income replacement benefits for a period of not less than three months under an accident or health plan covering employees of the individual's employer.

2.20. "Disabled Participant" means:

(a) A Participant whose employment or, in the case of a Participant who is an Outside Director, a Participant whose service as an Outside Director, is terminated by reason of Disability;

(b) The duly-appointed legal guardian of an individual described in Section 2.20(a) acting on behalf of such individual.

- 2.21. "Eligible Employee" means:
 - (a) Each Grandfathered Employee.

(b) Each employee of a Participating Company whose Annual Rate of Pay is \$200,000 or more as of both (i) the date on which an Initial Election is filed with the Administrator and (ii) the first day of the calendar year in which such Initial Election is filed.

- (c) Each New Key Employee.
- (d) Each other employee of a Participating Company who is designated by the Committee, in its discretion, as an Eligible Employee.
- 2.22. "Fair Market Value"

(a) If shares of Company Stock are listed on a stock exchange, Fair Market Value shall be determined based on the last reported sale price of a share on the principal exchange on which shares are listed on the date of determination, or if such date is not a trading day, the next trading date.

(b) If shares of Company Stock are not so listed, but trades of shares are reported on the Nasdaq National Market, Fair Market Value shall be determined based on the last quoted sale price of a share on the Nasdaq National Market on the date of determination, or if such date is not a trading day, the next trading date.



(c) If shares of Company Stock are not so listed nor trades of shares so reported, Fair Market Value shall be determined by the Committee in good faith.

2.23. "Grandfathered Employee" means:

(a) Each employee of a Participating Company who, as of December 31, 1989, was eligible to participate in the Prior Plan and who has been in continuous service to the Company or an Affiliate since December 31, 1989.

(b) Each employee of a Participating Company who was, at any time before January 1, 1995, eligible to participate in the Comcast Corporation Deferred Compensation Plan and whose Annual Rate of Pay is \$90,000 or more as of both (i) the date on which an Initial Election is filed with the Administrator and (ii) the first day of each calendar year beginning after December 31, 1994.

(c) Each individual who was an employee of an entity that was a Participating Company in the Prior Plan as of June 30, 2002 and who has an Annual Rate of Pay of \$125,000 as of each of (i) June 30, 2002; (ii) the date on which an Initial Election is filed with the Administrator and (iii) the first day of each calendar year beginning after December 31, 2002.

(d) Each employee of a Participating Company who (i) as of December 31, 2002, was an "Eligible Employee" within the meaning of Section 2.34 of the AT&T Broadband Deferred Compensation Plan (as amended and restated, effective November 18, 2002) with respect to whom an account was maintained, and (ii) for the period beginning on December 31, 2002 and extending through any date of determination, has been actively and continuously in service to the Company or an Affiliate.

2.24. "Hardship" means a Participant's severe financial hardship due to an unforeseeable emergency resulting from a sudden and unexpected illness or accident of the Participant, or, a sudden and unexpected illness or accident of a dependent (as defined by section 152(a) of the Code) of the Participant, or loss of the Participant's property due to casualty, or other similar and extraordinary unforeseeable circumstances arising as a result of events beyond the control of the Participant. A need to send the Participant's child to college or a desire to purchase a home is not an unforeseeable emergency. No Hardship shall be deemed to exist to the extent that the financial hardship is or may be relieved (a) through reimbursement or compensation by insurance or otherwise, (b) by borrowing from commercial sources on reasonable commercial terms to the extent that this borrowing would not itself cause a severe financial hardship, (c) by cessation of deferrals under the Plan, or (d) by liquidation of the Participant's other assets (including assets of the Participant's spouse and minor children that are reasonably available to the Participant) to the extent that this liquidation would not itself cause severe financial hardship. For the purposes of the preceding sentence, the Participant's resources shall be deemed to include those assets of his spouse and minor children that are reasonably available to the Participant; however, property held for the Participant's child under an irrevocable trust or under a Uniform Gifts to Minors Act custodianship or Uniform Transfers to Minors Act custodianship shall not be treated as a resource of the Participant. The Board shall determine whether the circumstances of the Participant constitute an unforeseeable emergency and thus a Hardship within the meaning of this Section. Following a uniform procedure, the Board's determination shall consider any facts or conditions deemed necessary or advisable by the Board, and the Participant shall be required to submit any evidence of the Participant's circumstances that the Board requires. The determination as to whether the Participant's circumstances are a case of Hardship shall be based on the facts of each case; provided however, that all determinations as to Hardship shall be uniformly and consistently made according to the provisions of this Section for all Participants in similar circumstances.

2.25. "Inactive Participant" means each Participant (other than a Retired Participant, Deceased Participant or Disabled Participant) who is not in active service as an Outside Director and is not actively employed by a Participating Company.

2.26. "Income Fund" means a hypothetical investment fund pursuant to which income, gains and losses are credited to a Participant's Account as if the Account, to the extent deemed invested in the Income Fund, were credited with interest at the Applicable Interest Rate.

2.27. "Initial Election" means a written election on a form provided by the Administrator, filed with the Administrator in accordance with Article 3, pursuant to which an Outside Director or an Eligible Employee may:

(a) Elect to defer all or any portion of the Compensation payable for the performance of services as an Outside Director or as an Eligible Employee following the time that such election is filed; and

(b) Designate the time of payment of the amount of deferred Compensation to which the Initial Election relates.

2.28. "New Key Employee" means each employee of a Participating Company:

(a) who becomes an employee of a Participating Company and has an Annual Rate of Pay of \$200,000 or more as of his employment commencement date, or

(b) who has an Annual Rate of Pay that is increased to \$200,000 or more and who, immediately preceding such increase, was not an Eligible Employee.

2.29. "Normal Retirement" means:

(a) For a Participant who is an employee of a Participating Company immediately preceding his termination of employment, a termination of employment that is treated by the Participating Company as a retirement under its employment policies and practices as in effect from time to time; and

(b) For a Participant who is an Outside Director immediately preceding his termination of service, his normal retirement from the Board.

2.30. "Outside Director" means a member of the Board, who is not an employee of a Participating Company.

2.31. "Participant" means each individual who has made an Initial Election, or for whom an Account is established pursuant to Section 5.1, and who has an undistributed amount credited to an Account under the Plan, including an Active Participant, a Deceased Participant and an Inactive Participant.

- 2.32. "Participating Company" means:
 - (a) The Company;
 - (b) Comcast Holdings Corporation;
 - (c) Comcast Cable Communications, LLC, and its subsidiaries;
 - (d) Comcast International Holdings, Inc.;
 - (e) Comcast Online Communications, Inc.;
 - (f) Comcast Business Communications, Inc.;
 - (g) Comcast Cable Communications Holdings, Inc. and its subsidiaries;

(h) Comcast Shared Services Corporation ("CSSC"), to the extent individual employees of CSSC or groups of CSSC employees, categorized by their secondment, are designated as eligible to participate by the Committee or its delegate;

- (i) Comcast Sports Management Services, LLC;
- (j) Home Team Sports Limited Partnership; and
- (k) Any other entities that are subsidiaries of the Company as designated by the Committee in its sole discretion.

2.33. "Performance-Based Compensation" means "performance-based compensation" within the meaning of Q&A 22 of IRS Notice 2005-1, or such other guidance as may be issued by the Department of the Treasury under section 409A of the Code.

2.34. "Performance Period" means a period of at least 12 months during which a Participant may earn Performance-Based Compensation.



2.35. "Person" means an individual, a corporation, a partnership, an association, a trust or any other entity or organization.

2.36. "Plan" means the Comcast Corporation 2005 Deferred Compensation Plan, as set forth herein, and as amended from time to time.

2.37. "Prime Rate" means, for any calendar year, the interest rate that, when compounded daily pursuant to rules established by the Administrator from time to time, is mathematically equivalent to the prime rate of interest (compounded annually) as published in the Eastern Edition of *The Wall Street Journal* on the last business day preceding the first day of such calendar year, and as adjusted as of the last business day preceding the first day of each calendar year beginning thereafter.

2.38. "Prior Plan" means the Comcast Corporation 2002 Deferred Compensation Plan.

2.39. "Retired Participant" means a Participant who has terminated service pursuant to a Normal Retirement.

2.40. "Severance Pay" means any amount that is payable in cash and is identified by a Participating Company as severance pay, or any amount which is payable on account of periods beginning after the last date on which an employee (or former employee) is required to report for work for a Participating Company.

2.41. "Subsequent Election" means a written election on a form provided by the Administrator, filed with the Administrator in accordance with Article 3, pursuant to which a Participant or Beneficiary may elect to defer the time of payment of amounts previously deferred in accordance with the terms of a previously made Initial Election or Subsequent Election.

2.42. "Surviving Spouse" means the widow or widower, as the case may be, of a Deceased Participant or a Deceased Beneficiary (as applicable).

2.43. "Third Party" means any Person, together with such Person's Affiliates, provided that the term "Third Party" shall not include the Company or an Affiliate of the Company.

ARTICLE 3—INITIAL AND SUBSEQUENT ELECTIONS

3.1. Elections.

(a) Initial Elections. Each Outside Director and Eligible Employee shall have the right to defer all or any portion of the Compensation that he would otherwise be entitled to receive in a calendar year (net of applicable withholdings) by filing an Initial Election at the time and in the manner described in this Article 3. The Compensation of such Outside Director or Eligible Employee for a calendar year shall be reduced in an amount equal to the portion of the Compensation deferred by such Outside Director or Eligible Employee for such calendar year pursuant to such Outside Director's or Eligible Employee's Initial Election. Such reduction shall be effected on a pro rata basis from each periodic installment payment of such Outside Director's or Eligible Employee's Compensation for the calendar year (in accordance with the general pay practices of the Participating Company), and credited, as a bookkeeping entry, to such Outside Director's or Eligible Employee's Account in accordance with Section 5.1. Amounts credited to the Accounts of Outside Directors in the form of Company Stock shall be credited to the Company Stock Fund and credited with income, gains and losses in accordance with Section 5.2(c).

(b) Subsequent Elections. Each Participant or Beneficiary shall have the right to elect to defer the time of payment or to change the manner of payment of amounts previously deferred in accordance with the terms of a previously made Initial Election pursuant to the terms of the Plan by filing a Subsequent Election at the time, to the extent, and in the manner described in this Article 3.

(c) Special Transition Rule. Pursuant to Q-A 20 of *IRS Notice 2005-1*, to the extent provided by the Committee or its delegate, a Participant may, on or before December 31, 2005, terminate the deferral of Compensation pursuant to an Initial Election or cancel an Initial Election with regard to

amounts deferred under the Plan, provided that if a Participant terminates the deferral of Compensation pursuant to an Initial Election under this Section 3.1(c), the Company shall pay the Participant the Compensation that would have been deferred if the deferral of Compensation had not been terminated, and provided further that if a Participant cancels an Initial Election with regard to amounts deferred under the Plan, the Company shall pay the Participant the amount deferred pursuant to such Initial Election through the cancellation date, without adjustment for income, gains and losses credited under Section 5.2.

3.2. Filing of Initial Election: General. An Initial Election shall be made on the form provided by the Administrator for this purpose. Except as provided in Section 3.3, no such Initial Election shall be effective with respect to Compensation other than Performance-Based Compensation unless it is filed with the Administrator on or before December 31 of the calendar year preceding the calendar year to which the Initial Election applies, provided that pursuant to Q-A 21 of *IRS Notice 2005-1*, to the extent provided by the Committee or its delegate, a Participant may, on or before March 15, 2005, make an Initial Election with respect to Compensation that relates in full or in part to services provided on or before December 31, 2005, provided further that the amounts to which the Initial Election relates have not been paid or become payable at the time the Initial Election is filed. No such Initial Election shall be effective with respect to Performance-Based Compensation unless it is filed with the Administrator at least six months before the end of the Performance Period during which such Performance-Based Compensation may be earned.

3.3. Filing of Initial Election by New Key Employees and New Outside Directors.

(a) New Key Employees. Notwithstanding Section 3.1 and Section 3.2, a New Key Employee may elect to defer all or any portion of his Compensation that he would otherwise be entitled to receive in the calendar year in which the New Key Employee was employed, beginning with the payroll period next following the filing of an Initial Election with the Administrator and before the close of such calendar year by making and filing the Initial Election with the Administrator within 30 days of such New Key Employee's date of hire or within 30 days of the date such New Key Employee first becomes eligible to participate in the Plan. Any Initial Election by such New Key Employee for succeeding calendar years shall be made in accordance with Section 3.1 and Section 3.2.

(b) New Outside Directors. Notwithstanding Section 3.1 and Section 3.2, an Outside Director may elect to defer all or any portion of his Compensation that he would otherwise be entitled to receive in the calendar year in which an Outside Director's election as a member of the Board becomes effective (provided that such Outside Director is not a member of the Board immediately preceding such effective date), beginning with Compensation payable following the filing of an Initial Election with the Administrator and before the close of such calendar year by making and filing the Initial Election with the Administrator within 30 days of the effective date of such Outside Director's election. Any Initial Election by such Outside Director for succeeding calendar years shall be made in accordance with Section 3.1 and Section 3.2

3.4. Calendar Years to which Initial Election May Apply. A separate Initial Election may be made for each calendar year as to which an Outside Director or Eligible Employee desires to defer all or any portion of such Outside Director's or Eligible Employee's Compensation. The failure of an Outside Director or Eligible Employee to make an Initial Election for any calendar year shall not affect such Outside Director's or Eligible Employee's right to make an Initial Election for any other calendar year.

(a) Initial Election of Distribution Date. Each Outside Director or Eligible Employee shall, contemporaneously with an Initial Election, also elect the time of payment of the amount of the deferred Compensation to which such Initial Election relates; provided, however, that, subject to acceleration (to the extent permitted under the AJCA) pursuant to Section 3.5(e), Section 3.7, Section 7.1, Section 7.2, or Article 8, no distribution may commence earlier than January 2nd of the second calendar year beginning after the date the compensation subject to the Initial Election would be paid but for the Initial Election, nor later than January 2nd of the tenth calendar year beginning after the date the date the compensation subject to the Initial Election would be paid but for the Initial



Election. Further, each Outside Director or Eligible Employee may select with each Initial Election the manner of distribution in accordance with Article 4.

3.5. Subsequent Elections and Elections to Accelerate Payment on Death or Disability. No Subsequent Election shall be effective until 12 months after the date on which such Subsequent Election is made.

(a) Active Participants. Each Active Participant, who has made an Initial Election, or who has made a Subsequent Election, may elect to defer the time of payment of any part or all of such Participant's Account for a minimum of five and a maximum of ten additional years from the previously-elected payment date, by filing a Subsequent Election with the Administrator The number of Subsequent Elections under this Section 3.5(a) shall not be limited.

(b) Inactive Participants. The Committee may, in its sole and absolute discretion, permit an Inactive Participant to make a Subsequent Election defer the time of payment of any part or all of such Inactive Participant's Account for a minimum of five years and a maximum of ten additional years from the previously-elected payment date, by filing a Subsequent Election with the Administrator on or before the close of business on June 30 of the calendar year preceding the calendar year in which the lump-sum distribution or initial installment payment would otherwise be made. The number of Subsequent Elections under this Section 3.5(b) shall be determined by the Committee in its sole and absolute discretion.

(c) Surviving Spouses.

(i) Acceleration Election. To the extent permitted under the AJCA (except to the extent that Section 3.7(b) applies), a Surviving Spouse who is a Deceased Participant's Beneficiary may elect to accelerate the time of payment of the Deceased Participant's Account from the date payment would otherwise be made to a time that is as soon as reasonably practicable following the Deceased Participant's date of death.

(ii) Subsequent Election. A Surviving Spouse who is a Deceased Participant's Beneficiary may elect to defer the time of payment of any part or all of such Deceased Participant's Account the payment of which would be made more than 12 months after the date of such election. Such election shall be made by filing a Subsequent Election with the Administrator in which the Surviving Spouse shall specify the change in the time of payment, which shall be no less than five (5) years nor more than ten (10) years from the previously- elected payment date, or such Surviving Spouse may elect to defer payment until such Surviving Spouse's death. A Surviving Spouse may make a total of two (2) Subsequent Elections under this Section 3.5(c)(ii), with respect to all or any part of the Deceased Participant's Account. Subsequent Elections pursuant to this Section 3.5(c)(ii) may specify different changes with respect to different parts of the Deceased Participant's Account.

(d) Beneficiary of a Deceased Participant Other Than a Surviving Spouse.

(i) Acceleration Election. To the extent permitted under the AJCA (except to the extent that Section 3.7(b) applies), a Beneficiary of a Deceased Participant other than a Surviving Spouse may elect to accelerate the time of payment of the Deceased Participant's Account from the date payment would otherwise be made to a time that is as soon as reasonably practicable following the Deceased Participant's date of death.

(ii) Subsequent Election. A Beneficiary of a Deceased Participant other than a Surviving Spouse may elect to defer the time of payment, of any part or all of such Deceased Participant's Account the payment of which would be made more than 12 months after the date of such election. Such election shall be made by filing a Subsequent Election with the Administrator in which the Beneficiary shall specify the deferral of the time of payment, which shall be no less than five (5) years nor more than ten (10) years from the previously-elected payment date. A Beneficiary may make one (1) Subsequent Election under this Section 3.5(d)(i), with respect to all or any part

of the Deceased Participant's Account. Subsequent Elections pursuant to this Section 3.5(d)(i) may specify different changes with respect to different parts of the Deceased Participant's Account.

(e) Disabled Participant. To the extent permitted under the AJCA, a Disabled Participant may elect to accelerate the time of payment of the Disabled Participant's Account from the date payment would otherwise be made to a time that is as soon as reasonably practicable following the time the Disability occurred.

(f) Retired Participants and Disabled Participants. The Committee may, in its sole and absolute discretion, permit a Retired Participant or a Disabled Participant to make a Subsequent Election to defer the time of payment of any part or all of such Retired or Disabled Participant's Account that would not otherwise become payable within twelve (12) months of such Subsequent Election for a minimum of five (5) years and a maximum of ten (10) additional years from the previously-elected payment date, by filing a Subsequent Election with the Administrator on or before the close of business on the date that is at least twelve (12) months before the date on which the lump-sum distribution or initial installment payment would otherwise be made. The number of Subsequent Elections under this Section 3.5(f) shall be determined by the Committee in its sole and absolute discretion.

(g) Most Recently Filed Initial Election or Subsequent Election Controlling. Subject to acceleration pursuant to Section 3.5(e), Section 3.7 or Section 7.1 (to the extent permitted under the AJCA), no distribution of the amounts deferred by a Participant for any calendar year shall be made before the payment date designated by the Participant or Beneficiary on the most recently filed Initial Election or Subsequent Election with respect to each deferred amount.

3.6. Discretion to Provide for Distribution in Full Upon or Following a Change of Control. To the extent permitted by IRS Notice 2005–1, in connection with a Change of Control, and for the 12-month period following a Change of Control, the Committee may exercise its discretion to terminate the Plan and, notwithstanding any other provision of the Plan or the terms of any Initial Election or Subsequent Election, distribute the Account balance of each Participant in full and thereby effect the revocation of any outstanding Initial Elections or Subsequent Elections.

3.7. Withholding and Payment of Death Taxes.

(a) Notwithstanding any other provisions of this Plan to the contrary, including but not limited to the provisions of Article 3 and Article 7, or any Initial or Subsequent Election filed by a Deceased Participant or a Deceased Participant's Beneficiary (for purposes of this Section, the "Decedent"), and to the extent permitted by IRS Notice 2005-1, the Administrator shall apply the terms of Section 3.7(b) to the Decedent's Account unless the Decedent affirmatively has elected, in writing, filed with the Administrator, to waive the application of Section 3.7(b).

(b) Unless the Decedent affirmatively has elected, pursuant to Section 3.7(a), that the terms of this Section 3.7(b) not apply, but only to the extent permitted under the AJCA:

(i) The Administrator shall prohibit the Decedent's Beneficiary from taking any action under any of the provisions of the Plan with regard to the Decedent's Account other than the Beneficiary's making of a Subsequent Election pursuant to Section 3.5;

(ii) The Administrator shall defer payment of the Decedent's Account until the later of the Death Tax Clearance Date and the payment date designated in the Decedent's Initial Election or Subsequent Election;

(iii) The Administrator shall withdraw from the Decedent's Account such amount or amounts as the Decedent's Personal Representative shall certify to the Administrator as being necessary to pay the Death Taxes apportioned against the Decedent's Account; the Administrator shall remit the amounts so withdrawn to the Personal Representative, who shall apply the same to the payment of the Decedent's Death Taxes, or the Administrator may pay such amounts directly to any taxing authority as payment on account of Decedent's Death Taxes, as the Administrator elects;

(iv) If the Administrator makes a withdrawal from the Decedent's Account to pay the Decedent's Death Taxes and such withdrawal causes the recognition of income to the Beneficiary, the Administrator shall pay to the Beneficiary from the Decedent's Account, within thirty (30) days of the Beneficiary's request, the amount necessary to enable the Beneficiary to pay the Beneficiary's income tax liability resulting from such recognition of income; additionally, the Administrator shall pay to the Beneficiary from the Decedent's Account, within thirty (30) days of the Beneficiary's request, such additional amounts as are required to enable the Beneficiary to pay the Beneficiary's income tax liability attributable to the Beneficiary's recognition of income resulting from a distribution from the Decedent's Account pursuant to this Section 3.7(b)(iv);

(v) Amounts withdrawn from the Decedent's Account by the Administrator pursuant to Sections 3.7(b)(iii) and 3.7(b)(iv) shall be withdrawn from the portions of Decedent's Account having the earliest distribution dates as specified in Decedent's Initial Election or Subsequent Election; and

(vi) Within a reasonable time after the later to occur of the Death Tax Clearance Date and the payment date designated in the Decedent's Initial Election or Subsequent Election, the Administrator shall pay the Decedent's Account to the Beneficiary.

ARTICLE 4—MANNER OF DISTRIBUTION

4.1. Manner of Distribution.

(a) Amounts credited to an Account shall be distributed, pursuant to an Initial Election or Subsequent Election in either (i) a lump sum payment or (ii) substantially equal monthly or annual installments over a five (5), ten (10) or fifteen (15) year period. Installment distributions payable in the form of shares of Company Stock shall be rounded to the nearest whole share.

(b) To the extent permitted by Q-A 15(e) of IRS Notice 2005-1, notwithstanding any Initial Election, Subsequent Election or any other provision of the Plan to the contrary:

(i) distributions pursuant to Initial Elections or Subsequent Elections shall be made in one lump sum payment unless the portion of a Participant's Account subject to distribution, as of both the date of the Initial Election or Subsequent Election and the benefit commencement date, has a value of more than \$10,000;

(ii) following a Participant's termination of employment for any reason, if the amount credited to the Participant's Account has a value of \$10,000 or less, the Administrator may, in its sole discretion, direct that such amount be distributed to the Participant (or Beneficiary, as applicable) in one lump sum payment, provided that the payment is made on or before the later of (i) December 31 of the calendar year in which the Participant terminates employment or (ii) the date two and one-half months after the Participant terminates employment.

4.2. Determination of Account Balances for Purposes of Distribution. The amount of any distribution made pursuant to Section 4.1 shall be based on the balances in the Participant's Account on the date of distribution. For this purpose, the balance in a Participant's Account shall be calculated by crediting income,


gains and losses under the Company Stock Fund and Income Fund, as applicable, through the date immediately preceding the date of distribution.

4.3. Plan-to-Plan Transfers. The Administrator may delegate its authority to arrange for plan-to-plan transfers as described in this Section 4.3 to an officer of the Company or committee of two or more officers of the Company.

(a) The Administrator may, with a Participant's consent, make such arrangements as it may deem appropriate to transfer the Company's obligation to pay benefits with respect to such Participant which have not become payable under this Plan, to another employer, whether through a deferred compensation plan, program or arrangement sponsored by such other employer or otherwise, or to another deferred compensation plan, program or arrangement sponsored by the Company or an Affiliate. Following the completion of such transfer, with respect to the benefit transferred, the Participant shall have no further right to payment under this Plan.

(b) The Administrator may, with a Participant's consent, make such arrangements as it may deem appropriate to assume another employer's obligation to pay benefits with respect to such Participant which have not become payable under the deferred compensation plan, program or arrangement under which such future right to payment arose, to the Plan, or to assume a future payment obligation of the Company or an Affiliate under another plan, program or arrangement sponsored by the Company or an Affiliate. Upon the completion of the Plan's assumption of such payment obligation, the Administrator shall establish an Account for such Participant, and the Account shall be subject to the rules of this Plan, as in effect from time to time.

(c) Pursuant to Q-A 19(c) of *IRS Notice* 2005-1, to the extent provided by the Committee or its delegate, a Participant may, on or before December 31, 2005, with respect to all or any portion of his or her account under the 2002 Plan, make new payment elections as to the form and timing of payment of such amounts as may be permitted under this Plan, provided that following the completion of such new payment election, such amounts shall not be treated as grandfathered benefits, subject to the rules of this Plan.

ARTICLE 5-BOOK ACCOUNTS

5.1. Deferred Compensation Account. A deferred Compensation Account shall be established for each Outside Director and Eligible Employee when such Outside Director or Eligible Employee becomes a Participant. Compensation deferred pursuant to the Plan shall be credited to the Account on the date such Compensation would otherwise have been payable to the Participant.

5.2. Crediting of Income, Gains and Losses on Accounts.

(a) In General. Except as otherwise provided in this Section 5.2, the Administrator shall credit income, gains and losses with respect to each Participant's Account as if it were invested in the Income Fund.

(b) Investment Fund Elections. Except for amounts credited to the Accounts of Participants who are Outside Directors who have elected to defer the receipt of Compensation payable in the form of Company Stock, all amounts credited to Participants' Accounts shall be credited with income, gains and losses as if it were invested in the Income Fund.

(c) Outside Director Stock Fund Credits. Amounts credited to the Accounts of Outside Directors in the form of Company Stock shall be credited with income, gains and losses as if they were invested in the Company Stock Fund. No portion of such Participant's Account may be deemed transferred to the Income Fund. Distributions of amounts credited to the Company Stock Fund with respect to Outside Directors' Accounts shall be distributable in the form of Company Stock, rounded to the nearest whole share.

(d) Timing of Credits. Compensation deferred pursuant to the Plan shall be deemed invested in the Income Fund on the date such Compensation would otherwise have been payable to the Participant.

Accumulated Account balances subject to an investment fund election under Section 5.2(b) shall be deemed invested in the applicable investment fund as of the effective date of such election. The value of amounts deemed invested in the Company Stock Fund shall be based on hypothetical purchases and sales of Company Stock at Fair Market Value as of the effective date of an investment election

5.3. Status of Deferred Amounts. Regardless of whether or not the Company is a Participant's employer, all Compensation deferred under this Plan shall continue for all purposes to be a part of the general funds of the Company.

5.4. Participants' Status as General Creditors. Regardless of whether or not the Company is a Participant's employer, an Account shall at all times represent a general obligation of the Company. The Participant shall be a general creditor of the Company with respect to this obligation, and shall not have a secured or preferred position with respect to the Participant's Accounts. Nothing contained herein shall be deemed to create an escrow, trust, custodial account or fiduciary relationship of any kind. Nothing contained herein shall be construed to eliminate any priority or preferred position of a Participant in a bankruptcy matter with respect to claims for wages.

ARTICLE 6-NO ALIENATION OF BENEFITS; PAYEE DESIGNATION

Except as otherwise required by applicable law, the right of any Participant or Beneficiary to any benefit or interest under any of the provisions of this Plan shall not be subject to encumbrance, attachment, execution, garnishment, assignment, pledge, alienation, sale, transfer, or anticipation, either by the voluntary or involuntary act of any Participant or any Participant's Beneficiary or by operation of law, nor shall such payment, right, or interest be subject to any other legal or equitable process. However, subject to the terms and conditions of the Plan, a Participant or Beneficiary may direct that any amount payable pursuant to an Initial Election or a Subsequent Election on any date designated for payment be paid to any person or persons or legal entity or entities, including, but not limited to, an organization exempt from federal income tax under section 501(c)(3) of the Code, instead of to the Participant or Beneficiary. Such a payee designation shall be provided to the Administrator by the Participant or Beneficiary in writing on a form provided by the Administrator, and shall not be effective unless it is provided inimmediately preceding the time of payment. The Company's payment pursuant to such a payee designation shall relieve the Company and its Affiliates of all liability for such payment.

ARTICLE 7-DEATH OF PARTICIPANT

7.1. Death of Participant. A Deceased Participant's Account shall be distributed in accordance with the last Initial Election or Subsequent Election made by the Deceased Participant before the Deceased Participant's death, unless the Deceased Participant's Surviving Spouse or other Beneficiary timely elects to accelerate or defer the time of payment pursuant to Section 3.5.

7.2. Designation of Beneficiaries. Each Participant and Beneficiary shall have the right to designate one or more Beneficiaries to receive distributions in the event of the Participant's or Beneficiary's death by filing with the Administrator a Beneficiary designation on the form provided by the Administrator for such purpose. The designation of a Beneficiary or Beneficiaries may be changed by a Participant or Beneficiary at any time prior to such Participant's or Beneficiary's death by the delivery to the Administrator of a new Beneficiary designation form.

ARTICLE 8—HARDSHIP AND OTHER ACCELERATION EVENTS

8.1. Hardship. Notwithstanding the terms of an Initial Election or Subsequent Election, if, at the Participant's request, the Board determines that the Participant has incurred a Hardship, the Board may, in its discretion, authorize the immediate distribution of all or any portion of the Participant's Account.

8.2. Other Acceleration Events. To the extent permitted by Q-A 15 of *IRS Notice* 2005–1, notwithstanding the terms of an Initial Election or Subsequent Election, distribution of all or part of a Participant's Account may be made:

(a) To the extent necessary to fulfill a domestic relations order (as defined in section 414(p)(1)(B) of the Code).

(b) To the extent necessary to comply with a certificate of divestiture (as defined in section 1043(b)(2) of the Code).

(c) To pay the Federal Insurance Contribution Act ("*FICA*") tax imposed under sections 3101 and 3121(v)(2) of the Code on compensation deferred under the Plan (the "*FICA Amount*") plus the income tax at source on wages imposed under section 3401 of the Code with respect to the FICA Amount, and to pay the additional income tax at source on wages attributable to the pyramiding section 3401 wages and taxes, provided that the total amount distributable under this Section 8.2(c) shall not exceed the sum of the FICA Amount and the income tax withholding related to such FICA Amount.

ARTICLE 9—INTERPRETATION

9.1. Authority of Committee. The Committee shall have full and exclusive authority to construe, interpret and administer this Plan and the Committee's construction and interpretation thereof shall be binding and conclusive on all persons for all purposes.

9.2. Claims Procedure. If an individual (hereinafter referred to as the "Applicant," which reference shall include the legal representative, if any, of the individual) does not receive timely payment of benefits to which the Applicant believes he is entitled under the Plan, the Applicant may make a claim for benefits in the manner hereinafter provided.

An Applicant may file a claim for benefits with the Administrator on a form supplied by the Administrator. If the Administrator wholly or partially denies a claim, the Administrator shall provide the Applicant with a written notice stating:

(a) The specific reason or reasons for the denial;

(b) Specific reference to pertinent Plan provisions on which the denial is based;

(c) A description of any additional material or information necessary for the Applicant to perfect the claim and an explanation of why such material or information is necessary; and

(d) Appropriate information as to the steps to be taken in order to submit a claim for review.

Written notice of a denial of a claim shall be provided within 90 days of the receipt of the claim, provided that if special circumstances require an extension of time for processing the claim, the Administrator may notify the Applicant in writing that an additional period of up to 90 days will be required to process the claim.

If the Applicant's claim is denied, the Applicant shall have 60 days from the date of receipt of written notice of the denial of the claim to request a review of the denial of the claim by the Administrator. Request for review of the denial of a claim must be submitted in writing. The Applicant shall have the right to review pertinent documents and submit issues and comments to the Administrator in writing. The Administrator shall provide a written decision within 60 days of its receipt of the Applicant's request for review, provided that if special circumstances require an extension of time for processing the review of the Applicant's request for review.

It is intended that the claims procedures of this Plan be administered in accordance with the claims procedure regulations of the Department of Labor set forth in 29 CFR § 2560.503-1.

Claims for benefits under the Plan must be filed with the Administrator at the following address:

Comcast Corporation 1500 Market Street Philadelphia, PA 19102 Attention: General Counsel

ARTICLE 10—AMENDMENT OR TERMINATION

10.1. Amendment or Termination. Except as otherwise provided by Section 10.2, the Company, by action of the Board or by action of the Committee, shall have the right at any time, or from time to time, to amend or modify this Plan. The Company, by action of the Board, shall have the right to terminate this Plan at any time.

10.2. Amendment of Rate of Credited Earnings. No amendment shall change the Applicable Interest Rate with respect to the portion of a Participant's Account that is attributable to an Initial Election or Subsequent Election made with respect to Compensation earned in a calendar year and filed with the Administrator before the date of adoption of such amendment by the Board. For purposes of this Section 10.2, a Subsequent Election to defer the payment of part or all of an Account for an additional period after a previously-elected payment date (as described in Section 3.5) shall be treated as a separate Subsequent Election with respect to such Account.

ARTICLE 11-WITHHOLDING OF TAXES

Whenever the Participating Company is required to credit deferred Compensation to the Account of a Participant, the Participating Company shall have the right to require the Participant to remit to the Participating Company an amount sufficient to satisfy any federal, state and local withholding tax requirements prior to the date on which the deferred Compensation shall be deemed credited to the Account of the Participant, or take any action whatever that it deems necessary to protect its interests with respect to tax liabilities. The Participating Company's obligation to credit deferred Compensation to an Account shall be conditioned on the Participant's compliance, to the Participating Company's satisfaction, with any withholding requirement. To the maximum extent possible, the Participating Company shall satisfy all applicable withholding tax requirements by withholding tax from other Compensation payable by the Participating Company to the Participant, or by the Participant's delivery of cash to the Participating Company in an amount equal to the applicable withholding tax.

ARTICLE 12—MISCELLANEOUS PROVISIONS

12.1. No Right to Continued Employment. Nothing contained herein shall be construed as conferring upon any Participant the right to remain in service as an Outside Director or in the employment of a Participating Company as an executive or in any other capacity.

12.2. Expenses of Plan. All expenses of the Plan shall be paid by the Participating Companies.

12.3. Gender and Number. Whenever any words are used herein in any specific gender, they shall be construed as though they were also used in any other applicable gender. The singular form, whenever used herein, shall mean or include the plural form, and vice versa, as the context may require.

12.4. Law Governing Construction. The construction and administration of the Plan and all questions pertaining thereto, shall be governed by the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), and other applicable federal law and, to the extent not governed by federal law, by the laws of the Commonwealth of Pennsylvania.

12.5. Headings Not a Part Hereof. Any headings preceding the text of the several Articles, Sections, subsections, or paragraphs hereof are inserted solely for convenience of reference and shall not constitute a part of the Plan, nor shall they affect its meaning, construction, or effect.

12.6. Severability of Provisions. If any provision of this Plan is determined to be void by any court of competent jurisdiction, the Plan shall continue to operate and, for the purposes of the jurisdiction of that court only, shall be deemed not to include the provision determined to be void.

· ··-

- ----

ARTICLE 13-EFFECTIVE DATE

The effective date of this amendment and restatement of the Plan shall be January 1, 2005. The original effective date of the Plan is January 1, 2005.

IN WITNESS WHEREOF, COMCAST CORPORATION has caused this Plan to be executed by its officers thereunto duly authorized, and its corporate seal to be affixed hereto, as of the 1st day of January, 2005.

COMCAST CORPORATION

- -

BY:

ATTEST:

QuickLinks

-

COMCAST CORPORATION 2005 DEFERRED COMPENSATION PLAN ARTICLE 1-BACKGROUND AND COVERAGE OF PLAN ARTICLE 2-DEFINITIONS ARTICLE 3-INITIAL AND SUBSEQUENT ELECTIONS

· · - ----

· - ·

OuickLinks -- Click here to rapidly navigate through this document

Exhibit 10.6

COMCAST CORPORATION 2002 DEFERRED STOCK OPTION PLAN

ARTICLE 1-CONTINUATION AND COVERAGE OF PLAN

1.1. Freeze of Plan.

(a) COMCAST CORPORATION, a Pennsylvania corporation, hereby amends, restates and freezes the Comcast Corporation 2002 Deferred Stock Option Plan (the "*Plan*"), effective January 30, 2004. The Plan was initially adopted effective September 16, 1997 and was amended and restated effective June 21, 1999, December 19, 2000, November 29, 2001, April 29, 2002, November 18, 2002, February 26, 2003, July 30, 2003 and January 30, 2004.

(b) In order to preserve the favorable tax treatment available to deferrals that were made under the Plan before January 1, 2005 in light of the American Jobs Creation Act of 2004, *IRS Notice 2005-1*, and the regulations issued by the Department of the Treasury thereunder (the "AJCA"), deferrals under the Plan of amounts that were not earned and vested as of December 31, 2004 shall not be permitted. Accordingly, notwithstanding anything in the Plan to the contrary, on or after January 1, 2005, no Participant may make an Initial Election. Initial Elections made by Participants before January 1, 2005 shall be honored only to the extent that under Q-A 16(b) of *IRS Notice 2005-1*, the date of deferral is treated as having occurred before January 1, 2005.

(c) Amounts earned and vested prior to January 1, 2005 are and will remain subject to the terms and conditions of the Plan.

1.2. Plan Unfunded and Limited to Outside Directors and Select Group of Management or Highly Compensated Employees. The Plan is unfunded and is maintained primarily for the purpose of providing outside directors and a select group of management or highly compensated employees the opportunity to defer compensation otherwise payable to such outside directors and management or highly compensated employees. The Plan provides an opportunity for outside directors and management or highly compensated employees. The Plan provides an opportunity for outside directors and management or highly compensated employees to defer the receipt of Shares upon the exercise of Options and to convert the right to receive Shares to the right to receive the cash value thereof as of the date of such conversion, plus interest thereon from the date of such conversion, in accordance with the terms of the Plan.

ARTICLE 2-DEFINITIONS

2.1. "Account" means unfunded bookkeeping accounts established pursuant to Section 5.1 and maintained by the Administrator in the names of the respective Participants (a) to which Deferred Stock Units, dividend equivalents and earnings on dividend equivalents shall be credited with respect to the portion of the Account allocated to the Company Stock Fund and (b) to which an amount equal to the Fair Market Value of Deferred Stock Units with respect to which a Diversification Election has been made and interest thereon from the date of such election shall be credited with respect to the portion of the Account allocated to the Income Fund, and from which all amounts distributed pursuant to the Plan shall be debited.

- 2.2. "Active Participant" means:
 - (a) Each Participant who is in active service as an Outside Director;
 - (b) Each Participant who is actively employed by a Participating Company as an Eligible Employee; and
 - (c) A Permitted Transferee of an individual described in Section 2.2(a) or Section 2.2(b), if applicable.
- 2.3. "Administrator" means the Committee.

2.4. "Affiliate" means, with respect to any Person, any other Person that, directly or indirectly, is in control of, is controlled by, or is under common control with, such Person. For purposes of this definition, the term "control," including its correlative terms "controlled by" and "under common control with," mean,

with respect to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract or otherwise.

2.5. "Annual Rate of Pay" means, as of any date, an employee's annualized base pay rate. An employee's Annual Rate of Pay shall not include sales commissions or other similar payments or awards.

2.6. "Applicable Interest Rate" means:

(a) Except as otherwise provided in Section 2.6(b), the Applicable Interest Rate means 8% per annum, compounded annually as of the last day of the calendar year, or such other interest rate established by the Administrator from time to time. The effective date of any reduction in the Applicable Interest Rate shall not precede the latest of (i) November 29, 2003, (ii) the 30thday following the date of the Administrator's action to establish a reduced rate or (ii) the lapse of 24 full calendar months from the date of the most recent adjustment of the Applicable Interest Rate by the Administrator.

(b) Effective for the period extending from a Participant's employment termination date to the date the Participant's Account is distributed in full, the Administrator, in its sole and absolute discretion, may designate the term "Applicable Interest Rate" for such Participant's Account to mean the lesser of (i) the rate in effect under Section 2.6(a) or (ii) the Prime Rate plus one percent, compounded annually as of the last day of the calendar year. Notwithstanding the foregoing, the Administrator may delegate its authority to determine the Applicable Interest Rate under this Section 2.6(b) to an officer of the Company or committee of two or more officers of the Company.

2.7. "AT&T Broadband Transaction" means the acquisition of AT&T Broadband Corp. (now known as Comcast Cable Communications Holdings, Inc.) by the Company.

2.8. "Beneficiary" means such person or persons or legal entity or entities, including, but not limited to, an organization exempt from federal income tax under section 501(c)(3) of the Code, designated by a Participant or Beneficiary to receive benefits pursuant to the terms of the Plan after such Participant's or Beneficiary's death. If no Beneficiary is designated by the Participant or Beneficiary or if no Beneficiary survives the Participant or Beneficiary (as the case may be), the Participant's Beneficiary shall be the Participant's Surviving Spouse if the Participant has a Surviving Spouse and otherwise the Beneficiary's estate.

2.9. "Board" means the Board of Directors of the Company.

2.10. "Change of Control" means any transaction or series of transactions as a result of which any Person who was a Third Party immediately before such transaction or series of transactions owns then-outstanding securities of the Company such that such Person has the ability to direct the management of the Company, as determined by the Board in its discretion. The Board may also determine that a Change of Control shall occur upon the completion of one or more proposed transactions. The Board's determination shall be final and binding.

2.11. "Code" means the Internal Revenue Code of 1986, as amended.

2.12. "Comcast Option Plan or Plans" means the Comcast Corporation 1987 Stock Option Plan, or the Comcast Corporation 2002 Stock Option Plan, the AT&T Broadband Corp. Adjustment Plan, or any other incentive or non-qualified stock option plan subsequently adopted by the Company or a Related Corporation.

2.13. "Comcast Plan" means any restricted stock, stock bonus, stock option or other compensation plan, program or arrangement established or maintained by the Company or an Affiliate, including, but not limited to this Plan, the Comcast Corporation 2002 Restricted Stock Plan and the Comcast Option Plans.

2.14. "Committee" means the Compensation Committee of the Board of Directors of the Company.

2.15. "Common Stock" means Company's Class A Common Stock, par value \$.01 per share, including a fractional share.

2.16. "Company" means Concast Corporation, a Pennsylvania corporation, as successor to Comcast Corporation, including any successor thereto by merger, consolidation, acquisition of all or substantially all the assets thereof, or otherwise.

2.17. "Company Stock" means Common Stock or such other securities as may be issued by the Company pursuant to adjustments as provided in Article 11.

2.18. "Company Stock Fund" means a hypothetical investment fund pursuant to which Deferred Stock Units are credited with respect to an Option subject to an Initial Election, and thereafter until the date of distribution or the effective date of a Diversification Election, to the extent a Diversification Election applies to such Deferred Stock Units, as applicable. The portion of a Participant's Account deemed invested in the Company Stock Fund shall be treated as if such portion of the Account were invested in hypothetical shares of Common Stock or Special Common Stock otherwise deliverable as Option Shares on the exercise of an Option, and all dividends and other distributions paid with respect to Common Stock or Special Common Stock were held uninvested in cash and credited with interest at the Applicable Interest Rate as of the next succeeding December 31 (to the extent the Account continues to be deemed credited in the form of Deferred Stock Units through such December 31).

2.19. "Date of Grant" means the date as of which an Option is granted.

2.20. "Death Tax Clearance Date" means the date upon which a Deceased Participant's or a deceased Beneficiary's Personal Representative certifies to the Administrator that (a) such Deceased Participant's or deceased Beneficiary's Death Taxes have been finally determined, (b) all of such Deceased Participant's or deceased Beneficiary's Account have been paid in full and (c) all potential liability for Death Taxes with respect to the Deceased Participant's or deceased Beneficiary's Account has been satisfied.

2.21. "Death Taxes" means any and all estate, inheritance, generation-skipping transfer, and other death taxes as well as any interest and penalties thereon imposed by any governmental entity (a "taxing authority") as a result of the death of the Participant or the Participant's Beneficiary.

2.22. "Deceased Participant" means:

(a) A Participant whose employment, or, in the case of a Participant who was an Outside Director, a Participant whose service as an Outside Director, is terminated by death;

- (b) A Participant who dies following termination of active employment or active service; or
- (c) A Permitted Transferee of an individual described in Section 2.22(a) or 2.22(b), if applicable.

2.23. "Deferred Stock Units" mean the number of hypothetical Shares determined as the excess of (a) the number of Option Shares over (b) the number of Other Available Shares having a Fair Market Value as of the date of exercise of an Option equal to the exercise price for such Option Shares (hereinafter referred to in this Section 2.23 as the "Payment Shares"), as to which an Outside Director, Former Outside Director, Eligible Employee, Former Eligible Employee or Successor-in-Interest provides to the Company evidence of ownership of sufficient Payment Shares to pay the exercise price for such Option Shares; provided, however, that if the Option is for Common Stock, the Deferred Stock Units shall be credited to the Participant's Account as Deferred Common Stock Units. Provision of a notarized statement under oath to the Company by the Outside Director, Former Outside Director, Eligible Employee, Former Eligible Employee or Successor-in-Interest attesting to the number of Payment Shares owned by the Outside Director, Former Outside Director, Eligible Employee, Former Eligible Employee, Former Eligible Employee or Successor-in-Interest and held by a securities broker for the Outside Director, Former Outside Director, Eligible Employee, Former Eligible Employee, Former Eligible Employee or Successor-in-Interest in "street name" or provision of the certificate numbers to the Company by the Outside Director, Former Outside Director, Eligible Employee, Former Outside Director, Eligible Employee, Former Eligible Employee, Former Outside Director, Eligible Employee, Former Eligible Employee, Former Outside Director, Former Outside Director, Eligible Employee, Former Eligible Employee, Former Outside Director, Former O

Eligible Employee, Former Eligible Employee or Successor-in-Interest of the Payment Share stock certificates actually held by the Outside Director, Former Outside Director, Eligible Employee, Former Eligible Employee or Successor-in-Interest shall constitute acceptable evidence of ownership.

2.24. "Disabled Participant" means

(a) A Participant whose employment or, in the case of a Participant who is an Outside Director, a Participant whose service as an Outside Director, is terminated by reason of disability;

(b) A Participant who becomes disabled (as determined by the Committee) following termination of active service;

- (c) The duly-appointed legal guardian of an individual described in Section 2.24(a) or 2.24(b) acting on behalf of such individual; or
- (d) A Permitted Transferee of an individual described in Section 2.24(a) or 2.24(b), if applicable.

2.25. "Diversification Election" means a Participant's election to have a portion of the Participant's Account credited in the form of Deferred Stock Units under the Company Stock Fund deemed liquidated and credited thereafter under the Income Fund, as provided in Section 5.2(b).

2.26. "Eligible Employee" means:

(a) Each employee of a Participating Company whose Annual Rate of Pay is \$200,000 or more as of both (i) the date on which an Initial Election is filed with the Administrator and (ii) the first day of the calendar year in which such Initial Election is filed

(b) Each employee of a Participating Company who has an Annual Rate of Pay of \$125,000 as of each of (i) June 30, 2002; (ii) the date on which an Initial Election is filed with the Administrator and (iii) the first day of each calendar year beginning after December 31, 2002.

(c) Each New Key Employee.

(d) Each other employee of a Participating Company who is designated by the Committee, in its sole and absolute discretion, as an Eligible Employee.

2.27. "Fair Market Value" shall mean:

(a) If Shares are listed on a stock exchange, Fair Market Value shall be determined based on the last reported sale price of a Share on the principal exchange on which Shares are listed on the date of determination, or if such date is not a trading day, the next trading date.

(b) If Shares are not so listed, but trades of Shares are reported on the Nasdaq National Market, Fair Market Value shall be determined based on the last quoted sale price of a Share on the Nasdaq National Market on the date of determination, or if such date is not a trading day, the next trading date.

(c) If Shares are not so listed nor trades of Shares so reported, Fair Market Value shall be determined by the Committee in good faith.

2.28. "Former Eligible Employee" means an individual who has ceased to be actively employed by a Participating Company for any reason but who, immediately preceding his termination of employment, was an Eligible Employee.

2.29. "Former Outside Director" means an individual who has ceased to be a member of the Board, but who, immediately preceding his cessation of service as a member of the Board was an Outside Director.

2.30. "Immediate Family" means an Outside Director's, Former Outside Director's, Eligible Employee's or Former Eligible Employee's spouse and lineal descendants, any trust all beneficiaries of which are any of such persons and any other entity all members or owners of which are any of such persons.



2.31. "Income Fund" means a hypothetical investment fund pursuant to which an amount equal to the Fair Market Value of Deferred Stock Units subject to a Diversification Election is credited as of the effective date of such Diversification Election and as to which interest is credited thereafter until the date of distribution at the Applicable Interest Rate.

2.32. "Initial Election" means a written election on a form provided by the Administrator, filed with the Administrator in accordance with Article 3, pursuant to which an Outside Director, Former Outside Director, Eligible Employee, Former Eligible Employee, Successor-in-Interest or Permitted Transferee who:

(a) Elects, within the time or times specified in Article 3, to defer the receipt of Shares pursuant to the exercise of all or part of an Option; and

(b) Designates the time that such Shares and any dividend equivalents shall be distributed.

2.33. "New Key Employee"

(a) means each employee of a Participating Company:

(b) Who becomes an employee of a Participating Company and has an Annual Rate of Pay of \$200,000 or more as of his employment commencement date; and

(c) Who has an Annual Rate of Pay that is increased to \$200,000 or more and who, immediately preceding such increase, was not an Eligible Employee.

2.34. "Normal Retirement" means:

(a) For a Participant who is an employee of a Participating Company immediately preceding his termination of employment, a termination of employment that is treated by the Participating Company as a retirement under its employment policies and practices as in effect from time to time; and

(b) For a Participant who is an Outside Director immediately preceding his termination of service, his normal retirement from the Board.

2.35. "Option" means a non-qualified stock option to purchase Shares granted pursuant to an Comcast Option Plan; provided that each Option with a different Date of Grant shall be considered a separate Option.

2.36. "Option Shares" mean the Shares that are subject to the portion of an Option as to which an Initial Election or Subsequent Election is in effect as adjusted to reflect a Share Withholding Election.

2.37. "Other Available Shares" means, as of any date, the sum of:

(a) the total number of Shares owned by a Participant that were not acquired by such Participant pursuant to a Comcast Plan or otherwise in connection with the performance of services to the Company or an Affiliate; plus

(b) the excess, if any, of:

- (i) the total number of Shares owned by a Participant other than the Shares described in Section 2.37(a); over
- (ii) the excess, if any of:
 - (A) The sum of:
 - (1) The number of such Shares owned by such Participant for less than six months; plus

(2) The number of such Shares owned by such Participant that has, within the preceding six months, been the subject of a withholding certification under any Comcast Plan; plus

(3) The number of such Shares owned by such Participant that has, within the preceding six months, been received in exchange for Shares surrendered as payment, in full or in part, or as to which ownership was attested to as payment, in full or in part, of the exercise price for an option to purchase any securities of the Company or an Affiliate of the Company, under any Comcast Plan, but only to the extent of the number of Shares surrendered or attested to; plus

(4) The number of such Shares owned by such Participant as to which evidence of ownership has, within the preceding six months, been provided to the Company in connection with the crediting of Deferred Stock Units to such Participant's Account.

For purposes of this Section 2.37, a Share that is subject to a deferral election pursuant to this Plan or another Comcast Plan shall not be treated as owned by a Person until all conditions to the delivery of such Share have lapsed. The number of Other Available Shares shall be determined separately for Common Stock and Special Common Stock. For purposes of determining the number of Other Available Shares, the term "Shares" shall also include the securities held by a Participant immediately before the consummation of the AT&T Broadband Transaction that became Common Stock and Special Common Stock as a result of the AT&T Broadband Transaction.

2.38. "Outside Director" means a member of the Board, who is not an employee of a Participating Company.

2.39. "Participant" means each Outside Director, Former Outside Director, Eligible Employee, Former Eligible Employee, Successor-in-Interest or Permitted Transferee who is the grantee or transferee of an Option that has made an Initial Election or Subsequent Election and that has an undistributed amount credited to an Account under the Plan.

2.40. "Participating Company" means Concast Corporation and each Related Corporation with respect to Concast Corporation. Effective January 1, 2003, "Participating Company" means the Company and each Related Corporation.

2.41. "Permitted Transferee" means a member of the Immediate Family of an Outside Director, Former Outside Director, Eligible Employee or Former Eligible Employee to whom the right to exercise an Option has been transferred pursuant to an Comcast Option Plan.

2.42. "Person" means an individual, a corporation, a partnership, an association, a trust or any other entity or organization.

2.43. "Personal Representative" means the executor, the administrator, or the personal representative of a deceased individual's estate.

2.44. "Plan" means the Concast Corporation 2002 Deferred Stock Option Plan, as set forth herein, and as amended from time to time.

2.45. "Prime Rate" means the annual rate of interest identified by PNC Bank as its prime rate as of the first day of each calendar year.

2.46. "Related Corporation" means a subsidiary of Comcast Corporation or, effective January 1, 2003, a subsidiary of the Company, as defined in section 424(f) of the Code.

2.47. "Retired Participant" means a Participant who has terminated employment pursuant to a Normal Retirement.

2.48. "Share" or "Shares."

(a) Except as provided in this Section 2.48, a share or shares Common Stock or Special Common Stock.

(b) The term "Share" or "Shares" also means such other securities issued by the Sponsor as may be the subject of an adjustment under Section 11, or for purposes of Section 2.37 and Section 10, as

may have been the subject of a similar adjustment under similar provisions of a Comcast Plan as now in effect or as may have been in effect before the AT&T Broadband Transaction.

2.49. "Share Withholding Election" means a written election on a form provided by the Administrator, filed with the Administrator in accordance with the rules applicable to the filing of Initial Elections under Article 3, pursuant to which an Eligible Employee, Former Eligible Employee, Successor-in-Interest or Permitted Transferee elects to have the number of Shares deferred pursuant to the exercise of all or part of an Option and credited under the Plan as Deferred Stock Units adjusted so that Deferred Stock Units that would, but for a Share Withholding Election, be credited to an Account under the Plan, shall be deemed distributed pursuant to the Plan to satisfy applicable withholding tax liabilities, as described in Section 10.2.

2.50. "Special Common Stock" means the Company's Class A Special Common Stock, par value \$.01 per share, including a fractional share.

2.51. "Special Diversification Election" means a Diversification Election by a Participant other than an Outside Director to have more than 40 percent of the Deferred Stock Units credited to such Participant's Account in the Company Stock Fund that are attributable to any Option deemed liquidated and credited thereafter under the Income Fund, as provided in Section 5.2(b), if (and to the extent that) it is approved by the Administrator in accordance with Section 5.2(b). An Outside Director may not make a Special Diversification Election.

2.52 "Subsequent Election" means a written election on a form provided by the Administrator, filed with the Administrator in accordance with Article 3, pursuant to which a Participant or Beneficiary may elect to defer (or, in limited cases, accelerate) the time of receipt of amounts credited to an Account previously deferred in accordance with the terms of a previously made Initial Election or Subsequent Election.

2.53 "Successor-in-Interest" means the Beneficiary of a deceased Former Outside Director, a deceased Former Eligible Employee or another deceased Participant, to whom the right to exercise an Option or the right to payment under the Plan shall have passed, as applicable.

2.54 "Surviving Spouse" means the widow or widower, as the case may be, of a Deceased Participant or a Deceased Beneficiary (as applicable).

- 2.55 "Terminating Event" means either of the following events:
 - (a) the liquidation of the Company; or
 - (b) a Change of Control.

2.56 "Third Party" means any Person, together with such Person's Affiliates, provided that the term "Third Party" shall not include the Company or an Affiliate of the Company.

ARTICLE 3-INITIAL AND SUBSEQUENT ELECTIONS

3.1. Elections.

(a) Initial Elections. Each Outside Director, Former Outside Director, Eligible Employee, Former Eligible Employee, Successor-in-Interest or Permitted Transferee who is the grantee or transferee of an Option, shall have the right to make an Initial Election to defer the receipt of Shares upon exercise of all or part of such Option by filing an Initial Election at the time and in the manner described in this Article 3. Unless otherwise specifically provided in the Initial Election, following a Diversification Election, an Initial Election shall apply to the portion of a Participant's Account credited to the Income Fund on the same basis as the portion of such Participant's Account credited to the Company Stock Fund.

(b) Subsequent Elections. Each Participant and Beneficiary shall have the right to elect to defer (or, in limited cases, accelerate) the time of receipt of amounts previously deferred in accordance with the terms of a previously made Initial Election by filing a Subsequent Election at the time, to the extent, and in the manner described in this Article 3. Unless otherwise specifically provided in the Subsequent

Election, a Subsequent Election shall apply to the portion of a Participant's Account credited to the Income Fund on the same basis as the portion of such Participant's Account credited to the Company Stock Fund.

3.2. Filing of Initial Election: General. An Initial Election shall be made on the form provided by the Administrator for this purpose. No such Initial Election shall be effective unless it is filed with the Administrator on or before a date that is both (i) more than six (6) months prior to the exercise of such Option and (ii) in the calendar year preceding the calendar year in which such Option is exercised, provided that an Initial Election filed with the Administrator on or before December 31, 1997, shall be effective with respect to the exercise of any Option after December 31, 1997.

3.3. Options to which Initial Elections May Apply. A separate Initial Election may be made for each Option, or a portion of such Option, with respect to which an Outside Director, Former Outside Director, Eligible Employee, Former Eligible Employee, Successor-in-Interest or Permitted Transferee desires to defer receipt of Shares upon exercise of all or a portion of such Option. The failure of such a Person to make an Initial Election with respect to an Option shall not affect such Person's right to make an Initial Election for any other Option.

3.4. Initial Election of Distribution Date. Each Participant who elects to defer the receipt of Shares shall, on the Initial Election, also elect the distribution date for such Shares or any corresponding amounts which may be credited to the Income Fund following a Diversification Election; provided, however, that subject to acceleration pursuant to Section 3.5(d), Section 3.5(e), Section 3.6, Section 3.7, Section 3.8 or Section 7.1, no distribution may be made earlier than January 2nd of the third calendar year beginning after the date of the Initial Election nor later than January 2nd of the eleventh calendar year beginning after the date of the Initial Election.

3.5. Subsequent Elections.

(a) Active Participants. Each Active Participant who has made an Initial Election, or who has made a Subsequent Election pursuant to this Section 3.5(a), may elect to defer the time of payment of part or all of such Active Participant's Account for a minimum of two and a maximum of ten additional years from the previously-elected payment date, by filing a Subsequent Election with the Administrator on or before the close of business on June 30 of the calendar year preceding the calendar year in which the distribution would otherwise be made. The number of Subsequent Elections under this Section 3.5(a) shall not be limited.

(b) Surviving Spouses.

(i) General Rule. A Surviving Spouse who is a Deceased Participant's Beneficiary may elect to defer the time of payment of any part or all of such Deceased Participant's Account the payment of which would be made neither within six (6) months after, nor within the calendar year of, the date of such election. Such election shall be made by filing a Subsequent Election with the Administrator in which the Surviving Spouse shall specify the change in the time of payment, which shall be no less than two nor more than ten years from the previously-elected payment date, or such Surviving Spouse may elect to defer payment until such Surviving Spouse's death. A Surviving Spouse may make a total of two (2) Subsequent Elections under this Section 3.5(b)(i) with respect to all or any part of the Deceased Participant's Account. Subsequent Elections pursuant to this Section 3.5(b)(i) may specify different changes with respect to different parts of the Deceased Participant's Account.

(ii) Exception. Notwithstanding the above Section 3.5(b)(i), a Subsequent Election may be made by a Surviving Spouse within sixty (60) days of the Deceased Participant's death; provided, however, such election may only be made with respect to amounts which would not be paid under the Deceased Participant's election as in effect on the date of the Deceased Participant's death until a date which is at least six (6) months from the Deceased Participant's date of death. Such election shall be made by filing a Subsequent Election with the Administrator in which the Surviving Spouse

shall specify the change in the time of payment, which shall be no less than two (2) nor more than ten (10) years from the previously-elected payment date, or such Surviving Spouse may elect to defer payment until such Surviving Spouse's death. A Surviving Spouse may only make one (1) Subsequent Election under this Section 3.5(b)(ii) with respect to all or any part of the Deceased Participant's Account. Such Surviving Spouse may, however, make one additional Subsequent Election under Section 3.5(b)(i) in accordance with the terms of Section 3.5(b)(i). The one (1) Subsequent Election permitted under this Section 3.5(b)(ii) may specify different changes for different parts of the Deceased Participant's Account.

(c) Beneficiary of a Deceased Participant Other Than a Surviving Spouse.

(i) General Rule. A Beneficiary of a Deceased Participant (other than a Surviving Spouse) may elect to defer the time of payment, of any part or all of such Deceased Participant's Account the payment of which would be made neither within six (6) months after, nor within the calendar year of, the date of such election. Such election shall be made by filing a Subsequent Election with the Administrator in which the Beneficiary shall specify the change in the time of payment, which shall be no less than two (2) nor more than ten (10) years from the previously-elected payment date. A Beneficiary may make one (1) Subsequent Election under this Section 3.5(c)(i), with respect to all or any part of the Deceased Participant's Account. Subsequent Elections pursuant to this Section 3.5(c)(i) may specify different changes for different parts of the Deceased Participant's Account.

(ii) Exception. Notwithstanding the above Section 3.5(c)(i), a Subsequent Election may be made by a Beneficiary within sixty (60) days of the Deceased Participant's death; provided, however, such election may only be made with respect to amounts which would not be paid under the Deceased Participant's election as in effect on the date of the Deceased Participant's death until a date which is at least six (6) months from the Deceased Participant's date of death. Such election shall be made by filing a Subsequent Election with the Administrator in which the Beneficiary shall specify the change in the time of payment, which shall be no less than two (2) nor more than ten (10) years from the previously-elected payment date. A Beneficiary may make one (1) Subsequent Election under this Section 3.5(c)(ii) with respect to all or any part of the Deceased Participant's Account. Subsequent Elections pursuant to this Section 3.5(c)(ii) may specify different changes for different parts of the Deceased Participant's Account.

(d) Other Deferral and Acceleration by a Beneficiary. Any Beneficiary (other than a Surviving Spouse who has made a Subsequent Election under Section 3.5(b) or a Beneficiary who has made a Subsequent Election under Section 3.5(c)) may elect to:

(i) Defer the time of payment of any part or all of the Deceased Participant's Account or deceased Beneficiary's Account for one additional year from the date payment would otherwise be made (provided that if a Subsequent Election is made pursuant to this Section 3.5(d)(i), the Deceased Participant's Account or deceased Beneficiary's Account shall be in all events distributed in full on or before the fifth anniversary of the Deceased Participant's or deceased Beneficiary's death); or

(ii) Accelerate the time of payment of a Deceased Participant's Account or deceased Beneficiary's Account from the date or dates that payment would otherwise be made to the date that is the later of (A) six (6) months after the date of the Deceased Participant's or deceased Beneficiary's death and (B) January 2nd of the calendar year beginning after the Deceased Participant's or deceased Beneficiary's death, provided that if a Subsequent Election is made pursuant to this Section 3.5(d)(ii), the Deceased Participant's Account or deceased Beneficiary's Account shall be distributed in full on such accelerated payment date.

A Subsequent Election pursuant to this Section 3.5(d) must be filed with the Administrator within one hundred twenty (120) days following the Deceased Participant's or deceased Beneficiary's death. One and only one Subsequent Election shall be permitted pursuant to this Section 3.5(d) with respect to a

Deceased Participant's Account or deceased Beneficiary's Account, although if such Subsequent Election is filed pursuant to Section 3.5(d)(i), it may specify different changes for different parts of the Account.

(e) Acceleration by Disabled Participant or Permitted Transferee of Disabled Participant. A Disabled Participant, or the Permitted Transferee of a Disabled Participant if applicable, may elect to accelerate the time of payment of the Disabled Participant's Account from the date payment would otherwise be made to January 2nd of the calendar year beginning after the Participant became disabled. A Subsequent Election pursuant to this Section 3.5(e) must be filed with the Administrator on or before the close of business on the later of (i) the June 30 following the date the Participant becomes a Disabled Participant becomes a Disabled Participant on or before May 1 of a calendar year, (ii) the 60th day following the date the Participant becomes a Disabled Participant after May 1 and before November 2 of a calendar year or (iii) the December 31 following the date the Participant becomes a Disabled Participant if the Participant becomes a Disabled Participant if a calendar year.

(f) Retired Participants and Disabled Participants. The Committee may, in its sole and absolute discretion, permit a Retired Participant or a Disabled Participant to make a Subsequent Election to defer the time of payment of any part or all of such Retired or Disabled Participant's Account for a minimum of two years and a maximum of ten additional years from the previously-elected payment date, by filing a Subsequent Election with the Administrator on or before the close of business on June 30 of the calendar year preceding the calendar year in which the lump-sum distribution or initial installment payment would otherwise be made. The number of Subsequent Elections under this Section 3.5(f) shall be determined by the Committee in its sole and absolute discretion.

(g) Retired Participant or Permitted Transferee of Retired Participant. A Retired Participant (who has not been permitted to make a Subsequent Election under Section 3.5(f)) or a Permitted Transferee of a Retired Participant may elect to defer the time of payment of the Retired Participant's Account for a minimum of two additional years from the date payment would otherwise be made (provided that if a Subsequent Election is made pursuant to this Section 3.5(g), the Retired Participant's Account shall be distributed in full on or before the fifth anniversary of the Retired Participant's Normal Retirement). A Subsequent Election pursuant to this Section 3.5(g) must be filed with the Administrator on or before the close of business on the later of (i) the June 30 following the Participant's Normal Retirement on or before May 1 of a calendar year, (ii) the 60th day following the Participant's Normal Retirement after May 1 and before November 2 of a calendar year or (iii) the December 31 following the Participant's Normal Retirement after November 1 of a calendar year.

(h) Disabled Participant or Permitted Transferee of Disabled Participant. A Disabled Participant (who has not been permitted to make a Subsequent Election under 3.5(f)) or a Permitted Transferee of a Disabled Participant may elect to defer the time of payment of the Disabled Participant's Account for a minimum of two additional years from the date payment would otherwise be made (provided that if a Subsequent Election is made pursuant to this Section 3.5(h), the Disabled Participant's Account shall be distributed in full on or before the fifth anniversary of the date the Participant became a Disabled Participant). A Subsequent Election pursuant to this Section 3.5(h) must be filed with the Administrator on or before the close of business on the later of (i) the June 30 following the date the Participant becomes a Disabled Participant on or before May 1 of a calendar year, (ii) the 60th day following the date the Participant becomes a Disabled Participant after May 1 and before November 2 of a calendar year or (iii) the December 31 following the date the Participant becomes a Disabled Participant if the Participant if the Participant if the Participant if the Participant becomes a Disabled Participant if the Participant after May 1 and before November 2 of a calendar year or (iii) the December 31 following the date the Participant after November 1 of a calendar year.

(i) Most Recently Filed Initial Election or Subsequent Election Controlling. Subject to acceleration pursuant to Section 3.5(d), or 3.5(e), Section 3.6 or 7.1, no distribution of the amounts deferred pursuant to this Article 3 for any calendar year shall be made before the distribution date designated by the Participant or Beneficiary, Permitted Transferee or Successor-in-Interest, as applicable, on the most recently filed Initial Election or Subsequent Election with respect to each deferred amount.

3.6. Distribution in Full upon Terminating Event. The Company shall give Participants at least thirty (30) days notice (or, if not practicable, such shorter notice as may be reasonably practicable) prior to the anticipated date of the consummation of a Terminating Event. The Company may, in its sole and absolute discretion, provide in such notice that notwithstanding any other provision of the Plan or the terms of any Initial or Subsequent Election, upon the consummation of a Terminating Event, the Account balance of each Participant shall be distributed in full and any outstanding Initial Elections or Subsequent Elections shall be revoked.

3.7. Withholding and Payment of Death Taxes.

(a) Notwithstanding any other provisions of this Plan to the contrary, including but not limited to the provisions of Article 3 and Article 7, or any Initial or Subsequent Election filed by a Deceased Participant or a Deceased Participant's Beneficiary (for purposes of this Section, the *"Decedent"*), the Administrator shall apply the terms of Section 3.7(b) to the Decedent's Account unless the Decedent affirmatively has elected, in writing, filed with the Administrator, to waive the application of Section 3.7(b).

(b) Unless the Decedent affirmatively has elected, pursuant to Section 3.7(a), that the terms of this Section 3.7(b) not apply:

(i) The Administrator shall prohibit the Decedent's Beneficiary from taking any action under any of the provisions of the Plan with regard to the Decedent's Account other than the Beneficiary's making of a Subsequent Election pursuant to Section 3.5;

(ii) The Administrator shall defer payment of the Decedent's Account until the later of the Death Tax Clearance Date and the payment date designated in the Decedent's Initial Election or Subsequent Election;

(iii) The Administrator shall withdraw from the Decedent's Account such amount or amounts as the Decedent's Personal Representative shall certify to the Administrator as being necessary to pay the Death Taxes apportioned against the Decedent's Account; the Administrator shall remit the amounts so withdrawn to the Personal Representative, who shall apply the same to the payment of the Decedent's Death Taxes, or the Administrator may pay such amounts directly to any taxing authority as payment on account of Decedent's Death Taxes, as the Administrator elects; (iv) If the Administrator makes a withdrawal from the Decedent's Account to pay the Decedent's Death Taxes and such withdrawal causes the recognition of income to the Beneficiary, the Administrator shall pay to the Beneficiary from the Decedent's Account, within thirty (30) days of the Beneficiary's request, the amount necessary to enable the Beneficiary to pay the Beneficiary's income tax liability resulting from such recognition of income; additionally, the Administrator shall pay to the Beneficiary from the Decedent's Account, within thirty (30) days of the Beneficiary's request, such additional amounts as are required to enable the Beneficiary to pay the Beneficiary's income tax liability attributable to the Beneficiary's recognition of income resulting from a distribution from the Decedent's Account pursuant to this Section 3.7(b)(iv);

(v) Amounts withdrawn from the Decedent's Account by the Administrator pursuant to Sections 3.7(b)(iii) and 3.7(b)(iv) shall be withdrawn from the portions of Decedent's Account having the earliest distribution dates as specified in Decedent's Initial Election or Subsequent Election; and

(vi) Within a reasonable time after the later to occur of the Death Tax Clearance Date and the payment date designated in the Decedent's Initial Election or Subsequent Election, the Administrator shall pay the Decedent's Account to the Beneficiary.

3.8. Effect of Distribution within Five Years of Effective Date of Diversification Election. If, pursuant to Section 3.1 through 3.7, Shares distributable with respect to Deferred Stock Units credited to the Company Stock Fund that are attributable to the Option as to which a Diversification Election was made are distributed on or before the fifth anniversary of the effective date of such Diversification Election (and, in the case of a Participant who is a Successor-in-Interest or a Permitted Transferee, whether or not such Diversification Election was made by a Participant's predecessor-in-interest), then, except as may otherwise be provided by the Committee in its sole and absolute discretion, the following percentage of the Participant's Account credited to the Income Fund and attributable to such Diversification Election shall be distributed simultaneously with such Shares, without regard to any election to the contrary:

Time that Shares are Distributable	Distributable Percentage of Corresponding Income Fund Amount
On or before the third anniversary of a Diversification Election	60%
After the third anniversary of a Diversification Election and on or before the fourth anniversary of a Diversification Election	40%
After the fourth anniversary of a Diversification Election and on or before the fifth anniversary of a Diversification Election	20%
After the fifth anniversary of a Diversification Election	0%

ARTICLE 4—MANNER OF DISTRIBUTION

4.1. Manner of Distribution.

(a) Deferred Stock Units credited to an Account shall be distributed in a lump sum in shares of Common Stock and/or Special Common Stock, as applicable. Dividend equivalents shall be distributed in a lump sum in cash. Amounts credited to the Income Fund pursuant to a Diversification Election shall be distributed in a lump sum in cash.

(b) Notwithstanding any Initial Election or Subsequent Election or any other provision of the Plan to the contrary, following a Participant's termination of employment for any reason, if the amount credited to the Participant's Account has a value of \$25,000 or less, the Administrator may, in its sole discretion, direct that such amount be distributed to the Participant (or Beneficiary, as applicable) in one lump sum payment; provided, however, that this Section 4.1(b) shall not apply to any amount credited to a Participant's Account until the expiration of the deferral period applicable under any Initial Election or Subsequent Election in effect as of April 29, 2002.

ARTICLE 5—BOOK ACCOUNTS

5.1. Account. An Account shall be established for each Outside Director, Former Outside Director, Eligible Employee, Former Eligible Employee, Successor-in-Interest or Permitted Transferee when such Person becomes a Participant. Deferred Stock Units shall be credited to the Account as of the date of exercise of an Option as to which an Initial or Subsequent Election is in effect. Each Deferred Stock Unit that represented a hypothetical share of Comcast Corporation Class A Common Stock, par value \$1.00 immediately before the consummation of the AT&T Broadband Transaction shall be treated as a hypothetical share of Common Stock. Each Deferred Stock Unit that represented a hypothetical share of Comcast Corporation Class A Special Common Stock, par value \$1.00 shall be treated as a hypothetical share of Special Common Stock. To the extent an Account is deemed invested in the Income Fund, the Administrator shall credit earnings with respect to such Account at the Applicable Interest Rate, as further provided in Section 5.2.

5.2. Crediting of Income, Gains and Losses on Accounts.

(a) In General. Except as otherwise provided in this Section 5.2, the value of a Participant's Account as of any date shall be determined as if it were invested in the Company Stock Fund.

(b) Diversification Elections.

(i) In General. A Diversification Election shall be available (A) at any time that a Registration Statement filed under the Securities Act of 1933, as amended (a "Registration Statement"), is effective with respect to the Plan and (B) with respect to a Special Diversification Election, if and to the extent that the opportunity to make such a Special Diversification Election has been approved by the Administrator. No approval is required for a Diversification Election other than a Special Diversification Election.

(ii) Administrator Approval of Special Diversification Elections. The opportunity to make a Special Diversification Election and the extent to which a Special Diversification Election applies to Deferred Stock Units credited to the Company Stock Fund may be approved or rejected by the Administrator in its sole discretion. A Special Diversification Election shall only be effective if (and to the extent) approved by the Administrator.

(iii) Conversion of Deferred Stock Units to Cash Equivalents. Each Participant who is an Outside Director, Former Outside Director, Eligible Employee, Former Eligible Employee, Successor-in-Interest or Permitted Transferee may make a Diversification Election to convert up to 40 percent (or in the case of a Special Diversification Election, up to the approved percentage) of Deferred Stock Units credited to the Company Stock Fund that are attributable to any Option to the Income Fund. No deemed transfers shall be permitted from the Income Fund to the Company Stock Fund. Diversification Elections under this Section 5.2(b) shall be prospectively effective on the later of (A) the date designated by the Participant on a Diversification Election filed with the Administrator or (B) the business day next following the lapse of six months from the date Deferred Stock Units subject to the Diversification Election are credited to the Participant's Account.

(c) Timing of Credits. Account balances subject to a Diversification Election under Section 5.2(b) shall be deemed transferred from the Company Stock Fund to the Income Fund as of the effective date of such Diversification Election. The value of amounts deemed invested in the Income Fund immediately following the effective date of a Diversification Election shall be based on hypothetical sales of Company Stock underlying liquidated Deferred Stock Units at Fair Market Value as of the effective date of a Diversification Election.

5.3. Status of Deferred Amounts. Regardless of whether or not the Company is a Participant's employer, all amounts deferred under this Plan shall continue for all purposes to be a part of the general funds of the Company.

5.4. Participants' Status as General Creditors. Regardless of whether or not the Company is a Participant's employer, an Account shall at all times represent a general obligation of the Company. The



Participant shall be a general creditor of the Company with respect to this obligation, and shall not have a secured or preferred position with respect to the Participant's Accounts. Nothing contained herein shall be deemed to create an escrow, trust, custodial account or fiduciary relationship of any kind. Nothing contained herein shall be construed to eliminate any priority or preferred position of a Participant in a bankruptcy matter with respect to claims for wages.

ARTICLE 6-NONALIENATION OF BENEFITS

6.1. Alienation Prohibited. Except as otherwise required by applicable law, the right of any Participant or Beneficiary to any benefit or interest under any of the provisions of this Plan shall not be subject to encumbrance, attachment, execution, garnishment, assignment, pledge, alienation, sale, transfer, or anticipation, either by the voluntary or involuntary act of any Participant or any Participant's Beneficiary or by operation of law, nor shall such payment, right, or interest be subject to any other legal or equitable process.

ARTICLE 7-DEATH OF PARTICIPANT

7.1. Death of Participant. Except as provided in Section 3.7, a Deceased Participant's Account shall be distributed in accordance with the last Initial Election or Subsequent Election made by the Deceased Participant before the Deceased Participant's death, unless the Deceased Participant's Surviving Spouse, Permitted Transferee, Successor-in-Interest or Beneficiary timely elects to accelerate or defer the time of payment pursuant to Section 3.5(b), Section 3.5(c), Section 3.5(d), Section 3.5(e), or Section 3.5(f).

7.2. Designation of Beneficiaries. Each Participant and Beneficiary shall have the right to designate one or more Beneficiaries to receive distributions in the event of the Participant's or Beneficiary's death by filing with the Administrator a Beneficiary designation on the form provided by the Administrator for such purpose. The designation of a Beneficiary or Beneficiaries may be changed by a Participant or Beneficiary at any time prior to such Participant's or Beneficiary's death by the delivery to the Administrator of a new Beneficiary designation form.

ARTICLE 8—INTERPRETATION

8.1. Authority of Committee. The Committee shall have full and exclusive authority to construct interpret and administer this Plan and the Committee's construction and interpretation thereof shall be binding and conclusive on all persons for all purposes.

8.2. Claims Procedure. If an individual (hereinafter referred to as the "Applicant," which reference shall include the legal representative, if any, of the individual) does not receive timely payment of benefits to which the Applicant believes he is entitled under the Plan, the Applicant may make a claim for benefits in the manner hereinafter provided.

An Applicant may file a claim for benefits with the Administrator on a form supplied by the Administrator. If the Administrator wholly or partially denies a claim, the Administrator shall provide the Applicant with a written notice stating:

- (a) The specific reason or reasons for the denial;
- (b) Specific reference to pertinent Plan provisions on which the denial is based;

(c) A description of any additional material or information necessary for Applicant to perfect the claim and an explanation of why such material or information is necessary; and

(d) Appropriate information as to the steps to be taken in order to submit a claim for review.

Written notice of a denial of a claim shall be provided within 90 days of the receipt of the claim, provided that if special circumstances require an extension of time for processing the claim, the Administrator may notify the Applicant in writing that an additional period of up to 90 days will be required to process the claim.

If the Applicant's claim is denied, the Applicant shall have 60 days from the date of receipt of written notice of the denial of the claim to request a review of the denial of the claim by the Administrator. Request for review of the denial of a claim must be submitted in writing. The Applicant shall have the right to review pertinent documents and submit issues and comments to the Administrator in writing. The Administrator shall provide a written decision within 60 days of its receipt of the Applicant's request for review, provided that if special circumstances require an extension of time for processing the review of the Applicant's request for review.

It is intended that the claims procedures of this Plan be administered in accordance with the claims procedure regulations of the Department of Labor set forth in 29 CFR § 2560.503-1.

Claims for benefits under the Plan must be filed with the Administrator at the following address:

Comcast Corporation 1500 Market Street Philadelphia, PA 19102 Attention: General Counsel

ARTICLE 9-AMENDMENT OR TERMINATION

9.1. Amendment or Termination. The Company, by action of the Board or by action of the Committee, shall have the right at any time, or from time to time, to amend or modify this Plan. The Company, by action of the Board, shall have the right to terminate this Plan at any time.

ARTICLE 10—WITHHOLDING OF TAXES ON EXERCISE OF OPTION

10.1. In General. Whenever the Company proposes or is required to credit Deferred Stock Units to an Account in connection with the exercise of an Option, the Company shall have the right to require the Participant to remit to the Company an amount sufficient to satisfy any federal, state and local withholding tax requirements prior to the date on which Deferred Stock Units shall be deemed credited to the Account, or take any action whatever that it deems necessary to protect its interests with respect to tax liabilities. The Company's obligation to credit Deferred Stock Units to an Account on the exercise of an Option subject to an Initial or Subsequent Election shall be conditioned on the Participant's compliance, to the Company's satisfaction, with any withholding requirement. Except as otherwise provided in Section 10.2, the Company shall satisfy all applicable withholding tax requirements by withholding tax from other compensation payable by the Company to the Participant, or by the Participant's delivery of cash or other property acceptable to the Company having a value equal to the applicable withholding tax.

10.2. Share Withholding Election. With respect to any Option subject to an Initial Election, an Eligible Employee, Former Eligible Employee, Successor-in-Interest or Permitted Transferee may elect to have the number of Option Shares determined such that Shares subject to such Option are withheld by the Company to the extent necessary to satisfy any withholding tax liabilities incurred in connection with the exercise of such Option. The number of Shares subject to an Option to be withheld pursuant to such a Share Withholding Election shall have a Fair Market Value approximately equal to the sum of:

(a) The minimum amount of withholding taxes required to be withheld by the Company under applicable law, plus

(b) Either (i) the minimum amount of withholding taxes arising because of the recognition of income (and consequent non-deferral of income) with respect to such withheld Shares, or (ii) the amount of withholding taxes arising because of the recognition of income (and consequent non-deferral of income) with respect to such withheld Shares, calculated at the highest applicable marginal tax rates, as indicated on the Share Withholding Election.

Notwithstanding any other provision of the Plan or the terms of any Initial or Subsequent Election, the number of Deferred Stock Units credited to Participants' Accounts shall be adjusted appropriately to reflect the withholding of Shares pursuant to such Share Withholding Elections.

ARTICLE 11-CAPITAL ADJUSTMENTS

11.1. Capital Adjustments. In the event that the Common Stock or Special Common Stock is changed into, or exchanged for, a different number or kind of shares of stock or other securities of the Company, whether through merger, consolidation, reorganization, recapitalization, stock dividends, stock split-ups or other substitution of securities of the Company, the Committee shall make appropriate equitable anti-dilution adjustments to the number of Deferred Stock Units credited to Participants' Accounts. The Committee's adjustment shall be effective and binding for all purposes of the Plan.

ARTICLE 12—MISCELLANEOUS PROVISIONS

12.1. No Right to Continued Employment. Nothing contained herein shall be construed as conferring upon any Participant the right to remain in service as an Outside Director or in the employment of a Participating Company as an executive or in any other capacity.

12.2. Expenses of Plan. All expenses of the Plan shall be paid by the Participating Companies.

12.3. Gender and Number. Whenever any words are used herein in any specific gender, they shall be construed as though they were also used in any other applicable gender. The singular form, whenever used herein, shall mean or include the plural form, and vice versa, as the context may require.

12.4. Law Governing Construction. The construction and administration of the Plan and all questions pertaining thereto, shall be governed by the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), and other applicable federal law and, to the extent not governed by federal law, by the laws of the Commonwealth of Pennsylvania.

12.5. Headings Not a Part Hereof. Any headings preceding the text of the several Articles, Sections, subsections, or paragraphs hereof are inserted solely for convenience of reference and shall not constitute a part of the Plan, nor shall they affect its meaning, construction, or effect.

12.6. Severability of Provisions. If any provision of this Plan is determined to be void by any court of competent jurisdiction, the Plan shall continue to operate and, for the purposes of the jurisdiction of that court only, shall be deemed not to include the provision determined to be void.

12.7. Expiration of Options. Notwithstanding any provision of the Plan or an Initial or Subsequent Election, no Initial or Subsequent Election shall be effective with respect to an Option that has expired. In addition, no provision of the Plan or an Initial or Subsequent Election shall be construed to extend the expiration date of any Option.

ARTICLE 13—EFFECTIVE DATE

13.1. Effective Date. The effective date of this amendment and restatement of the Plan shall be February 16, 2005.

IN WITNESS WHEREOF, COMCAST CORPORATION has caused this Plan to be executed by its officers thereunto duly authorized, and its corporate seal to be affixed hereto, as of the 16th day of February, 2005.

-

BY:

- -

ATTEST:

QuickLinks

COMCAST CORPORATION 2002 DEFERRED STOCK OPTION PLAN ARTICLE 1—CONTINUATION AND COVERAGE OF PLAN ARTICLE 2—DEFINITIONS ARTICLE 3—INITIAL AND SUBSEQUENT ELECTIONS ARTICLE 4—MANNER OF DISTRIBUTION ARTICLE 5—BOOK ACCOUNTS ARTICLE 6—NONALIENATION OF BENEFITS ARTICLE 6—NONALIENATION OF BENEFITS ARTICLE 6—NONALIENATION OF BENEFITS ARTICLE 8—INTERPRETATION ARTICLE 8—INTERPRETATION ARTICLE 9—AMENDMENT OR TERMINATION ARTICLE 10—WITHHOLDING OF TAXES ON EXERCISE OF OPTION ARTICLE 11—CAPITAL ADJUSTMENTS ARTICLE 12—MISCELLANEOUS PROVISIONS

4

·· —

ARTICLE 13—EFFECTIVE DATE

COMCAST CORPORATION 2002 RESTRICTED STOCK PLAN (As Amended And Restated, Effective January 1, 2005)

1. BACKGROUND AND PURPOSE

(a) Amendment and Restatement of Plan. COMCAST CORPORATION, a Pennsylvania corporation, hereby amends and restates the Comcast Corporation 2002 Restricted Stock Plan (the "Plan"), effective January 1, 2005. The purpose of the Plan is to promote the ability of Comcast Corporation to recruit and retain employees and enhance the growth and profitability of Comcast Corporation by providing the incentive of long-term awards for continued employment and the attainment of performance objectives.

(b) Purpose of the Amendment; Credits Affected. The Plan has been amended and restated, effective January 1, 2005 in order (i) to preserve the favorable tax treatment available to amounts deferred pursuant to the Plan before January 1, 2005 and the earnings credited in respect of such amounts (each a "Grandfathered Amount") in light of the American Jobs Creation Act of 2004, *IRS Notice* 2005-1, and the regulations issued by the Department of the Treasury thereunder (collectively, the "AJCA"), and (ii) with respect to all other amounts eligible to be deferred under the Plan, to comply with the requirements of the AJCA. Except as provided in Paragraph 8(f)(iii) of the Plan, Grandfathered Amounts will continue to be subject to the terms and conditions of the Plan as in effect prior to the Amendment Date. All amounts eligible to be deferred under the Plan other than Grandfathered Amounts will be subject to the terms of this amendment and restatement of the Plan and the AJCA.

(c) Reservation of Right to Amend to Comply with AJCA. The Board and the Committee reserve the right to amend the Plan, either retroactively or prospectively, in whatever respect is required to achieve and maintain compliance with the requirements of the AJCA.

(d) Deferral Provisions of Plan Unfunded and Limited to Select Group of Management or Highly Compensated Employees. Deferral Eligible Grantees and Non-Employee Directors may elect to defer the receipt of Restricted Stock and Restricted Stock Units as provided in Article VIII. The deferral provisions of Article VIII and the other provisions of the Plan relating to the deferral of Restricted Stock and Restricted Stock Units are unfunded and maintained primarily for the purpose of providing a select group of management or highly compensated employees the opportunity to defer the receipt of compensation otherwise payable to such eligible employees in accordance with the terms of the Plan.

2. **DEFINITIONS**

(a) "Acceleration Election" means a written election on a form provided by the Committee, pursuant to which a Deceased Grantee's Successor-in-Interest or a Disabled Grantee elects to accelerate the distribution date of Shares issuable with respect to Restricted Stock and/or Restricted Stock Units.

(b) "Account" means unfunded bookkeeping accounts established pursuant to Paragraph 8(e) and maintained by the Committee in the names of the respective Grantees (i) to which Deferred Stock Units are deemed credited and (ii) to which an amount equal to the Fair Market Value of Deferred Stock Units with respect to which a Diversification Election has been made and interest thereon are deemed credited, reduced by distributions in accordance with the Plan.

(c) "Active Grantee" means each Grantee who is actively employed by a Participating Company.

(d) "Affiliate" means, with respect to any Person, any other person that, directly or indirectly, is in control of, is controlled by, or is under common control with, such Person. For purposes of this definition, the term "control," including its correlative terms "controlled by" and "under common control with," mean, with respect to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract or otherwise.

(e) "AJCA" means the American Jobs Creation Act of 2004, IRS Notice 2005-1 and announcements, notices, revenue rulings and regulations issued under the American Jobs Creation Act of 2004.

(f) "Annual Rate of Pay" means, as of any date, an employee's annualized base pay rate. An employee's Annual Rate of Pay shall not include sales commissions or other similar payments or awards.

(g) "Applicable Interest Rate" means:

(i)

Except as otherwise provided in Paragraph 2(g)(ii), the Applicable Interest Rate means the interest rate that, when compounded annually pursuant to rules established by the Committee from time to time, is mathematically equivalent to 8% per annum, compounded annually, or such other interest rate established by the Committee from time to time. The effective date of any reduction in the Applicable Interest Rate shall not precede the later of: (A) the 30th day following the date of the Committee's action to establish a reduced rate; or (B) the lapse of 24 full calendar months from the date of the most recent adjustment of the Applicable Interest Rate by the Committee.

(ii)

Effective for the period extending from a Grantee's employment termination date to the date the Grantee's Account is distributed in full, the Committee, in its sole and absolute discretion, may designate the term "Applicable Interest Rate" for such Grantee's Account to mean the lesser of: (A) the rate in effect under Paragraph 2(g)(i) or (B) the interest rate that, when compounded annually pursuant to rules established by the Committee from time to time, is mathematically equivalent to the Prime Rate plus one percent, compounded annually as of the last day of the calendar year. Notwithstanding the foregoing, the Committee may delegate its authority to determine the Applicable Interest Rate under this Paragraph 2(g)(ii) to an officer of the Company or committee of two or more officers of the Company.

(h) "AT&T Broadband Transaction" means the acquisition of AT&T Broadband Corp. (now known as Concast Cable Communications Holdings, Inc.) by the Company.

- (i) "Award" means an award of Restricted Stock or Restricted Stock Units granted under the Plan.
- (j) "Board" means the Board of Directors of the Company.
- (k) "Change of Control" means:
 - (i)

For all purposes of the Plan other than Article VIII, any transaction or series of transactions as a result of which any Person who was a Third Party immediately before such transaction or series of transactions owns then-outstanding securities of the Company such that such Person has the ability to direct the management of the Company, as determined by the Board in its discretion. The Board may also determine that a Change of Control shall occur upon the completion of one or more proposed transactions. The Board's determination shall be final and binding.

(ii)

For purposes of Article VIII, any transaction or series of transactions that constitutes:

(1)

a change in the ownership of the Company, within the meaning of Q&A 12 of IRS Notice 2005-1;

- a change in effective control of the Company, within the meaning of Q&A 13 of IRS Notice 2005-1; or
- (3)

a change in the ownership of a substantial portion of the assets of the Company, within the meaning of Q&A 14 of *IRS Notice* 2005-1.

(1) "Code" means the Internal Revenue Code of 1986, as amended.

(m) "Comcast Plan" means any restricted stock, restricted stock unit, stock bonus, stock option or other compensation plan, program or arrangement established or maintained by the Company or an Affiliate, including but not limited to this Plan, the Comcast Corporation 2003 Stock Option Plan, the Comcast Corporation 2002 Stock Option Plan, the Comcast Corporation 1996 Stock Option Plan, Comcast Corporation 1987 Stock Option Plan and the Comcast Corporation 2002 Deferred Stock Option Plan.

- (n) "Committee" means the Compensation Committee of the Board.
- (o) "Common Stock" means Class A Common Stock, par value \$0.01, of the Company.

(p) "Company" means Concast Corporation, a Pennsylvania corporation, as successor to Concast Holdings Corporation (formerly known as Corncast Corporation), including any successor thereto by merger, consolidation, acquisition of all or substantially all the assets thereof, or otherwise.

(q) "Company Stock Fund" means a hypothetical investment fund pursuant to which Deferred Stock Units are credited with respect to a portion of an Award subject to an Election, and thereafter until (i) the date of distribution or (ii) the effective date of a Diversification Election, to the extent a Diversification Election applies to such Deferred Stock Units, as applicable. The portion of a Grantee's Account deemed invested in the Company Stock Fund shall be treated as if such portion of the Account were invested in hypothetical shares of Common Stock or Special Common Stock otherwise deliverable as Shares upon the Vesting Date associated with Restricted Stock or Restricted Stock Units, and all dividends and other distributions paid with respect to Common Stock or Special Common Stock were held uninvested in cash and credited with interest at the Applicable Interest Rate as of the next succeeding December 31 (to the extent the Account continues to be deemed credited in the form of Deferred Stock Units through such December 31).

.

- (r) "Date of Grant" means the date on which an Award is granted.
- (s) "Deceased Grantee" means:
 - A Grantee whose employment by a Participating Company is terminated by death; or
 - (ii) A Grantee who dies following termination of employment by a Participating Company.
- (t) "Deferral Eligible Employee" means:

(i)

(i)

An Eligible Employee whose Annual Rate of Pay is \$200,000 or more as of both: (i) the date on which an Initial Election is filed with the Committee; and (ii) the first day of the calendar year in which such Initial Election filed.

(ii)

An Eligible Employee whose Annual Rate of Pay is \$125,000 as of each of: (A) June 30, 2002; (B) the date on which an Initial Election is filed with the Committee; and (C) the first day of each calendar year beginning after December 31, 2002.

(iii)

Each New Key Employee.

(iv)

- Each other employee of a Participating Company who is designated by the Committee, in its sole and absolute discretion, as a Deferral Eligible Employee.
- (u) "Deferred Stock Units" means the number of hypothetical Shares subject to an Election.
- (v) "Disability" means:

(i)

An individual's inability to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment which can be expected to result in death or can be expected to last for a continuous period of not less than 12 months; or

(ii)

Circumstances under which, by reason of any medically determinable physical or mental impairment which can be expected to result in death or can be expected to last for a continuous period of not less than 12 months, an individual is receiving income replacement benefits for a period of not less than three months under an accident or health plan covering employees of the individual's employer.

(w) "Disabled Grantee" means:

(i)

A Grantee whose employment by a Participating Company is terminated by reason of Disability;

(ii)

The duly-appointed legal guardian of an individual described in Paragraph 2(w)(i) acting on behalf of such individual.

(x) "Diversification Election" means a Grantee's election to have a portion of the Grantee's Account credited in the form of Deferred Stock Units and attributable to any grant of Restricted Stock or Restricted Stock Units deemed liquidated and credited thereafter under the Income Fund, as provided in Paragraph 8(g).

- (y) "Election" means, as applicable, an Initial Election, a Subsequent Election, or an Acceleration Election.
- (z) "Eligible Employee" means an employee of a Participating Company, as determined by the Committee.

(aa) "Fair Market Value" means:

(i)

If Shares are listed on a stock exchange, Fair Market Value shall be determined based on the last reported sale price of a Share on the principal exchange on which Shares are listed on the date of determination, or if such date is not a trading day, the next trading date.

(ii)

If Shares are not so listed, but trades of Shares are reported on the Nasdaq National Market, Fair Market Value shall be determined based on the last quoted sale price of a Share on the Nasdaq National Market on the date of determination, or if such date is not a trading day, the next trading date.

(iii)

If Shares are not so listed nor trades of Shares so reported, Fair Market Value shall be determined by the Committee in good faith.

- (bb) "Grandfathered Amount" means Deferred Stock Units described in Paragraph 1(b).
- (cc) "Grantee" means an Eligible Employee or Non-Employee Director who is granted an Award.

(dd) "Hardship" means a Grantee's severe financial hardship due to an unforeseeable emergency resulting from a sudden and unexpected illness or accident of the Grantee, or, a sudden and unexpected illness or accident of a dependent (as defined by section 152(a) of the Code) of the Grantee, or loss of the Grantee's property due to casualty, or other similar and extraordinary unforeseeable circumstances arising as a result of events beyond the control of the Grantee. A need to send the Grantee's child to college or a desire to purchase a home is not an unforeseeable emergency. No Hardship shall be deemed to exist to the extent that the financial hardship is or may be relieved (a) through reimbursement or compensation by insurance or otherwise, (b) by borrowing from commercial sources on reasonable commercial terms to the extent that this borrowing would not itself cause a severe financial hardship, (c) by cessation of deferrals under the Plan, or (d) by liquidation of the Grantee's other assets (including assets of the Grantee's spouse and minor children that are reasonably available to the Grantee) to the extent that this liquidation would not itself cause severe financial hardship. For the purposes of the preceding sentence, the Grantee's resources shall be deemed to include those assets of his spouse and minor children that are reasonably available to the Grantee; however, property held for the Grantee's child under an irrevocable trust or under a Uniform Gifts to Minors Act custodianship or Uniform Transfers to Minors Act custodianship shall not be treated as a resource of the Grantee. The Committee shall determine whether the circumstances of the Grantee constitute an unforeseeable emergency and thus a Hardship within the meaning of this Paragraph 2(dd). Following a uniform procedure, the Committee's determination shall consider any facts or conditions deemed necessary or advisable by the Committee, and the Grantee shall be required to submit any evidence of the Grantee's circumstances that the Committee requires. The determination as to whether the Grantee's circumstances are a case of Hardship shall be based on the facts of each case; provided however, that all determinations as to Hardship shall be uniformly and consistently made according to the provisions of this Paragraph 2(dd) for all Grantees in similar circumstances.

(ee) "Income Fund" means a hypothetical investment fund pursuant to which an amount equal to the Fair Market Value of Deferred Stock Units subject to a Diversification Election is credited as of the effective date of such Diversification Election and as to which interest is credited thereafter until the date of distribution at the Applicable Interest Rate.

(ff) "Initial Election" means a written election on a form provided by the Committee, pursuant to which a Grantee: (i) elects, within the time or times specified in Paragraph 8(a), to defer the distribution date of Shares issuable with respect to Restricted Stock or Restricted Stock Units; and (ii) designates the distribution date of such Shares.

(gg) "New Key Employee" means each employee of a Participating Company who: (i) becomes an employee of a Participating Company and has an Annual Rate of Pay of \$200,000 or more as of his employment commencement date; or (ii) has an Annual Rate of Pay that is increased to \$200,000 or more and who, immediately preceding such increase, was not a Deferral Eligible Employee.

4

(hh) "Non-Employee Director" means an individual who is a member of the Board, and who is not an employee of the Company, including an individual who is a member of the Board and who previously was an employee of the Company.

(ii) "Normal Retirement" means a Grantee's termination of employment that is treated by the Participating Company as a retirement under its employment policies and practices as in effect from time to time.

(jj) "Other Available Shares" means, as of any date, the sum of:

(i)

The total number of Shares owned by a Grantee that were not acquired by such Grantee pursuant to a Comcast Plan or otherwise in connection with the performance of services to the Company or an Affiliate; plus

(ii)

The excess, if any of:

- (1)
- The total number of Shares owned by a Grantee other than the Shares described in Paragraph 2(jj)(i); over
- (2)
- The sum of:

(A)

(B)

The number of such Shares owned by such Grantee for less than six months; plus

The number of such Shares owned by such Grantee that has, within the preceding six months, been the subject of a withholding certification pursuant to Paragraph 9(c)(ii) or any similar withholding certification under any other Comcast Plan; plus

(C) The n

The number of such Shares owned by such Grantee that has, within the preceding six months, been received in exchange for Shares surrendered as payment, in full or in part, or as to which ownership was attested to as payment, in full or in part, of the exercise price for an option to purchase any securities of the Company or an Affiliate of the Company, under any Comcast Plan, but only to the extent of the number of Shares surrendered or attested to; plus

(D)

The number of such Shares owned by such Grantee as to which evidence of ownership has, within the preceding six months, been provided to the Company in connection with the crediting of "Deferred Stock Units" to such Grantee's Account under the Comcast Corporation 2002 Deferred Stock Option Plan (as in effect from time to time).

For purposes of this Paragraph 2(jj), a Share that is subject to an Election pursuant to Paragraph 8 or a deferral election pursuant to another Comcast Plan shall not be treated as owned by a Grantee until all conditions to the delivery of such Share have lapsed. The number of Other Available Shares shall be determined separately for Common Stock and Special Common Stock. For purposes of determining the number of Other Available Shares, the term "Shares" shall also include the securities held by a Grantee immediately before the consummation of the AT&T Broadband Transaction that became Shares as a result of the AT&T Broadband Transaction.

(kk) "Participating Company" means the Company and each of the Subsidiary Companies.

(11) "Performance-Based Compensation" means "performance-based compensation" within the meaning of Q&A 22 of IRS Notice 2005-1, or such other guidance as may be issued by the Department of the Treasury under section 409A of the Code.

(mm) "Performance Period" means a period of at least 12 months during which a Grantee may earn Performance-Based Compensation.

- (nn) "Person" means an individual, a corporation, a partnership, an association, a trust or any other entity or organization.
- (oo) "Plan" means the Concast Corporation 2002 Restricted Stock Plan, as set forth herein, and as amended from time to time.

(pp) "Prime Rate" means, for any calendar year, the interest rate that, when compounded daily pursuant to rules established by the Committee from time to time, is mathematically equivalent to the prime rate of interest (compounded annually) as published in the Eastern Edition of The Wall Street Journal on the last business day preceding the first day of such calendar year, and as adjusted as of the last business day preceding the first day of each calendar year beginning thereafter.

- (qq) "Restricted Stock" means Shares subject to restrictions as set forth in an Award.
- (IT) "Restricted Stock Unit" means a unit that entitles the Grantee, upon the Vesting Date set forth in an Award, to receive one Share.
- (ss) "Retired Grantee" means a Grantee who has terminated employment pursuant to a Normal Retirement.
- (tt) "Rule 16b-3" means Rule 16b-3 promulgated under the 1934 Act, as in effect from time to time.
- (uu) "Share" or "Shares" means:
 - (i)

except as provided in Paragraph 2(uu)(ii), a share or shares of Common Stock.

- (ii)
- with respect to Awards granted before the consummation of the AT&T Broadband Transaction as to which a Vesting Date has not occurred, and for purposes of Paragraphs 2(jj) and 9(c), the term "Share" or "Shares" also means a share or shares of Special Common Stock.
- (vv) "Special Common Stock" means Class A Special Common Stock, par value \$0.01, of the Company.

(ww) "Special Diversification Election" means, with respect to each separate grant of Restricted Stock or Restricted Stock Units, a Diversification Election by a Grantee to have more than 40 percent of the Deferred Stock Units credited to such Grantee's Account in the Company Stock Fund liquidated and credited thereafter under the Income Fund, as provided in Paragraph 8(g)(i), if (and to the extent that) it is approved by the Committee in accordance with Paragraph 8(g)(ii).

(xx) "Subsequent Election" means a written election on a form provided by the Committee, filed with the Committee in accordance with Paragraph 8(d), pursuant to which a Grantee: (i) elects, within the time or times specified in Paragraph 8(d), to further defer the distribution date of Shares issuable with respect to Restricted Stock or Restricted Stock Units; and (ii) designates the distribution date of such Shares.

(yy) "Subsidiary Companies" means all business entities that, at the time in question, are subsidiaries of the Company, within the meaning of section 424(f) of the Code.

(zz) "Successor-in-Interest" means the estate or beneficiary to whom the right to payment under the Plan shall have passed by will or the laws of descent and distribution.

- (aaa) "Terminating Event" means any of the following events:
 - (i)

the liquidation of the Company; or

(ii)

a Change of Control.

(bbb) "Third Party" means any Person, together with such Person's Affiliates, provided that the term "Third Party" shall not include the Company or an Affiliate of the Company.

(ccc) "Vesting Date" means, as applicable: (i) the date on which the restrictions imposed on a Share of Restricted Stock lapse or (ii) the date on which the Grantee vests in a Restricted Stock Unit.

(ddd) "1933 Act" means the Securities Act of 1933, as amended.

(eee) "1934 Act" means the Securities Exchange Act of 1934, as amended.

3. RIGHTS TO BE GRANTED

Rights that may be granted under the Plan are:

(a) Rights to Restricted Stock which gives the Grantee ownership rights in the Shares subject to the Award, subject to a substantial risk of forfeiture, as set forth in Paragraph 7, and to deferred payment, as set forth in Paragraph 8; and

(b) Rights to Restricted Stock Units which give the Grantee the right to receive Shares upon a Vesting Date, as set forth in Paragraph 7, and to deferred payment, as set forth in Paragraph 8. The maximum number of Shares subject to Awards that may be granted to any single individual in any calendar year, adjusted as provided in Paragraph 10, shall be one million Shares.

4. SHARES SUBJECT TO THE PLAN

(a) Not more than 15 million Shares in the aggregate may be issued under the Plan pursuant to the grant of Awards, subject to adjustment in accordance with Paragraph 10. The Shares issued under the Plan may, at the Company's option, be either Shares held in treasury or Shares originally issued for such purpose.

(b) If Restricted Stock or Restricted Stock Units are forfeited pursuant to the term of an Award, other Awards with respect to such Shares may be granted.

5.

ADMINISTRATION OF THE PLAN

(a) Administration. The Plan shall be administered by the Committee, provided that with respect to Awards to Non-Employee Directors, the rules of this Section 5 shall apply so that all references in this Section 5 to the Committee shall be treated as references to either the Board or the Committee acting alone.

- (b) Grants. Subject to the express terms and conditions set forth in the Plan, the Committee shall have the power, from time to time, to:
 - (i)

select those Employees and Non-Employee Directors to whom Awards shall be granted under the Plan, to determine the number of Shares and/or Restricted Stock Units, as applicable, to be granted pursuant to each Award, and, pursuant to the provisions of the Plan, to determine the terms and conditions of each Award, including the restrictions applicable to such Shares and the conditions upon which a Vesting Date shall occur; and

(ii)

interpret the Plan's provisions, prescribe, amend and rescind rules and regulations for the Plan, and make all other determinations necessary or advisable for the administration of the Plan.

The determination of the Committee in all matters as stated above shall be conclusive.

(c) Meetings. The Committee shall hold meetings at such times and places as it may determine. Acts approved at a meeting by a majority of the members of the Committee or acts approved in writing by the unanimous consent of the members of the Committee shall be the valid acts of the Committee.

(d) *Exculpation.* No member of the Committee shall be personally liable for monetary damages for any action taken or any failure to take any action in connection with the administration of the Plan or the granting of Awards thereunder unless (i) the member of the Committee has breached or failed to perform the duties of his office, and (ii) the breach or failure to perform constitutes self-dealing, willful misconduct or recklessness; provided, however, that the provisions of this Paragraph 5(d) shall not apply to the responsibility or liability of a member of the Committee pursuant to any criminal statute.

(e) Indemnification. Service on the Committee shall constitute service as a member of the Board. Each member of the Committee shall be entitled without further act on his part to indemnity from the Company to the fullest extent provided by applicable law and the Company's Articles of Incorporation and By-laws in connection with or arising out of any action, suit or proceeding with respect to the administration

of the Plan or the granting of Awards thereunder in which he may be involved by reason of his being or having been a member of the Committee, whether or not he continues to be such member of the Committee at the time of the action, suit or proceeding.

(f) Delegation of Authority. The Committee may delegate to an officer of the Company, or a committee of two or more officers of the Company, discretion under the Plan to grant Restricted Stock and/or Restricted Stock Units to any Grantee other than a Grantee who, at the time of the grant:

	(i)	has a base salary of \$500,000 or more;
	(ii)	holds a position with Comcast Corporation of Senior Vice President or a position of higher rank than Senior Vice President; or
	(iii)	is subject to the short-swing profit recapture rules of section 16(b) of the 1934 Act.
(g)	3) Termination of Delegation of Authority. Any delegation of authority described in Paragraph 5(f) shall continue in effect until the earliest	
	(i)	such time as the Committee shall, in its discretion, revoke such delegation of authority;
	(ii)	the delegate shall cease to be an employee of the Company for any reason; or

(iii) the delegate shall notify the Committee that he declines to continue exercise such authority.

6. ELIGIBILITY

Awards may be granted only to Eligible Employees and, subject to the approval of the shareholders of the Company at the Annual Meeting of Shareholders of the Company to be held in 2005, Non-Employee Directors.

7. RESTRICTED STOCK AND RESTRICTED STOCK UNIT AWARDS

The Committee may grant Awards in accordance with the Plan, provided that the Board or the Committee may grant Awards to Non-Employee Directors authorized by the Comcast Corporation 2002 Non-Employee Director Compensation Plan, or otherwise. With respect to Awards to Non-Employee Directors, the rules of this Section 7 shall apply so that either the Board or the Committee acting alone shall have all of the authority otherwise reserved in this Section 7 to the Committee.

The terms and conditions of Awards shall be set forth in writing as determined from time to time by the Committee, consistent, however, with the following:

(a) Time of Grant. All Awards shall be granted within ten (10) years from the date of adoption of the Plan by the Board.

(b) Terms of Awards. The provisions of Awards need not be the same with respect to each Grantee. No cash or other consideration shall be required to be paid by the Grantee in exchange for an Award.

(c) Awards and Agreements. Each Grantee shall be provided with an agreement specifying the terms of an Award. In addition, a certificate shall be issued to each Grantee in respect of Restricted Shares subject to an Award. Such certificate shall be registered in the name of the Grantee and shall bear an appropriate legend referring to the terms, conditions and restrictions applicable to such Award. The Company may require that the certificate evidencing such Restricted Stock be held by the Company until all restrictions on such Restricted Stock have lapsed.

(d) Restrictions. Subject to the provisions of the Plan and the Award, the Committee may establish a period commencing with the Date of Grant during which the Grantee shall not be permitted to sell, transfer, pledge or assign Restricted Stock awarded under the Plan.

(e) Vesting/Lapse of Restrictions. Subject to the provisions of the Plan and the Award, a Vesting Date for Restricted Stock or Restricted Stock Units subject to an Award shall occur at such time or times and on such terms and conditions as the Committee may determine and as are set forth in the Award; provided,

however, that except as otherwise provided by the Committee, a Vesting Date shall occur only if the Grantee is an employee of a Participating Company as of such Vesting Date, and has been an employee of a Participating Company continuously from the Date of Grant. The Award may provide for Restricted Stock or Restricted Stock Units to vest in installments, as determined by the Committee. The Committee may, in its sole discretion, waive, in whole or in part, any remaining conditions to vesting with respect to such Grantee's Restricted Stock or Restricted Stock Units. All references to Shares in Awards granted before the consummation of the AT&T Broadband Transaction as to which a Vesting Date has not occurred shall be deemed to be references to Special Common Stock.

(f) Rights of the Grantee. Grantees may have such rights with respect to Shares subject to an Award as may be determined by the Committee and set forth in the Award, including the right to vote such Shares, and the right to receive dividends paid with respect to such Shares. A Grantee whose Award consists of Restricted Stock Units shall not have the right to vote or to receive dividend equivalents with respect to such Restricted Stock Units.

(g) Termination of Grantee's Employment. A transfer of an Eligible Employee between two employers, each of which is a Participating Company, shall not be deemed a termination of employment. In the event that a Grantee terminates employment with all Participating Companies, all Restricted Shares and/or Restricted Stock Units as to which a Vesting Date has not occurred shall be forfeited by the Grantee and deemed canceled by the Company.

(h) Delivery of Shares. Except as otherwise provided by Paragraph 8, when a Vesting Date occurs with respect to all or a portion of an Award of Restricted Stock or Restricted Stock Units, the Company shall notify the Grantee that a Vesting Date has occurred, and shall deliver to the Grantee (or the Grantee's Successor-in-Interest) a certificate for the number of Shares as to which a Vesting Date has occurred (or in the case of Restricted Stock Units, the number of Shares represented by such Restricted Stock Units) without any legend or restrictions (except those that may be imposed by the Committee, in its sole judgment, under Paragraph 9(a)). The right to payment of any fractional Shares that may have accrued shall be satisfied in cash, measured by the product of the fractional amount times the Fair Market Value of a Share at the Vesting Date, as determined by the Committee.

8. DEFERRAL ELECTIONS

A Grantee may elect to defer the receipt of Shares that would otherwise be issuable with respect to Restricted Stock or Restricted Stock Units as to which a Vesting Date has occurred, as provided by the Committee in the Award, consistent, however, with the following:

(a) Initial Election.

(i) *Election.* Each Grantee who is a Non-Employee Director or a Deferral Eligible Employee shall have the right to defer the receipt of some or all of the Shares issuable with respect to Restricted Stock or Restricted Stock Units as to which a Vesting Date has not yet occurred, by filing an Initial Election to defer the receipt of such Shares on a form provided by the Committee for this purpose.

(ii) Deadline for Initial Election. No Initial Election to defer the receipt of Shares issuable with respect to Restricted Stock or Restricted Stock Units that are not Performance-Based Compensation shall be effective unless it is filed with the Committee on or before the 30 th day following the Date of Grant provided that pursuant to Q-A 21 of *IRS Notice 2005-1*, to the extent provided by the Committee or its delegate, a Grantee may, on or before March 15, 2005, make an Initial Election with respect to Restricted Stock or Restricted Stock Units that were granted before January 1, 2005 and were not vested on December 31, 2004, and with respect to Restricted Stock or Restricted Stock Units that may be granted after December 31, 2004, provided further that the Restricted Stock or Restricted Stock Units to which the Initial Election relates have not been vested at the time the Initial Election is filed. No Initial Election to defer the receipt of Shares issuable with respect to Restricted Stock or Restricted Stock Units that are Performance-Based Compensation shall be effective unless it is filed with the

Administrator at least six months before the end of the Performance Period during which such Performance-Based Compensation may be earned.

(iii) Special Transition Rule. Pursuant to Q-A 20 of IRS Notice 2005-1, to the extent provided by the Committee or its delegate, a Grantee may, on or before December 31, 2005, terminate the deferral of Restricted Stock or Restricted Stock Units pursuant to an Initial Election or cancel an Initial Election with regard to amounts deferred under the Plan, provided that if a Grantee terminates the deferral of Compensation pursuant to an Initial Election under this Paragraph 8(a)(iii), the Company shall pay the Grantee the Compensation that would have been deferred if the deferral of Compensation had not been terminated, and provided further that if a Grantee cancels an Initial Election with regard to amounts deferred under the Plan, the Company shall pay the Grantee the amount deferred pursuant to such Initial Election through the cancellation date, plus income, gains and losses credited with respect thereto as provided in this Article VIII.

(b) Effect of Failure of Vesting Date to Occur. An Election shall be null and void if a Vesting Date with respect to the Restricted Stock or Restricted Stock Units does not occur before the distribution date for Shares issuable with respect to such Restricted Stock or Restricted Stock Units identified in such Election.

(c) Deferral Period. Except as otherwise provided in Paragraph 8(d), all Shares issuable with respect to Restricted Stock or Restricted Stock Units that are subject to an Election shall be delivered to the Grantee (or the Grantee's Successor-in-Interest) without any legend or restrictions (except those that may be imposed by the Committee, in its sole judgment, under Paragraph 9(a)), on the distribution date for such Shares designated by the Grantee on the most recently filed Election. Subject to acceleration or deferral pursuant to Paragraph 8(d) or Paragraph 11, no distribution may be made earlier than January 2nd of the third calendar year beginning after the Vesting Date, nor later than January 2nd of the eleventh calendar year beginning after the Vesting Date. The distribution date may vary with each separate Election.

(d) Additional Elections. Notwithstanding anything in this Paragraph 8(d) to the contrary, no Subsequent Election shall be effective until 12 months after the date on which such Subsequent Election is made.

(i)

Each Active Grantee who has previously made an Initial Election to receive a distribution of part or all of his or her Account, or who, pursuant to this Paragraph 8(d)(i) has made a Subsequent Election to defer the distribution date for Shares issuable with respect to Restricted Stock or Restricted Stock Units for an additional period from the originally-elected distribution date, may elect to defer the distribution date for a minimum of five and a maximum of ten additional years from the previously-elected distribution date, by filing a Subsequent Election with the Committee on or before the close of business at least one year before the date on which the distribution would otherwise be made.

(ii)

A Deceased Grantee's Successor-in-Interest may elect to: (A) file a Subsequent Election to defer the distribution date for the Deceased Grantee's Shares issuable with respect to Restricted Stock or Restricted Stock Units for a minimum of five additional years from the date payment would otherwise be made; or (B) file an Acceleration Election to accelerate the distribution date for the Deceased Grantee's Shares issuable with respect to Restricted Stock or Restricted Stock Units from the date payment would otherwise be made; or (B) file an Acceleration Election to accelerate the distribution date for the Deceased Grantee's Shares issuable with respect to Restricted Stock or Restricted Stock Units from the date payment would otherwise be made to a date that is as soon as practicable following the Deceased Grantee's death. A Subsequent Election must be filed with the Committee at least one year before the date on which the distribution would otherwise be made, as reflected on the Deceased Grantee's last Election. An Acceleration Election pursuant to this Paragraph 8(d)(ii) must be filed with the Committee as soon as practicable following the Deceased Grantee's death, as determined by the Committee.

(iii)

A Disabled Grantee may elect to accelerate the distribution date of the Disabled Grantee's Shares issuable with respect to Restricted Stock or Restricted Stock Units from the date payment would otherwise be made to a date that is as soon as practicable following the date the Disabled Grantee became disabled. An Acceleration Election pursuant to this Paragraph 8(d)(iii) must be filed with

the Committee as soon as practicable following the Deceased Grantee's death, as determined by the Committee.

(iv)

A Retired Grantee may elect to defer the distribution date of the Retired Grantee's Shares issuable with respect to Restricted Stock or Restricted Stock Units for a minimum of five additional years from the date payment would otherwise be made (provided that if a Subsequent Election is made pursuant to this Paragraph 8(d)(iv), the Retired Grantee's Account shall be distributed in full on or before the later of the fifth anniversary of the Retired Grantee's Normal Retirement or the fifth anniversary of the date that payment would otherwise have been made. A Subsequent Election must be filed with the Committee at least one year before the date on which the distribution would otherwise be made, as reflected on the Retired Grantee's last Election.

(v)

Discretion to Provide for Distribution in Full Upon or Following a Change of Control. To the extent permitted by *IRS Notice 2005-1*, in connection with a Change of Control, and for the 12-month period following a Change of Control, the Committee may exercise its discretion to terminate the deferral provisions of the Plan and, notwithstanding any other provision of the Plan or the terms of any Initial Election or Subsequent Election, distribute the Account of each Grantee in full and thereby effect the revocation of any outstanding Initial Elections or Subsequent Elections.

(vi)

Hardship. Notwithstanding the terms of an Initial Election or Subsequent Election, if, at the Grantee's request, the Committee determines that the Grantee has incurred a Hardship, the Committee may, in its discretion, authorize the immediate distribution of all or any portion of the Grantee's Account.

(vii)

Other Acceleration Events. To the extent permitted by Q-A 15 of *IRS Notice 2005-1*, notwithstanding the terms of an Initial Election or Subsequent Election, distribution of all or part of a Grantee's Account may be made:

(1) To the extent necessary to fulfill a domestic relations order (as defined in section 414(p)(1)(B) of the Code).

(2) To the extent necessary to comply with a certificate of divestiture (as defined in section 1043(b)(2) of the Code).

(3)

To pay the Federal Insurance Contribution Act ("FICA") tax imposed under sections 3101 and 3121(v)(2) of the Code on compensation deferred under the Plan (the "FICA Amount") plus the income tax at source on wages imposed under section 3401 of the Code with respect to the FICA Amount, and to pay the additional income tax at source on wages attributable to the pyramiding section 3401 wages and taxes, provided that the total amount distributable under this Paragraph 8(d)(vii)(3) shall not exceed the sum of the FICA Amount and the income tax withholding related to such FICA Amount.

(e) Book Accounts. An Account shall be established for each Grantee who makes an Election. Deferred Stock Units shall be credited to the Account as of the date an Election becomes effective. Each Deferred Stock Unit will represent, as applicable, either a hypothetical share of Common Stock or a hypothetical share of Special Common Stock credited to the Account in lieu of delivery of the Shares to which the Election applies. To the extent an Account is deemed invested in the Income Fund, the Committee shall credit earnings with respect to such Account at the Applicable Interest Rate, as further provided in Paragraph 8(g).

(f) Plan-to-Plan Transfers. The Administrator may delegate its authority to arrange for plan-to-plan transfers as described in this Paragraph 8(f) to an officer of the Company or committee of two or more officers of the Company.

(i)

The Administrator may, with a Grantee's consent, make such arrangements as it may deem appropriate to transfer the Company's obligation to pay benefits with respect to such Grantee which have not become payable under this Plan, to another employer, whether through a deferred

compensation plan, program or arrangement sponsored by such other employer or otherwise, or to another deferred compensation plan, program or arrangement sponsored by the Company or an Affiliate. Following the completion of such transfer, with respect to the benefit transferred, the Grantee shall have no further right to payment under this Plan.

(ii)

The Administrator may, with a Grantee's consent, make such arrangements as it may deem appropriate to assume another employer's obligation to pay benefits with respect to such Grantee which have not become payable under the deferred compensation plan, program or arrangement under which such future right to payment arose, to the Plan, or to assume a future payment obligation of the Company or an Affiliate under another plan, program or arrangement sponsored by the Company or an Affiliate. Upon the completion of the Plan's assumption of such payment obligation, the Administrator shall establish an Account for such Grantee, and the Account shall be subject to the rules of this Plan, as in effect from time to time.

(iii)

Pursuant to Q-A 19(c) of *IRS Notice* 2005-1, to the extent provided by the Committee or its delegate, a Grantee may, on or before December 31, 2005, with respect to all or any portion of his or her Grandfathered Amount under the Plan as in effect on December 31, 2004, make new payment elections as to the form and timing of payment of such amounts as may be permitted under this Plan, provided that following the completion of such new payment election, such amounts shall not be treated as a Grandfathered Amount, but instead shall be treated as a non-Grandfathered Amount, subject to the rules of this Plan.

(g) Crediting of Income, Gains and Losses on Accounts. Except as otherwise provided in Paragraph 8(h), the value of a Grantee's Account as of any date shall be determined as if it were invested in the Company Stock Fund.

(h) Diversification Elections.

(i) In General. A Diversification Election shall be available: (A) at any time that a Registration Statement filed under the 1933 Act (a "Registration Statement") is effective with respect to the Plan; and (B) with respect to a Special Diversification Election, if and to the extent that the opportunity to make such a Special Diversification Election has been approved by the Committee. No approval is required for a Diversification Election.

(ii) Committee Approval of Special Diversification Elections. The opportunity to make a Special Diversification Election and the extent to which a Special Diversification Election applies to Deferred Stock Units credited to the Company Stock Fund may be approved or rejected by the Committee in its sole discretion. A Special Diversification Election shall only be effective if (and to the extent) approved by the Committee.

(iii) Timing and Manner of Making Diversification Elections. Each Grantee and, in the case of a Deceased Grantee, the Successor-in-Interest, may make a Diversification Election to convert up to 40 percent (or in the case of a Special Diversification Election, up to the approved percentage) of Deferred Stock Units attributable to each grant of Restricted Stock or Restricted Stock Units credited to the Company Stock Fund to the Income Fund. No deemed transfers shall be permitted from the Income Fund to the Company Stock Fund. Diversification Elections under this Paragraph 8(h)(iii) shall be prospectively effective on the later of: (A) the date designated by the Grantee on a Diversification Election filed with the Committee; or (B) the business day next following the lapse of six months from the date Deferred Stock Units subject to the Diversification Election are credited to the Grantee's Account. In no event may a Diversification Election be effective earlier than the business day next following the lapse of six (6) months from the date Deferred Stock Units are credited to the Account following the lapse of restrictions with respect to an Award.

(iv) *Timing of Credits.* Account balances subject to a Diversification Election under this Paragraph 8(h) shall be deemed transferred from the Company Stock Fund to the Income Fund immediately following the effective date of such Diversification Election. The value of amounts deemed invested in the Income Fund immediately following the effective date of a Diversification Election shall
be based on hypothetical sales of Common Stock or Special Common Stock, as applicable, underlying the liquidated Deferred Stock Units at Fair Market Value as of the effective date of a Diversification Election.

(i) Effect of Distributions within Five Years of Effective Date of Diversification Election. If, pursuant to Paragraphs 8(a) through 8(d), Shares distributable with respect to Deferred Stock Units credited to the Company Stock Fund that are attributable to an Award as to which a Diversification Election was made are distributed on or before the fifth anniversary of the effective date of such Diversification Election (and, in the case of a Grantee who is a Successor-in-Interest, whether or not such Diversification Election was made by a Grantee's predecessor-in-interest), then, except as may otherwise be provided by the Committee in its sole and absolute discretion, the following percentage of the Grantee's Account credited to the Income Fund and attributable to such Diversification Election to the contrary:

Time that Shares are Distributable Distributable Corresponding Income Fund	
On or before the third anniversary of a Diversification Election After the third anniversary of a Diversification Election and on or before the	60%
fourth anniversary of a Diversification Election After the fourth anniversary of a Diversification Election and on or before the	40%
fifth anniversary of a Diversification Election After the fifth anniversary of a Diversification Election	20% 0%

(j) Grantees' Status as General Creditors. A Grantee's right to delivery of Shares subject to an Election under this Paragraph 8, or to amounts deemed invested in the Income Fund pursuant to a Diversification Election, shall at all times represent the general obligation of the Company. The Grantee shall be a general creditor of the Company with respect to this obligation, and shall not have a secured or preferred position with respect to such obligation. Nothing contained in the Plan or an Award shall be deemed to create an escrow, trust, custodial account or fiduciary relationship of any kind. Nothing contained in the Plan or an Award shall be construed to eliminate any priority or preferred position of a Grantee in a bankruptcy matter with respect to claims for wages.

(k) Non-Assignability, Etc. The right of a Grantee to receive Shares subject to an Election under this Paragraph 8, or to amounts deemed invested in the Income Fund pursuant to a Diversification Election, shall not be subject in any manner to attachment or other legal process for the debts of such Grantee; and no right to receive Shares or cash payments hereunder shall be subject to anticipation, alienation, sale, transfer, assignment or encumbrance.

9. SECURITIES LAWS; TAXES

(a) Securities Laws. The Committee shall have the power to make each grant of Awards under the Plan subject to such conditions as it deems necessary or appropriate to comply with the then-existing requirements of the 1933 Act and the 1934 Act, including Rule 16b-3. Such conditions may include the delivery by the Grantee of an investment representation to the Company in connection with a Vesting Date occurring with respect to Shares subject to an Award, or the execution of an agreement by the Grantee to refrain from selling or otherwise disposing of the Shares acquired for a specified period of time or on specified terms.

(b) Taxes. Subject to the rules of Paragraph 9(c), the Company shall be entitled, if necessary or desirable, to withhold the amount of any tax, charge or assessment attributable to the grant of any Award or the occurrence of a Vesting Date with respect to any Award. The Company shall not be required to deliver Shares pursuant to any Award until it has been indemnified to its satisfaction for any such tax, charge or assessment.

(c) Payment of Tax Liabilities; Election to Withhold Shares or Pay Cash to Satisfy Tax Liability.

(i)

In connection with the grant of any Award or the occurrence of a Vesting Date under any Award, the Company shall have the right to (A) require the Grantee to remit to the Company an amount sufficient to satisfy any federal, state and/or local withholding tax requirements prior to the delivery or transfer of any certificate or certificates for Shares subject to such Award, or (B) take any action whatever that it deems necessary to protect its interests with respect to tax liabilities. The Company's obligation to make any delivery or transfer of Shares shall be conditioned on the Grantee's compliance, to the Company's satisfaction, with any withholding requirement.

(ii)

Except as otherwise provided in this Paragraph 9(c)(ii), any tax liabilities incurred in connection with grant of any Award or the occurrence of a Vesting Date under any Award under the Plan shall be satisfied by the Company's withholding a portion of the Shares subject to such Award having a Fair Market Value approximately equal to the minimum amount of taxes required to be withheld by the Company under applicable law, unless otherwise determined by the Committee with respect to any Grantee. Notwithstanding the foregoing, the Committee may permit a Grantee to elect one or both of the following: (A) to have taxes withheld in excess of the minimum amount required to be withheld by the Company under applicable law; provided that the Grantee certifies in writing to the Company at the time of such election that the Grantee owns Other Available Shares having a Fair Market Value that is at least equal to the Fair Market Value to be withheld by the Company in payment of withholding taxes in excess of such minimum amount; and (B) to pay to the Company in cash all or a portion of the taxes to be withheld in connection with such grant or Vesting Date. In all cases, the Shares so withheld by the Grantee. Any election pursuant to this Paragraph 9(c)(ii) must be in writing made prior to the date specified by the Committee, and in any event prior to the date the amount of tax to be withheld or paid is determined. An election pursuant to this Paragraph 9(c)(ii) shall be available for subsequent grants under the Plan. The Committee may add such other requirements and limitations regarding elections pursuant to this Paragraph 9(c)(ii) as it deems approve the Plan.

10. CHANGES IN CAPITALIZATION

The aggregate number of Shares and class of Shares as to which Awards may be granted and the number of Shares covered by each outstanding Award shall be appropriately adjusted in the event of a stock dividend, stock split, recapitalization or other change in the number or class of issued and outstanding equity securities of the Company resulting from a subdivision or consolidation of the Shares and/or other outstanding equity security or a recapitalization or other capital adjustment (not including the issuance of

Shares and/or other outstanding equity securities on the conversion of other securities of the Company which are convertible into Shares and/or other outstanding equity securities) affecting the Shares which is effected without receipt of consideration by the Company. The Committee shall have authority to determine the adjustments to be made under this Paragraph 10 and any such determination by the Committee shall be final, binding and conclusive.

11. TERMINATING EVENTS

The Committee shall give Grantees at least thirty (30) days' notice (or, if not practicable, such shorter notice as may be reasonably practicable) prior to the anticipated date of the consummation of a Terminating Event. The Committee may, in its discretion, provide in such notice that upon the consummation of such Terminating Event, any conditions to the occurrence of a Vesting Date with respect to an Award of Restricted Stock or Restricted Stock Units (other than Restricted Stock or Restricted Stock Units that have previously been forfeited) shall be eliminated, in full or in part. Further, the Committee may, in its discretion, provide in such notice that notwithstanding any other provision of the Plan or the terms of any Election made pursuant to Paragraph 8, upon the consummation of a Terminating Event, Shares issuable with respect to Restricted Stock or Restricted Stock Units subject to an Election made pursuant to Paragraph 8 shall be transferred to the Grantee, and all amounts credited to the Income Fund shall be paid to the Grantee.

12. CLAIMS PROCEDURE

If an individual (hereinafter referred to as the "Applicant," which reference shall include the legal representative, if any, of the individual) does not receive timely payment of benefits to which the Applicant believes he is entitled under Paragraph 8 of the Plan, the Applicant may make a claim for benefits in the manner hereinafter provided.

An Applicant may file a claim for benefits with the Committee on a form supplied by the Committee. If the Committee wholly or partially denies a claim, the Committee shall provide the Applicant with a written notice stating:

(a)

(b)

The specific reason or reasons for the denial;

- Specific reference to pertinent Plan provisions on which the denial is based;
- (c)

(d)

A description of any additional material or information necessary for Applicant to perfect the claim and an explanation of why such material or information is necessary; and

Appropriate information as to the steps to be taken in order to submit a claim for review.

Written notice of a denial of a claim shall be provided within 90 days of the receipt of the claim, provided that if special circumstances require an extension of time for processing the claim, the Committee may notify the Applicant in writing that an additional period of up to 90 days will be required to process the claim.

If the Applicant's claim is denied, the Applicant shall have 60 days from the date of receipt of written notice of the denial of the claim to request a review of the denial of the claim by the Committee. Request for review of the denial of a claim must be submitted in writing. The Applicant shall have the right to review pertinent documents and submit issues and comments to the Committee in writing. The Committee shall provide a written decision within 60 days of its receipt of the Applicant's request for review, provided that if special circumstances require an extension of time for processing the review of the Applicant's claim, the Committee may notify the Applicant in writing that an additional period of up to 60 days shall be required to process the Applicant's request for review.

It is intended that the claims procedures of this Plan be administered in accordance with the claims procedure regulations of the Department of Labor set forth in 29 CFR § 2560.503-1.



Claims for benefits under the Plan must be filed with the Committee at the following address:

Comcast Corporation 1500 Market Street Philadelphia, PA 19102 Attention: General Counsel

13. AMENDMENT AND TERMINATION

The Plan may be terminated by the Board at any time. The Plan may be amended by the Board or the Committee at any time. No Award shall be affected by any such termination or amendment without the written consent of the Grantee.

16

14. EFFECTIVE DATE

The effective date of this amendment and restatement of the Plan is January 1, 2005.

15. GOVERNING LAW

The Plan and all determinations made and actions taken pursuant to the Plan shall be governed in accordance with Pennsylvania law.

Executed on the 16th day of February, 2005.

COMCAST CORPO BY:	RATION	 	 -
ATTEST:	••••••••••••••••••••••••••••••••••••••	 	

į

QuickLinks

COMCAST CORPORATION 2002 RESTRICTED STOCK PLAN (As Amended And Restated, Effective January 1, 2005)

-

..

,

COMCAST CORPORATION 2002 NON-EMPLOYEE DIRECTOR COMPENSATION PLAN (As Amended and Restated, Effective January 12, 2005)

1. BACKGROUND AND PURPOSE

COMCAST CORPORATION, a Pennsylvania corporation, hereby amends and restates the Comcast Corporation 2002 Non-Employee Director Compensation Plan, effective January 12, 2005. The purpose of the Plan is to provide Non-Employee Directors of COMCAST CORPORATION (the "Company") with compensation for services to the Company.

2. **DEFINITIONS**

(a) "Annual Retainer" means the amount payable for service as a Non-Employee Director for a calendar year, as a member of the Board, and as a member of one or more Committees as determined under Paragraph 3(a) of the Plan.

(b) "Board" means the Board of Directors of the Company.

- (c) "Board Meeting" means a meeting of the Board, whether in person or by telephone.
- (d) "Committee" means a duly-constituted committee of the Board.

(e) "Committee Meeting" means a meeting of a Committee, whether in person or by telephone, other than a meeting of a Committee that is convened and held during a Board Meeting.

(f) "Company" means Concast Corporation, a Pennsylvania corporation, including any successor thereto by merger, consolidation, acquisition of all or substantially all the assets thereof, or otherwise.

(g) "Fair Market Value" means:

(i) If Shares are listed on a stock exchange, Fair Market Value shall be determined based on the last reported sale price of a Share on the principal exchange on which Shares are listed on the date of determination, or if such date is not a trading day, the next trading date.

(ii) If Shares are not so listed, but trades of Shares are reported on the Nasdaq National Market, Fair Market Value shall be determined based on the last quoted sale price of a Share on the Nasdaq National Market on the date of determination, or if such date is not a trading day, the next trading date.

(iii) If Shares are not so listed nor trades of Shares so reported, Fair Market Value shall be determined by the Committee in good faith.

(h) "Non-Employee Director" means an individual who is a member of the Board, and who is not an employee of the Company, including an individual who is a member of the Board and who previously was an employee of the Company.

(i) "Plan" means the Comcast Corporation 2003 Non-Employee Director Compensation Plan, as set forth herein, and as amended from time to time.

(j) "Plan Year" means (i) the period from November 18, 2002 through December 31, 2002 and (ii) each calendar year beginning after 2002.

(k) "Restricted Stock Plan" means the Comcast Corporation 2002 Restricted Stock Plan.

(1) "Restricted Stock Unit" means a Restricted Stock Unit granted under the Restricted Stock Plan.

(m) "Share" means a share of Comcast Corporation Class A Common Stock, par value \$0.01.

3. NON-EMPLOYEE DIRECTOR COMPENSATION

(a) Non-Employee Director Compensation Package. Effective as of January 1, 2005, Non-Employee Directors shall be entitled to payments, grants and awards determined as follows:

(i) Annual Retainer. The Annual Retainer for service to the Company as a Non-Employee Director shall be \$50,000.

(ii) Board Meeting Fee. The fee payable for attendance in person or via telephone at a Board Meeting shall be \$2,500. The Board Meeting Fee will also be paid when a member of the Board is asked to attend a meeting or otherwise to conduct business on behalf of the Company in his or her capacity as a Director.

(iii) Annual Retainer: Chair-Audit Committee. The Annual Retainer for service as Chair of the Audit Committee shall be \$20,000

(iv) Annual Retainer: Member-Audit Committee. The Annual Retainer for service as a member of the Audit Committee shall be \$10,000.

(v) Annual Retainer: Chair—Compensation Committee and Governance and Directors Nominating Committee. The Annual Retainer for service as Chair of the Compensation Committee and the Governance and Directors Nominating Committee shall be \$10,000.

(vi) Annual Retainer: Member-Compensation Committee and Governance and Directors Nominating Committee. The Annual Retainer for service as a member of the Compensation Committee and the Governance and Directors Nominating Committee shall be \$5,000.

(vii) Annual Retainer: Chair—Any Committee of the Board other than the Audit Committee, the Compensation Committee or the Governance and Directors Nominating Committee. The Annual Retainer for service as the Chair of any committee of the Board other than the Audit Committee, the Compensation Committee or the Governance and Directors Nominating Committee shall be \$5,000.

(viii) Annual Retainer: Member—Any Committee of the Board other than the Audit Committee, the Compensation Committee or the Governance and Directors Nominating Committee. The Annual Retainer for service as a member of any committee of the Board other than the Audit Committee, the Compensation Committee or the Governance and Directors Nominating Committee of the Board other than the Audit Committee, the Compensation Committee or the Governance and Directors Nominating Committee shall be \$2,500.

(ix) Committee Meeting Fee—Audit Committee, Compensation Committee and Governance and Directors Nominating Committee. The fee payable for attendance in person or via telephone at a Committee Meeting of the Audit Committee, the Compensation Committee or the Governance and Directors Nominating Committee shall be \$2,500.

(x) Committee Meeting Fee—Any Committee of the Board other than the Audit Committee, the Compensation Committee or the Governance and Directors Nominating Committee. The fee payable for attendance in person or via telephone at a Committee Meeting of any Committee other than the Audit Committee, the Compensation Committee or the Governance and Directors Nominating Committee shall be \$1,000.

(xi) Stock Grants. This Paragraph 3(a)(xi) shall apply as of November 20, 2005 and as of November 20 of each Plan Year beginning after 2005, provided that, if the shareholders of the Company decline to approve the amendments to the Restricted Stock Plan at the Annual Meeting of Shareholders of the Company to be held in 2005 to expand eligibility for grants under the 2002 Restricted Stock Plan to include Non-Employee Directors, no grants of Restricted Stock or Restricted Stock Units shall be

made to Non-Employee Directors under this Paragraph 3(a)(xi) and Paragraph 3(a)(xii) shall continue to apply.

(A) The Board shall grant Restricted Stock for Shares having a Fair Market Value on the date of grant of \$100,000, provided that with respect to each individual who first becomes a Non-Employee Director after November 20, 2005, the Board shall grant Restricted Stock for Shares determined as follows:

Date of Commencement of Service as a Non-Employee Director

Number of Shares Subject to Grant of Restricted Stock

After November 20 of a Plan Year and before the next following February 20 States and set of the se	Shares having a Fair/Market Value on the date of grant of . \$100,000
On or after February 20 of a Plan Year and before the next	Shares having a Fair Market Value on the date of grant of
following May 20	\$75,000
On or after May 20 of a Plan Year and before the next 1 , 111 following August 20, 121 (1997)	Shares having a Fair Market Value on the date of grant of 550,000
On or after August 20 of a Plan Year and before the next	Shares having a Fair Market Value on the date of grant of
following November 20	\$25,000

Each Share of Restricted Stock shall (1) be fully and immediately vested on the date of grant, and (2) bear such other terms and conditions as shall be determined by the Board in its discretion.

(B) In the event that Shares are changed into or exchanged for a different number or kind of shares of stock or other securities of the Company, whether through merger, consolidation, reorganization, recapitalization, stock dividend, stock split-up or other substitution of securities of the Company, the number and class of shares of stock subject to the grant of Restricted Stock Units under the Plan shall be adjusted consistent with the adjustment made pursuant to the Restricted Stock Plan (or such other more recently-adopted generally applicable Plan pursuant to which the Company grants restricted stock or restricted stock units), and such adjustment shall be effective and binding for all purposes of this Plan.

(xii) Stock Options. This Paragraph 3(a)(xii) shall apply as of November 20, 2002 and as of November 20 of each Plan Year beginning after 2002, provided that, if the shareholders of the Company approve the amendments to the Restricted Stock Plan at the Annual Meeting of Shareholders of the Company to be held in 2005 to expand eligibility for grants under the 2002 Restricted Stock Plan to include Non-Employee Directors, no grants of non-qualified options shall be made to Non-Employee Directors under this Paragraph 3(a)(xii) after November 19, 2005, and Paragraph 3(a)(xi) shall apply.

(A) The Board shall grant non-qualified options to purchase 7,500 Shares to each Non-Employee Director who is in service as of each such date; provided that with respect to each individual who first becomes a Non-Employee Director after November 20, 2002, the Board shall grant a number of non-qualified options to purchase Shares determined as follows:

Date of Commencement of Service as a Non-Employee Director	Number of Shares Subject to Grant of Non-Qualified Options
After November 20 of a Plan Year and before the next	
following February 20 -	лууна тараат 17;500
On or after February 20 of a Plan Year and before the next	
following May 20	5,625
On or after May 20 of a Plan Year and before the next	113.1117年7月,人民的方法是一些资料的关系是一些资源的资源的资源。1992年1月,1992年1月,1992年1月,1992年1月,1992年1月,1992年1月,1992年1月,1992年1月,1992年1月,199
following August 20	3,750
On or after August 20 of a Plan Year and before the next	
following November 20	1,875
	3

Each non-qualified option shall (1) generally be exercisable for 10 years from the date of grant, provided that options, to the extent then exercisable, shall be exercisable for 90 days following a termination of service for any reason other than death, disability or attainment of a mandatory retirement age; (2) with respect to any non-qualified option granted after January 12, 2005, be vested and exercisable in full on the date of grant; (3) have an option price equal to the fair market value of the option share on the date of grant; and (4) bear such other terms and conditions as shall be determined by the Board in its discretion.

(B) In the event that Shares are changed into or exchanged for a different number or kind of shares of stock or other securities of the Company, whether through merger, consolidation, reorganization, recapitalization, stock dividend, stock split-up or other substitution of securities of the Company, the number and class of shares of stock subject to the grant of Non-Qualified Options under the Plan shall be adjusted consistent with the adjustment made pursuant to the Comcast Corporation 2002 Stock Option Plan (or such other more recently-adopted generally applicable Plan pursuant to which the Company grants stock options), and such adjustment shall be effective and binding for all purposes of this Plan.

(b) Payment Practices. Payments, grants and awards described in Paragraph 3(a) of the Plan shall be subject to the following payment practices:

(i) Annual Retainer payments described in Paragraphs 3(a)(i), 3(a)(ii), 3(a)(v), 3(a)(v), 3(a)(vi), 3(a)(vi) and 3(a)(viii) are payable as soon as reasonably practicable following the close of each calendar quarter, in arrears. Payments shall be pro-rated for partial years of service as a Non-Employee Director or on a Committee of the Board, so that a Non-Employee Director shall be entitled to one-quarter of each Annual Retainer payment referenced in this Paragraph 3(b)(i) for each calendar quarter within which such Non-Employee Director has one or more days of service as a Non-Employee Director. The Annual Retainer amounts adopted as part of the amendment and restatement of the Plan effective February 26, 2003 shall apply for the first calendar quarter of 2003 for any Non-Employee Director in service as a Non-Employee Director (including with respect to Committee assignments) for the period from February 26, 2003 through March 31, 2003.

(ii) A Non-Employee Director may elect to receive up to 50% of the Annual Retainer amount described in Paragraph 3(a)(i) and payable after 2002 in the form of Shares. The number of Shares payable to a Non-Employee Director shall be determined based on the closing price of Shares on the last business day of each calendar quarter.

4. ADMINISTRATION OF THE PLAN

The Plan shall be administered by the Board. Subject to the express terms and conditions set forth in the Plan, the Board shall have the power, from time to time, to interpret the Plan's provisions, prescribe, amend and rescind rules and regulations for the Plan, and make all other determinations necessary or advisable for the administration of the Plan. The determination of the Board in all matters as stated above shall be conclusive.

5. TAXES

The Company shall withhold the amount of any federal, state, local or other tax, charge or assessment attributable to the grant of any Award or lapse of restrictions under any Award as it may deem necessary or appropriate, in its sole discretion.

6. AMENDMENT AND TERMINATION

The Plan may be amended or terminated by the Board at any time. No accrued right to payment as determined under Paragraph 3 shall be affected by any such termination or amendment without the written consent of the affected Non-Employee Director.

7. EFFECTIVE DATE

- --

The effective date of this amendment and restatement of the Plan is January 12, 2005. The original effective date of the Plan is November 18, 2002.

8. GOVERNING LAW

.

The Plan and all determinations made and actions taken pursuant to the Plan shall be governed in accordance with Pennsylvania law.

(COMCAST CORPORATION		
i	BY:		
	ATTEST:	<u></u>	
	5		

SCHEDULE I COMCAST CORPORATION NON-EMPLOYEE DIRECTOR COMPENSATION 2005

 $\cdot = \cdot$

Director Annual Retainer	\$50,000, subject to election to receive up to half in the form of Concast
	Corporation Class A Common Stock
Board Meeting Fee	\$2,500 http://www.www.www.www.www.www.www.www.www.w
Audit Committee Annual Retainer-Chair	\$20,000
Compensation Committee Annual Retainer—Chair	\$10,000
Governance and Directors Nominating Committee Annual Retainer-Chair	\$10,000
Other Committee Annual Retainer-Chair	\$5,000
Audit Committee Annual Retainer-Member 3	\$10,000
Compensation Committee Annual Retainer—Member	\$5,000
Governance and Directors Nominating Committee Annual Retainer-Member	\$5,000 Weight Washington Provide Automatic Street Stre
Other Committee Annual Retainer—Member	\$2,500
Committee Meeting Fee-Audit Committee - 11 - 22 - 22	\$2,500
Committee Meeting Fee-Compensation Committee	\$2,500
Committee Meeting Fee-Covernance and Directors Nominating Committee	\$2,500
Committee Meeting Fee-Other Committee	\$1,000
Annual Restricted Stock Grant ²	Shares having a Pair Market Value on the date of grant of \$100,000
E THE FORMER FOR THE STREET OF THE THE STREET OF THE ST	ara lanan u kata a mandinan tara tara tarakin dari di kata dari kata bina bina bara dari da tara dari kata dari Arabiteri dari kata dari dari dari dari dari dari dari kata dari dari da dari da dari dari dari da

I

Fee will also be paid when a member of the Board is asked to attend a meeting or otherwise to conduct business on behalf of the Company in his/her capacity as Director.

2

Subject to shareholder approval, as described in Paragraphs 3(a)(xi) and 3(a)(xii) of the Plan. If the shareholders decline to approve the amendments to the Restricted Stock Plan at the Annual Meeting of Shareholders of the Company to be held in 2005 to expand eligibility for grants under the 2002 Restricted Stock Plan to include Non-Employee Directors, Non-Employee Directors shall be entitled to an annual grant of non-qualified stock options, as further described in Paragraph 3(a)(xii) of the Plan.

QuickLinks

COMCAST CORPORATION 2002 NON-EMPLOYEE DIRECTOR COMPENSATION PLAN (As Amended and Restated, Effective January 12, 2005)

-

COMCAST CORPORATION

2002 EMPLOYEE STOCK PURCHASE PLAN

(As Amended and Restated, Effective January 1, 2005)

1. Purpose.

COMCAST CORPORATION, a Pennsylvania corporation, hereby amends and restates the Comcast Corporation 2002 Employee Stock Purchase Plan (the "Plan"), effective January 1, 2005. The Plan is intended to encourage and facilitate the purchase of shares of common stock of Comcast Corporation by Eligible Employees of the Company and any Participating Companies, thereby providing such Eligible Employees with a personal stake in the Company and a long-range inducement to remain in the employ of the Company and Participating Companies. It is the intention of the Company that the Plan qualify as an "employee stock purchase plan" within the meaning of section 423 of the Code.

2. Definitions.

(a) "Account" means a bookkeeping account established by the Committee on behalf of a Participant to hold Payroll Deductions.

(b) "Affiliate" means, with respect to any Person, any other Person that, directly or indirectly, is in control of, is controlled by, or is under common control with, such Person. For purposes of this definition, the term "control," including its correlative terms "controlled by" and "under common control with," mean, with respect to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract or otherwise.

(c) "Board" means the Board of Directors of the Company.

(d) "Brokerage Account" means the brokerage account established under the Plan by the Company for each Participant, to which Shares purchased under the Plan shall be credited.

(e) "Change of Control" means any transaction or series of transactions as a result of which any Person who was a Third Party immediately before such transaction or series of transactions owns then-outstanding securities of the Company such that such Person has the ability to direct the management of the Company, as determined by the Board in its discretion. The Board may also determine that a Change of Control shall occur upon the completion of one or more proposed transactions. The Board's determination shall be final and binding.

- (f) "Code" means the Internal Revenue Code of 1986, as amended.
- (g) "Committee" means the Compensation Committee of the Board.

(h) "Company" means Concast Corporation, a Pennsylvania corporation, including any successor thereto by merger, consolidation, acquisition of all or substantially all the assets thereof, or otherwise.

(i) "Compensation" means an Eligible Employee's wages as reported on Form W-2 (*i.e.*, wages as defined in section 3401(a) of the Code and all other payments of compensation for which the Participating Company is required to furnish the employee a written statement under sections 6041(d) and 6051(a)(3) of the Code) from a Participating Company, reduced by reimbursements or other expense allowances, fringe benefits (cash and non-cash), moving expenses, deferred compensation, and welfare benefits, but including salary reduction contributions and elective contributions that are not includible in gross income under sections 125 or 402(a)(8) of the Code.

(j) "Election Form" means the written or electronic form acceptable to the Committee which an Eligible Employee shall use to make an election to purchase Shares through Payroll Deductions pursuant to the Plan.

(k) "Eligible Employee" means an Employee who is not an Ineligible Employee. Notwithstanding the foregoing to the contrary, solely for purposes of the Offering Period commencing on October 1, 2002, the

term "Eligible Employee" means an Employee who was eligible to participate in this Plan immediately before October 1, 2002.

- (1) "Eligible Employer" means the Company and any subsidiary of the Company, within the meaning of section 424(f) of the Code.
- (m) "Employee" means a person who is an employee of a Participating Company.

(n) "Fair Market Value" means the closing price per Share on the principal national securities exchange on which the Shares are listed or admitted to trading or, if not listed or traded on any such exchange, on the National Market System of the National Association of Securities Dealers Automated Quotation System ("NASDAQ"), or if not listed or traded on any such exchange or system, the fair market value as reasonably determined by the Board or the Committee, which determination shall be conclusive.

- (o) "Five Percent Owner" means an Employee who, with respect to a Participating Company, is described in section 423(b)(3) of the Code.
- (p) "Ineligible Employee" means an Employee who, as of an Offering Commencement Date:
 - (1) is a Five Percent Owner;
 - (2) has been continuously employed by a Participating Company on a full-time basis for less than 90 days;
 - (3) has been continuously employed by a Participating Company on a part-time basis for less than one year; or
 - (4) is restricted from participating under Paragraph 3(b).

For purposes of this Paragraph 2(p), an Employee is employed on a part-time basis if the Employee customarily works less than 20 hours per week. For purposes of this Paragraph 2(p), an Employee is employed on a full-time basis if the Employee customarily works 20 or more hours per week.

(q) "Offering" means an offering of Shares by the Company to Eligible Employees pursuant to the Plan.

(r) "Offering Commencement Date" means the first day of each January 1, April 1, July 1 and October 1 beginning on or after Offerings are authorized by the Board or the Committee, until the Plan Termination Date, provided that the first Offering Commencement Date shall be on the Effective Date.

(s) "Offering Period" means the period extending from an Offering Commencement Date through the following Offering Termination Date.

(t) "Offering Termination Date" means the last day of each March, June, September and December following an Offering Commencement Date, or such other Offering Termination Date established in connection with a Terminating Event.

(u) "Participant" means an Eligible Employee who has timely delivered an Election Form to the Committee in accordance with procedures established by the Committee.

(v) "Participating Company" means, as provided in Schedule A to the Plan, the Eligible Employers, if any, that are approved by the Board or the Committee from time to time.

(w) "Payroll Deductions" means amounts withheld from a Participant's Compensation pursuant to the Plan, as described in Paragraph 5.

(x) "Person" means an individual, a corporation, a partnership, an association, a trust or any other entity or organization.

(y) "Plan" means the Comcast Corporation 2002 Employee Stock Purchase Plan, as set forth in this document, and as may be amended from time to time.

(z) "Plan Termination Date" means the earlier of:

(1) the Offering Termination Date for the Offering in which the maximum number of Shares specified in Paragraph 9 have been issued pursuant to the Plan; or

(2) the date as of which the Board or the Committee chooses to terminate the Plan as provided in Paragraph 14.

(aa) "Purchase Price" means 85 percent of the lesser of: (1) the Fair Market Value per Share on the Offering Commencement Date, or if such date is not a trading day, then on the next trading day thereafter or (2) the Fair Market Value per Share on the Offering Termination Date, or if such date is not a trading day, then on the trading day immediately preceding the Offering Termination Date.

(bb) "Shares" means:

- (1) except as otherwise provided in Paragraph 2(bb)(2), shares of Comcast Corporation Class A Common Stock, par value \$0.01.
- (2) for the Offering Period commencing on October 1, 2002, shares of Comcast Corporation Class A Special Common Stock, par value \$0.01.

(cc) "Successor-in-Interest" means the Participant's executor or administrator, or such other person or entity to whom the Participant's rights under the Plan shall have passed by will or the laws of descent and distribution.

(dd) "Terminating Event" means any of the following events:

- (1) the liquidation of the Company; or
- (2) a Change of Control.

(ee) "Third Party" means any Person, together with such Person's Affiliates, provided that the term "Third Party" shall not include the Company or an Affiliate of the Company.

(ff) "Termination Form" means the written or electronic form acceptable to the Committee which an Employee shall use to discontinue participation during an Offering Period pursuant to Paragraph 7(b).

3. Eligibility and Participation.

(a) Eligibility. Except to the extent participation is restricted under Paragraph 3(b), each Eligible Employee shall be eligible to participate in the Plan.

(b) Restrictions on Participation. Notwithstanding any provisions of the Plan to the contrary, no Employee shall be eligible to purchase Shares in an Offering to the extent that:

(1) immediately after the purchase of Shares, such Employee would be a Five Percent Owner; or

(2) a purchase of Shares would permit such Employee's rights to purchase stock under all employee stock purchase plans of the Participating Companies which meet the requirements of section 423(b) of the Code to accrue at a rate which exceeds \$25,000 in fair market value (as determined pursuant to section 423(b)(8) of the Code) for each calendar year in which such right to purchase Shares is outstanding.

(c) Commencement of Participation. An Eligible Employee shall become a Participant by completing an Election Form and filing it with the Committee on or before the 15th day of the month immediately preceding the Offering Commencement Date for the first Offering to which such Election Form applies. Payroll Deductions for a Participant shall commence on first payroll period ending after the applicable Offering Commencement Date when his or her authorization for Payroll Deductions becomes effective, and shall end on the Plan Termination Date, unless sooner terminated by the Participant pursuant to Paragraph 7(b).

4. Shares Per Offering.

The Plan shall be implemented by a series of Offerings that shall commence after Offerings have been authorized by the Board or the Committee, and terminate on the Plan Termination Date. Offerings shall be made with respect to Compensation accumulated during each Offering Period for the period commencing with the first day of the first Offering Period (when such Offering Period is authorized by the Board or the Committee) and ending with the Plan Termination Date. Shares available for any Offering shall be the difference between the maximum number of Shares that may be issued under the Plan, as determined pursuant to Paragraph 8(a), for all of the Offerings, less the actual number of Shares purchased by Participants pursuant to prior Offerings. If the total number of Shares subject to purchase under the Plan on any Offering Termination Date exceeds the maximum number of Shares available, the Board or the Committee shall make a pro rata allocation of Shares available for delivery and distribution in as nearly a uniform manner as practicable, and as it shall determine to be fair and equitable, and the unapplied Account balances shall be returned to Participants as soon as practicable following the Offering Termination Date.

5. Payroll Deductions.

(a) Amount of Payroll Deductions. On the Election Form, an Eligible Employee may elect to have Payroll Deductions of not more than 10 percent of Compensation earned for each payroll period ending within the Offering Period, subject to the limitation that the maximum amount of Payroll Deductions for any Eligible Employee for any calendar year shall not exceed \$10,000. The rules established by the Committee regarding Payroll Deductions, as reflected on the Election Form, shall be consistent with section 423(b)(5) of the Code.

(b) Participants' Accounts. All Payroll Deductions with respect to a Participant pursuant to Paragraph 5(a) shall be credited to the Participant's Account under the Plan.

(c) Changes in Payroll Deductions. A Participant may discontinue Payroll Deductions during an Offering Period by providing a Termination Form to the Committee at any time before the Offering Termination Date applicable to any Offering. No other change can be made during an Offering, including, but not limited to, changes in the amount of Payroll Deductions for such Offering. A Participant may change the amount of Payroll Deductions for subsequent Offerings by giving written notice (or notice in another form pursuant to procedures established by the Committee) of such change to the Committee on or before the 15th day of the month immediately preceding the Offering Commencement Date for the Offering for which such change is effective.

6. Purchase of Shares.

(a) In General. On each Offering Termination Date, each Participant shall be deemed to have purchased a number of whole Shares equal to the quotient obtained by dividing the balance credited to the Participant's Account as of the Offering Termination Date, by the Purchase Price, rounded to the next lowest whole Share. Shares deemed purchased by a Participant under the Plan shall be credited to the Participant's Brokerage Account as soon as practicable following the Offering Termination Date.

(b) Terminating Events. The Company shall give Participants at least 30 days' notice (or, if not practicable, such shorter notice as may be reasonably practicable) prior to the anticipated date of the consummation of a Terminating Event. The 20th day following the issuance of such notice by the Company (or such earlier date as the Board or the Committee may reasonably determine) shall constitute the Offering Termination Date for any outstanding Offering.

(c) Fractional Shares and Minimum Number of Shares. Fractional Shares shall not be issued under the Plan. Amounts credited to an Account remaining after the application of such Account to the purchase of Shares under the Plan shall be credited to the Participant's Account for the next succeeding Offering, or, at the Participant's election, returned to the Participant as soon as practicable following the Offering Termination Date, without interest.

(d) Transferability of Rights to Purchase Shares. No right to purchase Shares pursuant to the Plan shall be transferable other than by will or by the laws of descent and distribution, and no such right to purchase Shares pursuant to the Plan shall be exercisable during the Participant's lifetime other than by the Participant.

7. Termination of Participation.

(a) Account. Except as provided in Paragraph 7(c), no amounts shall be distributed from Participants' Accounts during an Offering Period.

(b) Suspension of Participation. A Participant may discontinue Payroll Deductions during an Offering Period by providing a Termination Form to the Committee at any time before the Offering Termination Date applicable to any Offering. All amounts credited to such Participant's Account shall be applied to the purchase of Shares pursuant to Paragraph 6. A Participant who discontinues Payroll Deductions during an Offering Period shall not be eligible to participate in the Offering next following the date on which the Participant delivers the Termination Form to the Committee.

(c) Termination of Employment. Upon termination of a Participant's employment for any reason, all amounts credited to such Participant's Account shall be returned to the Participant, or, following the Participant's death, to the Participant's Successor-in-Interest.

8. Interest.

No interest shall be paid or allowed with respect to Payroll Deductions paid into the Plan or credited to any Participant's Account.

9. Shares.

(a) Maximum Number of Shares; Adjustments. Subject to adjustment as provided in this Paragraph 9, not more than 4,250,000 Shares in the aggregate may be issued pursuant to the Plan pursuant to Offerings under the Plan, including Offerings commenced since the Plan first became effective as the Comcast Corporation 2001 Employee Stock Purchase Plan. Shares delivered pursuant to the Plan may, at the Company's option, be either treasury Shares or Shares originally issued for such purpose. In the event that Shares are changed into or exchanged for a different number or kind of shares of stock or other securities of the Company, whether through merger, consolidation, reorganization, recapitalization, stock dividend, stock split-up or other substitution of securities of the Company, the Board or the Committee shall make appropriate equitable anti-dilution adjustments to the number and class of stock available for issuance under the Plan, to the number and class of shares of stock subject to outstanding Offerings and to the Purchase Price. Any reference to the Purchase Price in the Plan and in any related documents shall be a reference to the appropriate number and class of shares of stock available for issuance under the Plan, as adjusted pursuant to this Paragraph 9. The Board's or the Committee's adjustment shall be effective and binding for all purposes of this Plan. All Shares issued pursuant to the Plan shall be validly issued, fully paid and nonassessable.

(b) Participant's Interest in Shares. A Participant shall have no interest in Shares offered under the Plan until Shares are credited to the Participant's Brokerage Account.

(c) Crediting of Shares to Brokerage Account. Shares purchased under the Plan shall be credited to the Participant's Brokerage Account as soon as practicable following the Offering Termination Date.

(d) Restrictions on Purchase. The Board or the Committee may, in its discretion, require as conditions to the purchase of any Shares under the Plan such conditions as it may deem necessary to assure that such purchase of Shares is in compliance with applicable securities laws.

10. Expenses.

The Participating Companies shall pay all fees and expenses incurred (excluding individual Federal, state, local or other taxes) in connection with the Plan. No charge or deduction for any such expenses will be made to a Participant upon the termination of his or her participation under the Plan or upon the distribution of certificates representing Shares purchased with his or her Payroll Deductions.

11. Taxes.

The Participating Companies shall have the right to withhold from each Participant's Compensation an amount equal to all federal, state, city or other taxes as the Participating Companies shall determine are required to be withheld by them in connection with the purchase of Shares under the Plan and in connection with the sale of Shares acquired under the Plan. In connection with such withholding, the Participating Companies may make any such arrangements as they may deem necessary or appropriate to protect their interests.

12. Plan and Contributions Not to Affect Employment.

The Plan shall not confer upon any Eligible Employee any right to continue in the employ of the Participating Companies.

13. Administration.

The Plan shall be administered by the Committee. The Board and the Committee shall have authority to interpret the Plan, to prescribe, amend and rescind rules and regulations relating to it, and to make all other determinations deemed necessary or advisable in administering the Plan, with or without the advice of counsel. The Committee may delegate its administrative duties, subject to its review and supervision, to the appropriate officers and employees of the Company. The determinations of the Board and the Committee on the matters referred to in this Paragraph 13 shall be conclusive and binding.

14. Amendment and Termination.

The Board or the Committee may terminate the Plan at any time and may amend the Plan from time to time in any respect; provided, however, that upon any termination of the Plan, all Shares or Payroll Deductions (to the extent not yet applied to the purchase of Shares) under the Plan shall be distributed to the Participants, provided further, that no amendment to the Plan shall affect the right of any Participant to receive his or her proportionate interest in the Shares or his or her Payroll Deductions (to the extent not yet applied to the purchase of Shares) under the Plan, and provided further that the Company may seek shareholder approval of an amendment to the Plan if such approval is determined to be required by or advisable under the regulations of the Securities and Exchange Commission or the Internal Revenue Service, the rules of any stock exchange or system on which the Shares are listed or other applicable law or regulation.

15. Effective Date.

The original effective date of the Plan was December 20, 2000. This amendment and restatement of the Plan is effective on January 1, 2005.

16. Government and Other Regulations.

(a) In General. The purchase of Shares under the Plan shall be subject to all applicable laws, rules and regulations, and to such approvals by any governmental agencies as may be required.

(b) Securities Law. The Committee shall have the power to make each Offering under the Plan subject to such conditions as it deems necessary or appropriate to comply with the then-existing requirements of the Securities Act of 1933, as amended, and the Securities Exchange Act of 1934, as amended, including Rule 16b-3 (or any similar rule) promulgated by the Securities and Exchange Commission thereunder.

17. Non-Alienation.

No Participant shall be permitted to assign, alienate, sell, transfer, pledge or otherwise encumber his right to purchase Shares under the Plan prior to time that Shares are credited to the Participant's Brokerage Account. Any attempt at assignment, alienation, sale, transfer, pledge or other encumbrance shall be void and of no effect.

18. Notices.

Any notice required or permitted hereunder shall be sufficiently given only if delivered personally, telecopied, or sent by first class mail, postage prepaid, and addressed:

If to the Company:

Comcast Corporation 1500 Market Street Philadelphia, PA, 19102 Fax: 215–981–7794 Attention: General Counsel

Or any other address provided pursuant to notice provided by the Committee.

If to the Participant:

At the address on file with the Participating Company from time to time, or to such other address as either party may hereafter designate in writing (or via such other means of communication permitted by the Committee) by notice similarly given by one party to the other.

19. Successors.

The Plan shall be binding upon and inure to the benefit of any successors or assigns of the Company.

20. Severability.

If any part of this Plan shall be determined to be invalid or void in any respect, such determination shall not affect, impair, invalidate or nullify the remaining provisions of this Plan which shall continue in full force and effect.

21. Acceptance.

The election by any Eligible Employee to participate in this Plan constitutes his or her acceptance of the terms of the Plan and his or her agreement to be bound hereby.

22. Applicable Law.

This Plan shall be construed in accordance with the laws of the Commonwealth of Pennsylvania, to the extent not preempted by applicable Federal law.

Executed as of the 1st day of January, 2005.

COMCAST CORPORATION

BY:

ATTEST:

- -

.

Effective as of January 1, 2005
Comcast Business Communications Holdings, Inc. and its subsidiaries
Comcast Cable Communications Holdings, Inc. and its subsidiaries
Comcast Cable Communications, LLC, and its subsidiaries
Comcast Corporation
Comcast Holdings Corporation
Comcast Online Communications, Inc.
Comcast Shared Services Corporation
Comcast SportsNet West, Inc.
G4 Media, LLC
Home Team Sports Limited Partnership
International Channel
Outdoor Life Network, LLC
Philadelphia Sports Media, L.P.

.

Entity Name Org State 1080/Sports_LLC6_T ABB MOG-WM, Inc. CO ABB RFL, LLC 医丁基酚基基合生甲酸氢酸胡 》DE 学习电话,带出话 ABB TS Assets, LLC Alabama T.V. Cable, Inc. American Microwave & Communications, Inc. DE MI American Televenture of Minersville, Incl. COMENTIAL 行而列表现 化多多电缆 叶紫 Atlantic American Cablevision of Florida, LLC Atlantic American Cablevision, LLC FL **新了的新作用**了4月8 1.1.73。在16月1日中学习5月 DE Atlantic American Holdings, Inc. FL Auantic Cablevision of Florida, Inc. 档 DE包 **Bay Area Interconnect** ĊA Beatrice Cable TV Company, alter and the second s NE Box Office Enterprises, Inc. CT 合同的Therapy DELETION。DELETION。 Devere A. AL BroadNet Austria GmbH Austria BroadNet Belgium S.A Belgium AS ALL SHE WAT THE ALL BroadNet Czech a.s. Czech Republic Czech Republic 2.4. BroadNet Czech s.r.o. 57 BroadNet Danmark ApS Denmark BroadNet Europe SPRL 这些原则是不是"这个呢?" 第二章 Belgium BroadNet France S.A.S. BroadNet Hellas S.A. France Greece W. M. Ful **和卡爾尼加德德曼比加阿爾阿阿阿阿阿**爾德加爾德加爾 10-5-47-5 The Netherlands Italy BroadNet Holdings, B.V. BroadNet Italy SPA E CALL AND LADA Hungary BroadNet Magyarorszag Kft BroadNet Norge AtS Norway Portugal BroadNet Portugal, S.A BroadNet Slovakia sir.o. BroadNet Suisse A.S. Switzerland BroadNet UK Ltd. UK ·清朝台: [1] - [1] - [1] Cable Accounting, Inc. Cable Enterprises, Inc. CO 14.5.2 、3.6.4.5.1 网络普索德国新语 DE ्रभू Cable Programming Ventures, LLC DE Cable Sports Southeast, LLC BDB AN A REAL AND A CONTRACT OF 大的资源和特殊信 Cable Television Advertising Group, Inc. Cable Television of Gary, Inc. 1971 WY IN COMPANY SERVICE Cablevision Associates of Gary Joint Venture IN Cablevision Investment of Detroit, Inc. M

Exhibit 21

Cablevision of Arcadia/Sierra Madre, Inc.	
CCC=NIRT, Inc.	
CCF Management Services, Inc.	
CCP Management Services, Inc.	
Convianagement Services, Inc. 1977 Annual Services and Annual Ser	
Circ Development Corp. 1977 March 1977 Circles and 1977 C	
Classic Services, Inc.	
Clinton Cable TV Investors Inc. 1	
Clinton TV Cable Company, LLC	
Coastal Cable TV Inc	
Colorado Terrace Tower II Corporation	
COM Indiana, LLC	
COM Indianapolis, LLC DE	
COM Takster The	
COM MH, LLC DE	
COM/South_ELC	
COM Sports Holding Company, Inc. DE	
COM Sports Ventures Inc	
Combined Conditional Access Development and Support, LLC Comcast 38GHZ, Inc.	
Comcast 38GHZ, Inc.	
Comcast ABB Business Services, Inc.	
Comcast ABB Cablevision V. Inc. 2007 1000 1000 1000 1000 1000 1000 1000	
Comcast ABB CSC Holdings, Inc.	
Comcast ABB CSC II. Inc.	
Comcast ABB Holdings I: Inc	
Comcast ABB Holdings II, Inc. Comcast ABB LCI, Inc. DE	
Comcast ABB Management Corporation CO Comcast ABB Network Solutions, Inc.	
Comcast ABB NOC, LLC DE	
Concast ABB Note, LLC DE Consolidation Inc.	
Concest ABB of Clinton	
Comcast ABB of Georgia II; LLC 2:1	
Comesst ABB of Kiowa LLC	
Comcast ABB of Mississippi/Towa_LLC	
Comcast ABB of Payette, Inc. OR	
Comcast ABB Optionee Payroll*LLC	
Comcast ABB Overseas Holdings 1, LLC DE	
Comcast ABB Overseas Holdings II. LLC	
Comcast ABB Overseas Holdings, Inc. DE	
2	

Comcast ABB USC ILLC
Comcast Argentina, Inc. Comcast ASBC-Inc.
Comcast ASBC Inc. DE
Concast BroadNet Payroll Services Inc. 1990 A service of the servi
Comcast Business Communications of Virginia, LLC Comcast Business Communications, Inc. 1999 August 2019 August 2019 August 2019 August 2019 August 2019 August 2019
Concast Cable Communications Holdings, Inc.
Concast Cable Communications Holdings, LLC
Concast Cable Communications Management, LLC DE
Concast Cable Communications/ LLC
Comcast Cable Funding DE
Comcast Cable Funding GP/Inc. And Market States and State
Comcast Cable Funding I, Inc. DE
Comcast Cable Holdings/LLC, General States of Frances and States and States and DE States and DE States and Stat
Comcast Cable of Indiana, Inc.
Comcast Cable of /indiana/Michigan/Texas I/LL/C 342/2014/10/2014/10/2014/2014/2014/2014/2
Concest Cable of Maryland, Inc. DE
Comcast Cable SC Investment Inc.
Comcast Cable Trust II Comment of the Comment of th
Comcast Cable Trust III DE Comcast Cablevision Communications, Inc.
Comcast Cablevision of Baltimore City GP, Inc.
Comcast Cablevision of Garden State, Inc.
Concest Cablevision of Philadelphia Area I, LLC PA
Comcast Cablevision of Southeast Michigan, Inc.
Comcast Capital Corporation DE
Comcast CCH Subsidiary Holdings, Inc. 41 (55)
Comcast Cellular Holding Company, Inc. DE
Comcast Cellular Holdings Corporation 12
Comcast CHC Subsidiary Holdings, Inc. DE
Comcast Children's Network Holdings, LLC
Comeast CICG GP, Inc. DE
Concest CICGLP, Inc.
Conicast Commercial Services Financing Inc. 49 20 14 14 14 19 19 19 19 19 19 19 19 19 19 19 19 19
Comcast Commercial Services Group Holdings, LLC Comcast Commercial Services, Inc. 1997
Comcast Concurrent Holdings, Inc. DE
Concast Corporate Investments III/Inc.

Concast Corporation Political Action Committee
Comcast Corporation Political Action Committee of Maryland MD Comcast Corporation Political Action Committee-USA
Comcast Corporation Political Action Committee-USA1
Concast Corporation Trust I DE
Concast Corporation Trust II : 14
Comcast Corporation Trust III Comcast Crystalvision Inc.
Comcast Data Services, Inc. Comcast DC Radio Inc.
Concast Discrete Supervises, Inc.
Concast do Brasil Lida (control and the state of the stat
Comcast Encore, Inc.
Comcast Entertainment Holdings LLC
Comcast Financial Agency Corporation DE
Concast Florida Programming Investments Inc. 1991 2013 1991 1991 1991 1991 1991 1991 1991 1
Comcast Funding I, Inc. DE
Comcast Garden State, LLC 4311
Comcast Gateway Holdings, LLC DE
Comcast Greater Boston Advertising Holdings, LLC
Comcast Holdings Corporation PA
Comcast Holdings II-LLC
Comcast Holdings III, LLC DE
Comcast Holdings'IV LLC
Comcast Holdings V, LLC DE DE Comcast HTS Holdings, LLC
Comcast HTS, LLC DE Comcast ICCP/Inc++++++++++++++++++++++++++++++++++++
Comcast ICC, Inc.
Comcast In Demand Holdings, Inc.
Concast Interactive Capital, LP
Comcast International Holdings Inc.
Comcast Investment Holdings, Inc. Comcast IP, Holdings I, LLC, State
Concast IP Phone of Arizona, LLC DE
Comcast IP, Phone of California ULC
Comcast IP Phone of Central Indiana, LLC DE
Comcast IP. Phone of Colorado, LLC
Comcast IP Phone of Connecticut, LLC
Comcast IP Phone of Delaware; LLC
Comcast IP Phone of Florida, LLC
Comcast IP Phone of Georgia LLC
Comcast IP Phone of Illinois, LLC DE
4

Comcast IP. Phone of Maryland, LLC
Comcast IP Phone of Massachusetts, LLC DE
Comcast IPPhone of Michigan LLC NEW WAR WAR WAR WAR WAR WAR WAR WAR WAR WA
Conncast IP Phone of Minnesota, LLC DE
Comcast IP. Phone of New Hampshire, LLC 14, 19 19 19 19 19 19 19 19 19 19 19 19 19
Concast IP Phone of New Jersey, LLC Concast IP Phone of New Mexico, ILLC 17 First 1999 1997 1997 1997 1997 1997 1997 199
Concast IP Phone of Ohio, LLC DE
Comcast IP, Phone of Oregon, LLC
Comcast IP Phone of Pennsylvania, LLC PA
Comcast IP. Phone of Tennessee, LLC
Concest IP Phone of Texas, LLC DE
Comcast IP Phone of Virginia, LLC
Comcast IP Phone of Virginia, LLC Comcast IP, Phone of Washington, LUC
Comcast IP Phone of West Virginia, LLC
Comcast IP:Phone.line
Comcast IP Services II, Inc. DE
Comcast IP. Services, LLC
Comcast IPG/IV, LLC Comcast ISD, Inc.
Concast ISD, Inc. DE
Comcast LCP Inc.
Comcast Levittown Pinance. Inc.
Comcast Life Insurance Holding Company,
Comcast LMC E! Entertainment, Inc.
Comcast LMDS Communications, find the second s
Comcast Metaty, Inc. Comcast MH Holdings, LLC
Comcast Michigan Holdings, Inc.
Comcast Midwest Management Inc.
Comcast MO Cable Advertising of Metropolitan Atlanta, LLC CO
Comcast MO Cable News, Inc. 11
Comcast MO Capital Corporation CO Comcast MO Communications Holding Company, Inc. 1921
Comcast MO Communications Holding Company, inclusion and the second state of the secon
Comcast MO Digital Radio Inc.
Comcast MO Europe, Inc.
Comcast MO Express Midwest Inc.
Comcast MO Express of California, Inc.
Comcast MO Express of Florida, Inc.
Comcast MO Express of New England, Inc. MA

ļ

-

Concast MO Express of Virginia Inc.	
Comcast MO Federal Relations, Inc.	
Comcast MO Finance Corporation	
Comcast MO Finance Trust I DE	
Comcast MO Finance Trust II	
Comcast MO Finance Trust III Comcast MO Finance Trust IV Data Andrea State Andre	
Comcast MO Finance Trust IV II Comcast MO Finance Trust V	
Comcast MO Finance Trust V Comcast MO Finance Trust VI	
Concast MO Financial Services Foreign Sales, Inc.	
Comcase MO Financial Services; Inc.	
Concast MO Financing A DE	
Comcast MO Financing Bizer and the second	
Comcast MO Foreign Investments. Inc. CO	
Comcast MO ES Leasing 1995; Inc. 44 (March 1994)	
Comcast MO FSC One, Ltd. Bermuda	
Comcast MO FSC Three, Ltd	
Comcast MO FSC Two, Ltd. Comcast MO Group Funding, Inc. 1991 Berny State	
Comcast MO Group, Inc. DE Comcast MO Holdings I. Inc.	
Concest MO Holdings II, Inc.	
Comcast MO Information Technology Systems Inc.	
Comcast MO Interactive Services. Inc.	
Comcast MO International Holdings II, Inc. 44 and 14	
Comcast MO International Programming, Inc. MA	
Concast MO International Inc.	
Comcast MO Investments, Inc. DE Comcast MO Leveraged Lease Partners 1997; LP: 1997; LP	
Comcast MO Marketing Resources (UK) Limited UK Comcast MO of Burnsville/Eagan Inc.	
Comcast MO of Delaware, LLC Comcast MO of Minnesota Inc.	
Comcast MO of North Valley, Inc.	
Concast MO of round values, inc.	
Concess MO of the North Suburbs. Inc. MN	
Comcast MO Racing, Inc.	
Comcast MO Real Estate, Inc. CO	
Comcast MOSPCI/LEC	
Concast MO SPC III (LLC)	
6	

- ----

• •--

Comeas No SPC V1 LLC Comeas Mo SPE In: Comeas No	Comcast MO SPC IV LUC
Concast No. SPE, Inc. DE Gomeast Multicable Media, Inc. DE Gomeast Multicable Media, Inc. DE Gomeast Nach III (LLC Concast NCC Holdings II, LLC Concast NCC Holdings III, LLC Concast New Ool J, Inc. Concast New Ool J, Inc.	Concest MO SPC V, LLC
Concast No. SPE, Inc. DE Gomeast Multicable Media, Inc. DE Gomeast Multicable Media, Inc. DE Gomeast Nach III (LLC Concast NCC Holdings II, LLC Concast NCC Holdings III, LLC Concast New Ool J, Inc. Concast New Ool J, Inc.	Comcast MO SPC VI. LLC
Concast Multicable Media, Inc. DB Concast No.C Holdings II, LLC Concast NCC Holdings III, LLC Concast New Media Development, Inc. Concast New Media Development, Inc. Concast New Media Development, Inc. Concast New Of State, State Stat	Concast MO SPE. Inc.
Generativestimation DB Concast NCC Holdings III, LLC DB Concast New Media Development, Inc. PA Concast New Oils, Inc. DB Concast Newco IS, Inc. DB Concast Newco S, Inc	
Concast NCC Holdings I, LLC DE Concast NCC Holdings III, LLC DE Concast NCC Holdings III, LLC DE Concast New Media Development, Inc. Concast New Media Development, Inc. Concast New Media Development, Inc. Concast New Ol 19 Inc. Concast New Ol 10 Inc. C	
Concast New Co Holdings III, LLC DE Concast New Method Development, Inc. PA Concast New Method Development, Inc. DE Concast New Method Development, Inc. DE Concast New Method Development, Inc. DE Concast New Ool Jane, Holdings III, LLC DE Concast New Ool Jane, Holdings IIII, LLC DE Concast New Ool Jane, Holdings IIIII, LLC DE Concast New Ool Jane, Holdings IIIII, LLC DE Co	
Concast NCC Holdings III, LLC DE Concast New Media Development, Inc. PA Concast New Media Development, Inc. PA Concast New Media Development, Inc. DE Concast New Ot, Inc.	
Concest New Media Development, Inc. Concest New Media Development, Inc. Concest New Office Period Development, Inc. Concest Of Alabama, Inc. Concest Of Ala	
Concast New Mexico/Pennsylvania Finance/Inc. PA Concast Newcol 13/Inc. DB Concast Newcol 13/Inc. DB Concast Newcol 15/Inc. DB Concast Newcol 21/Inc. DB <	
Concast Newco 1, Inc. Concast Newco 2, Inc. Concast Newco 3, Inc. Concast Newco 4, Inc. Concast Newco 5, Inc. Concast Newco 7, Inc.	Comcast New Media Development, Inc. PA
Concast Newco 15, Inc. DB Concast Newco 17, Inc. DB Concast Newco 18, Inc. DB Concast Newco 19, Inc. DB Concast Newco 20, Inc. DB Concast Newco 21, Inc. DB Concast Newco 22, Inc. DB Concast Newco 21, Inc. DB Concast Newco 22, Inc. DB Concast Newco 31, Inc. DB Concast Newco 3, Inc. DB Concast Of Alameda, Inc. CA	Comcast New Mexico/Pennsylvania/Finance/Inc 1.1
Comcast Newco 15. Inc. Comcast Newco 16. Inc. Comcast Newco 17. Inc. Comcast Newco 19. Inc. Comcast Newco 20. Inc. Comcast Newco 20. Inc. Comcast Newco 20. Inc. Comcast Newco 21. Inc. Comcast Newco 22. Inc. Comcast Newco 23. Inc. Comcast Newco 23. Inc. Comcast Newco 23. Inc. Comcast Newco 24. Inc. Comcast Newco 24. Inc. Comcast Newco 26. Inc. Comcast Newco 26. Inc. Comcast Newco 26. Inc. Comcast Newco 27. Inc. Comcast Newco 26. Inc. Comcast Newco 27. Inc. Comcast Newco 26. Inc. Comcast Newco 7. Inc. Comcast Newco 7. Inc. Comcast Newco 9. Inc. Comcast Of Alabama, Inc. Comcast of Baltimore City, ILP. MD	
Concast Newco 16, Inc. DE Concast Newco 17, Inc. DE Concast Newco 2, Inc. DE Concast Newco 3, Inc. DE Concast Newco 4, Inc. DE Concast Newco 4, Inc. DE Concast Newco 7, Inc. DE Concast A (Alabama, Inc. DE Concast of Alameda, Inc. CO Concast o	
Concast Newco 17, Inc. DE Concast Newco 19, Inc. DE Concast Newco 20, Inc. DE Concast Newco 20, Inc. DE Concast Newco 20, Inc. DE Concast Newco 21, Inc. DE Concast Newco 22, Inc. DE Concast Newco 23, Inc. DE Concast Newco 23, Inc. DE Concast Newco 23, Inc. DE Concast Newco 23, Inc. DE Concast Newco 24, Inc. DE Concast Newco 24, Inc. DE Concast Newco 24, Inc. DE Concast Newco 25, Inc. DE Concast Newco 26, Inc. DE Concast Newco 5, Inc. DE Concast Newco 5, Inc. DE Concast Newco 6, Inc. Concast Newco 8, Inc. DE Concast Newco 9, Inc. DE Concast O Alabama, Inc. Concast O Alabama, Inc. Concast of Alamaa, Inc. MD Concast of Bellewe, Inc. MD	
Concast Newco 19, Inc. Concast Newco 20, Inc. Comcast Newco 20, Inc. Comcast Newco 21, Inc. Comcast Newco 22, Inc. Comcast Newco 23, Inc. Comcast Newco 23, Inc. Comcast Newco 23, Inc. Comcast Newco 3, Inc. Comcast Newco 3, Inc. Comcast Newco 4, Inc. Comcast Newco 4, Inc. Comcast Newco 6, Inc. Comcast Newco 7, Inc. Comcast Newco 7, Inc. Comcast Newco 9, Inc. Comcast Of Alabama Inc. Comcast of Alameda, Inc. Comcast of Baltimore City, Inc.	
Comcast Newco 19, Inc. DE Comcast Newco 20, Inc. DE Comcast Newco 21, Inc. DE Comcast Newco 22, Inc. DE Comcast Newco 23, Inc. DE Comcast Newco 23, Inc. DE Comcast Newco 3, Inc. DE Comcast Newco 4, Inc. DE Comcast Newco 4, Inc. DE Comcast Newco 5, Inc. DE Comcast Newco 6, Inc. DE Comcast Newco 6, Inc. DE Comcast Newco 8, Inc. DE Comcast Newco 9, Inc. DE Comcast Newco 9, Inc. DE Comcast Newco 8, Inc. DE Comcast Newco 9, Inc. DE Comcast of Alameda, Inc. CA	
Concast Newco 20, Inc. DE Concast Newco 20, Inc. DE Concast Newco 21, Inc. DE Concast Newco 23, Inc. DE Concast Newco 23, Inc. DE Concast Newco 3, Inc. DE Concast Newco 3, Inc. DE Concast Newco 4, Inc. DE Concast Newco 5, Inc. DE Concast Newco 6, Inc. If Concast Newco 6, Inc. DE Concast Newco 7, Inc. DE Concast Newco 9, Inc. DE Concast Newco 9, Inc. Concast Newco 9, Inc. DE Concast Newco 8, Inc. DE Concast Newco 8, Inc. Concast Newco 9, Inc. Concast of Alameda, Inc.	
Comcast Newco 20, Inc. DE Comcast Newco 21, Inc. DE Comcast Newco 22, Inc. DE Comcast Newco 23, Inc. DE Comcast Newco 3, Inc. DE Comcast Newco 4, Inc. DE Comcast Newco 5, Inc. Comcast Newco 5, Inc. DE Comcast Newco 6, Inc. DE Comcast Newco 6, Inc. DE Comcast Newco 6, Inc. DE Comcast Newco 7, Inc. DE Comcast Newco 7, Inc. DE Comcast Newco 8, Inc. DE Comcast Newco 8, Inc. DE Comcast of Alameda, Inc. Comcast of Alameda, Inc. DE Comcast of Alameda, Inc. DE Comcast of Alameda, Inc. DE Comcast of Alameda, Inc. DE Comcast of Alameda, Inc. MD Comcast of Baltimore City, Inc. MD Comcast of Baltimore City, Inc. WA	Concest Newco 27 Inc
Comcast Newco 22, Inc. DE Comcast Newco 23, Inc. DE Comcast Newco 3, Inc. DE Comcast Newco 5, Inc. DE Comcast Newco 5, Inc. DE Comcast Newco 6, Inc. DE Comcast Newco 6, Inc. DE Comcast Newco 8, Inc. DE Comcast Newco 8, Inc. DE Comcast of Alabama. Inc. Comcast of Alameda, Inc. Comcast of Baltimore City, Inc. MD Comcast of Baltimore City, Inc. MD Comcast of Baltimore City, Inc. WA	Comcast Newco 20, Inc. DE
Comcast Newco 23, Inc. Comcast Newco 3, Inc. Comcast Newco 4, Inc. Comcast Newco 5, Inc. Comcast Newco 5, Inc. Comcast Newco 7, Inc. Comcast Newco 8, Inc. Comcast Newco 8, Inc. Comcast of Alabama Inc. Comcast of Alabama Inc. Comcast of Alabama Inc. Comcast of Alabama Inc. Comcast of Arizona, Inc. Comcast of Arizona, Inc. Comcast of Arizona, Inc. Comcast of Arkansas, Inc. Comcast of Arkansas, Inc. Comcast of Baltimore City, Inc. Comcast Of B	Concast Newco.21 Inc. 24 Cast Part of the State of the St
Comcast Newco 3, Inc. DE Comcast Newco 4, Inc. DE Comcast Newco 5, Inc. DE Comcast Newco 6, Inc. DE Comcast Newco 8, Inc. Comcast Newco 9, Inc. DE Comcast Newco 9, Inc. DE Comcast of Alabama, Inc. Comcast of Alabama, Inc. Comcast of Alameda, Inc. Comcast of Alameda, Inc. Comcast of Arizona, Inc. Comcast of Baltimore City, Inc.	
Comcast Newco 4 Inc. Comcast Newco 5, Inc. Comcast Newco 6, Inc. Comcast Newco 7, Inc. Comcast Newco 8, Inc. Comcast Newco 9, Inc. Comcast of Alabama Inc. Comcast of Alameda, Inc. Comcast of Alameda, Inc. Comcast of Anizona Inc. Comcast of Arkansas, Inc. Comcast of Arkansas, Inc. Comcast of Arkansas, Inc. Comcast of Baltimore City, Inc. Comcast of Baltimore City, I. P. Comcast of Bellevue, Inc. Comcast of Boson, Inc.	
Comcast Newco 5, Inc. Comcast Newco 6, Inc. Comcast Newco 7, Inc. Comcast Newco 8, Inc. Comcast Newco 9, Inc. Comcast of Alabama, Inc. Comcast of Alameda, Inc. Comcast of Alameda, Inc. Comcast of Anzona Inc. Comcast of Arkansas, Inc. Comcast of Arkansas, Inc. Comcast of Avalona LICC. Comcast of Baltimore City, LP. Comcast of Baltimore City, LP. Comcast of Bellevue, Inc. Comcast of Bellevue, Inc. Comcast of Boston Inc.	
Concast Newco Gilnc.1 27 Concast Newco 7, Inc. DE Concast Newco 8, Inc. Concast Newco 9, Inc. DE Concast of Alabama, Inc. Concast of Alameda, Inc. CA Concast of Anzona, Inc. Concast of Arkansas, Inc. Concast of Arkansas, Inc. Concast of Avalout LLC Concast of Baltimore City, LP. Concast of Baltimore City, LP. Concast of Bellevue, Inc. Concast of Boston, Inc.	
Comcast Newco 7, Inc. DE Comcast Newco 8, Inc. Comcast Newco 9, Inc. DE Comcast of Alabama, Inc. Comcast of Alameda, Inc. CA Comcast of Antona Inc. Comcast of Antona Inc. Comcast of Arkansas, Inc. Comcast of Arkansas, Inc. Comcast of Baltimore City, Inc. Comcast of Baltimore City, LP.	Concast Newcologilia Lander and Anna Anna Anna Anna Anna Anna Anna
Comcast Newco 9, Inc. Comcast of Alabama, Inc. Comcast of Alameda, Inc. Comcast of Arizona Inc. Comcast of Arkansas, Inc. Comcast of Arkansas, Inc. Comcast of Arkansas, Inc. Comcast of Baltimore City, Inc. Comcast of Baltimore City, LP. Comcast of Bellevue, Inc. Comcast of Boston, Inc. Comcast of Boston, Inc.	Comcast Newco 7, Inc. DE
Concast of Alabama, Inc. Comcast of Alameda, Inc. Comcast of Arizona Inc. Comcast of Arkansas, Inc. Comcast of Avalou, LLC Comcast of Baltimore City, Inc. Comcast of Baltimore City, L.P. Comcast of Bellevue, Inc. Comcast of Bellevue, Inc. Comcast of Boston, Inc.	
Concast of Alameda, Inc. Comcast of Antona Inc. Comcast of Arkansas, Inc. Comcast of Avalou LLC Comcast of Baltimore City, Inc. Comcast of Baltimore City, L P Comcast of Bellevue, Inc. Comcast of Bellevue, Inc. Comcast of Boston Inc.	
Comcast of Arizona Inc. Comcast of Arkansas, Inc. Comcast of Avalou ILIC Comcast of Baltimore City, Inc. Comcast of Baltimore City, L.P. Comcast of Bellevue, Inc. Comcast of Bellevue, Inc. Comcast of Boston Inc. WA	
Concast of Arkansas, Inc. DE Concast of Avalou LLC Concast of Baltimore City, Inc. Concast of Baltimore City, L P Comcast of Bellevue, Inc. WA Concast of Boston Inc	
Concast of Avalou LLC Concast of Baltimore City, Inc. Concast of Baltimore City, LP Comcast of Bellevue, Inc. Comcast of Bellevue, Inc. Comcast of Boston, Inc.	Concast of Artenses Inc. and a state and the care of
Comcast of Baltimore City, Inc. Comcast of Baltimore City, L.P. Comcast of Bellevue, Inc. Comcast of Boston, Inc.	
Concast of Baltimore City L.P. Comcast of Bellevue, Inc. Comcast of Boston, Inc.	Comcast of Baltimore City. Inc.
Concast of Bellevue, Inc. Concast of Boston, Inc.	Concast of Balumore City LP at the second
	Concast of Bellevue, Inc. WA
Comcast of Brockton, Inc. DE	
	Comcast of Brockton, Inc. DE
